UNITED STATES BANKRUPTCY COURT SOUTHERN DISTRICT OF NEW YORK

In re:) Chapter 11	
BALLY TOTAL FITNESS OF GREATER NEW YORK, INC., et) <u>al.,</u>) Case No. 07-12395 (BRL	(ر
Debtors.		
) Jointly Administered	

FINDINGS OF FACT, CONCLUSIONS OF LAW AND ORDER (I) APPROVING DISCLOSURE STATEMENT AND (II) CONFIRMING THE DEBTORS' FIRST AMENDED JOINT PREPACKAGED PLAN OF REORGANIZATION UNDER CHAPTER 11 OF THE BANKRUPTCY CODE

The Joint Prepackaged Chapter 11 Plan Of Reorganization Of Bally Total Fitness Holding Corporation And Its Affiliate Debtors, dated June 27, 2007 (the "Original Plan"), having been filed with the Bankruptcy Court (the "Court") on July 31, 2007 (Docket No. 37) by the above-captioned debtors and debtors-in-possession (collectively, the "Debtors")¹; and the Disclosure Statement For The Joint Prepackaged Chapter 11 Plan Of Reorganization Of Bally

TI DI

The Debtors in these proceedings are: Bally Total Fitness of Greater New York, Inc., Bally Total Fitness Holding Corporation, Bally Total Fitness Corporation, Bally ARA Corporation, Bally Fitness Franchising, Inc., Bally Franchise RSC, Inc., Bally Franchise RSC, Inc., Bally Franchise RSC, Inc., Bally Total Fitness Franchising, Inc., Bally Total Fitness International, Inc., Bally Total Fitness of California, Inc., Bally Total Fitness of Colorado, Inc., Bally Total Fitness of Connecticut Coast, Inc., Bally Total Fitness of Connecticut Valley, Inc., Bally Total Fitness of Minnesota, Inc., Bally Total Fitness of Missouri, Inc., Bally Total Fitness of Philadelphia, Inc., Bally Total Fitness of Rhode Island, Inc., Bally Total Fitness of the Mid-Atlantic, Inc., Bally Total Fitness of the Midwest, Inc., Bally Total Fitness of the Southeast, Inc., Bally Total Fitness of Toledo, Inc., Bally Total Fitness of Upstate New York, Inc., BTF Cincinnati Corporation, BTF Europe Corporation, BTF Indianapolis Corporation, BTF Minneapolis Corporation, BTF/CFI, Inc., BTFCC, Inc., BTFF Corporation, Greater Philly No. 1 Holding Company, Greater Philly No. 2 Holding Company, Health & Tennis Corporation of New York, Holiday Health Clubs of the East Coast, Inc., Holiday/Southeast Holding Corp., Jack La Lanne Holding Corp., New Fitness Holding Co., Inc., Nycon Holding Co., Inc., Rhode Island Holding Company, Tidelands Holiday Health Clubs, Inc., and U.S. Health, Inc.

Total Fitness Holding Corporation And Its Affiliate Debtors, dated June 27, 2007 (the "Disclosure Statement"), having been filed with this Court on July 31, 2007 (Docket No. 36); and the Disclosure Statement, and appropriate Ballots for voting on the Original Plan, having been transmitted to Holders² of Class 5 Claims and Class 6-A Claims against the Debtors as set forth in that certain Order (A) Scheduling Combined Hearing On Disclosure Statement And Confirmation Of Plan And Establishing Procedures For Objecting To Disclosure Statement And Plan; (B) Approving Form And Manner Of Notice Of Combined Hearing On Disclosure Statement And Confirmation Of Plan; (C) Approving Prepetition Solicitation Procedures; And (D) Granting Related Relief, dated as of August 2, 2007 (Docket No. 92, the "Solicitation **Procedures Order**"); and The First Amended Joint Prepackaged Chapter 11 Plan Of Reorganization Of Bally Total Fitness Holding Corporation And Its Affiliate Debtors, dated August 17, 2007 (the "**Plan**"), having been filed with this Court on August 17, 2007 (Docket No. 253) by the Debtors; and this Court having entered that certain Order Authorizing The Debtors, Pursuant To Section 1127(A) Of The Bankruptcy Code And Bankruptcy Rule 3019, To Modify Their Joint Prepackaged Chapter 11 Plan Of Reorganization, dated August 21, 2007 (Docket No. 268, the "Plan Modification Approval Order"), finding that the Plan is deemed accepted by all creditors who had previously accepted the Original Plan and that the Debtors were not required to prepare or distribute a new disclosure statement with respect to the Plan; and a copy of the Plan, as approved by the Plan Modification Approval Order and as subsequently modified since the entry of the Plan Modification Approval Order, being attached hereto as Exhibit 1; and a redline copy of the Plan showing the modifications to the Plan between the Confirmation

² All capitalized terms used and not otherwise defined in this Confirmation Order shall have the meanings ascribed to them in the Plan (as defined below).

Hearing and entry of the Plan Modification Approval Order being attached hereto as Exhibit 2 (such plan modifications, the "Subsequent Plan Modifications"); and the Debtors having filed their Memorandum Of Law In Support Of Entry Of An Order (I) Approving (A) The Debtors' Disclosure Statement Pursuant To Sections 1125 And 1126(B) Of The Bankruptcy Code, (B) Solicitation Of Votes And Voting Procedures, And (C) Forms Of Ballots, And (II) Confirming The Debtors' Joint Prepackaged Plan Of Reorganization Pursuant To Chapter 11 Of The Bankruptcy Code, with this Court on September 13, 2007 (the "Confirmation Memorandum"); and the hearing to consider the adequacy of the Disclosure Statement and the confirmation of the Plan having been held before this Court on September 17, 2007 (the "Confirmation Hearing") after due and sufficient notice was given to Holders of Claims against, and Interests in, the Debtors and other parties in interest in accordance with the Solicitation Procedures Order, the Bankruptcy Code, the Federal Rules of Bankruptcy Procedure (the "Bankruptcy Rules"), the local bankruptcy rules of this Court and General Order 203 of the United States Bankruptcy Court for the Southern District of New York, "Adoption of Prepackaged Chapter 11 Case Guidelines," dated February 24, 1999 (the "Prepack Guidelines"), in each case as established by the affidavits of service, mailing and/or publication filed with this Court prior to the Confirmation Hearing (collectively, the "Notice Affidavits");³ and upon all of the proceedings held before this Court and after full consideration of: (i) each of the objections to the adequacy of the Disclosure Statement and the confirmation of the Plan filed with this Court and not subsequently withdrawn, settled or deemed moot (the "Objections"); (ii) the Amended Affidavit Of Tabulation Of Votes With Respect To Joint Prepackaged Chapter 11 Plan Of Reorganization

_

³ The Notice Affidavits are located at Docket Nos. 215, 247, 252, 254, 259, 316, 338, 377 & 378.

Of Bally Total Fitness Holding Corporation And Its Affiliate Debtors, filed on August 8, 2007 (Docket No. 143) by MacKenzie Partners, Inc., the Debtors' voting agent; (iii) testimony proffered or presented at the Confirmation Hearing, (iv) the declarations and/or affidavits filed with this Court; (v) all other evidence proffered or adduced at, memoranda and objections filed in connection with and arguments of counsel made at, the Confirmation Hearing; and (vi) the entire record of the above-captioned chapter 11 cases; and after due deliberation thereon; and good cause appearing therefor;

IT IS HEREBY FOUND AND DETERMINED THAT:⁴

A. <u>Chapter 11 Petitions</u>. On July 31, 2007 (the "<u>Petition Date</u>"), the Debtors commenced voluntary cases under chapter 11 of the Bankruptcy Code with this Court (the "<u>Chapter 11 Cases</u>"). The Debtors have operated their business and managed their properties as debtors-in-possession pursuant to sections 1107(a) and 1108 of the Bankruptcy Code. No trustee or examiner has been appointed pursuant to section 1104 of the Bankruptcy Code. No statutory committee of unsecured creditors has been appointed pursuant to section 1102 of the Bankruptcy Code.

B. Exclusive Jurisdiction; Venue; Core Proceeding (28 U.S.C. §§ 157(b)(2) and 1334(a)). This Court has jurisdiction over the Chapter 11 Cases pursuant to 28 U.S.C. §§ 157 and 1334. Venue is proper pursuant to 28 U.S.C. §§ 1408 and 1409. Approval of the Disclosure Statement and confirmation of the Plan is a core proceeding under 28 U.S.C. § 157(b)(2) and this

⁴ The findings and conclusions set forth in this Confirmation Order and in the record of the Confirmation Hearing constitute this Court's findings of fact and conclusions of law pursuant to Bankruptcy Rule 7052, made applicable to this proceeding pursuant to Bankruptcy Rule 9014. To the extent any of the following findings of fact constitute conclusions of law, they are adopted as such. To the extent any of the following conclusions of law constitute findings of fact, they are adopted as such.

Court has exclusive jurisdiction to determine whether the Plan complies with the applicable provisions of the Bankruptcy Code.

- C. <u>Judicial Notice</u>. This Court takes judicial notice of the docket of the Chapter 11 Cases maintained by the Clerk of this Court and/or its duly-appointed agent, including, without limitation, all pleadings and other documents filed, all orders entered and all evidence and arguments made, proffered or adduced at, the hearings held before this Court during the pendency of the Chapter 11 Cases.
- D. Adequacy of Solicitation and Disclosure Statement. Section 1126(b) of the Bankruptcy Code applies to the solicitation of acceptances and rejections of the Original Plan prior to the commencement of these Chapter 11 Cases. The solicitation of acceptances and rejections of the Original Plan was exempt from the registration requirements of the Securities Act of 1933 (as amended, and including the rules and regulations promulgated thereunder, the "Securities Act") and applicable state and local securities laws, and no other non-bankruptcy law applies to the solicitation. The Disclosure Statement contains adequate information within the meaning of, and for all purposes under, sections 1125 and 1126(b) of the Bankruptcy Code, and is hereby approved in all respects. Votes for acceptance or rejection of the Original Plan (as modified by the Plan and as approved in the Plan Modification Approval Order) were solicited in good faith and in compliance with sections 1125 and 1126 of the Bankruptcy Code, Bankruptcy Rules 3017 and 3018 and all other applicable provisions of the Bankruptcy Code, the Bankruptcy Rules, the local bankruptcy rules of this Court, the Prepack Guidelines and all other rules, laws and regulations. All procedures used to distribute Ballots to the applicable Holders of Claims and to tabulate the Ballots were fair and reasonable and conducted in accordance with the Solicitation Procedures Order and the applicable provisions of the Bankruptcy Code, the

Bankruptcy Rules, the local bankruptcy rules of this Court, the Prepack Guidelines and all other applicable rules, laws, and regulations. The forms of the Ballots were sufficiently consistent with the form of ballot annexed to the Prepack Guidelines and adequately addressed the particular needs of these Chapter 11 Cases and were appropriate for each class of Claims entitled to vote to accept or reject the Original Plan (as modified by the Plan and as approved in the Plan Modification Approval Order).

E. Transmittal and Mailing of Materials; Notice. The Debtors have given proper and sufficient notice of the hearing to approve the Disclosure Statement as required by Bankruptcy Rule 3017(a). The Debtors have given proper and sufficient notice of the Confirmation Hearing as required by Bankruptcy Rule 3017(d). The solicitation of votes to accept or reject the Original Plan (as modified by the Plan and as approved in the Plan Modification Approval Order) satisfies Bankruptcy Rule 3018. The Original Plan, the Plan and the Disclosure Statement were transmitted to all creditors entitled to vote on the Original Plan (as modified by the Plan and as approved in the Plan Modification Approval Order) and sufficient time was prescribed for such creditors to accept or reject the Original Plan (as modified by the Plan and as approved in the Plan Modification Approval Order). The solicitation materials and solicitation procedures comply with section 1126 of the Bankruptcy Code, thereby satisfying the requirements of Bankruptcy Rule 3018. Due, adequate and sufficient notice of the Disclosure Statement, the Original Plan, the Plan and the Confirmation Hearing, along with deadlines for voting on or filing objections to the Original Plan, the Plan and the Disclosure Statement, has been given to all known Holders of Claims and Interests substantially in accordance with the procedures set forth in the Solicitation Procedures Order. The Disclosure Statement, Original Plan, Plan, Ballots and Solicitation Procedures Order were transmitted and served in compliance

with the Solicitation Procedures Order and the applicable provisions of the Bankruptcy Code, the Bankruptcy Rules, the local bankruptcy rules of this Court, the Prepack Guidelines and all other applicable rules, laws, and regulations, and such transmittal and service were adequate and sufficient under the circumstances.

- F. <u>Subsequent Plan Modifications</u>. Adequate and sufficient notice of the Subsequent Plan Modifications has been given and no other or further notice is or shall be required and such Subsequent Plan Modifications are approved in full.
- G. <u>Burden of Proof.</u> The Debtors, as proponents of the Plan, have the burden of proving the satisfaction of the elements of Sections 1129(a) and (b) of the Bankruptcy Code by a preponderance of the evidence.
- H. <u>Plan Compliance with Bankruptcy Code (11 U.S.C. § 1129(a)(1))</u>. The Plan complies with the applicable provisions of the Bankruptcy Code, thereby satisfying Section 1129(a)(1) of the Bankruptcy Code.
 - (a) Proper Classification (11 U.S.C. §§ 1122 & 1123(a)(1)). In addition to Administrative Claims, DIP Lenders Claims and Priority Tax Claims, which need not be classified, the Plan designates eleven (11) Classes of Claims and Interests. The Claims and Interests placed in each Class are substantially similar to other Claims or Interests, as the case may be, in each such Class. Valid business, factual and legal reasons exist for separately classifying the various Classes of Claims and Interests created under the Plan and such Classes do not unfairly discriminate between Holders of Claims or Interests. Thus, the Plan satisfies Sections 1122 and 1123(a)(1) of the Bankruptcy Code.
 - (b) <u>Specify Unimpaired Classes (11 U.S.C. § 1123(a)(2))</u>. The Plan specifies that Class 1 Non-Tax Priority Claims, Class 2 Other Secured Claims, Class 3 Unimpaired

Unsecured Claims, Class 4 Prepetition Lenders Claims and Class 9 Old Equity Interests in Affiliate Debtors are Unimpaired under the Plan, thereby satisfying Section 1123(a)(2) of the Bankruptcy Code.

- (c) Specify Treatment of Impaired Classes (11 U.S.C. § 1123(a)(3)). The Plan designates Class 5 Prepetition Senior Notes Claims, Class 6-A Prepetition Senior Subordinated Notes Claims, Class 6-B-1 Rejection Claims Against Only Bally, Class 6-B-2 Rejection Claims Against Any Affiliate Debtor, Class 7 Old Common Stock and Class 8 Old Unexercised Equity Interests as Impaired and specifies the treatment of Claims and Interests in those Classes, thereby satisfying Section 1123(a)(3) of the Bankruptcy Code.
- (d) No Discrimination Within Classes (11 U.S.C. § 1123(a)(4)). The Plan provides for the same treatment by the Debtors for each Claim or Interest in each respective Class unless the Holder of a particular Claim or Interest has agreed to a less favorable treatment of such Claim or Interest, thereby satisfying Section 1123(a)(4) of the Bankruptcy Code.
- (e) <u>Classification Takes Into Account Subordination Rights</u>. The classification and manner of satisfying all Claims and Interests under the Plan takes into consideration all contractual, legal and equitable subordination and turnover rights, whether arising under general principles of equitable subordination, section 510 of the Bankruptcy Code, or otherwise, that a Holder of a Claim or Interest may have against other Holders of a Claim or Interest with respect to any distribution made pursuant to the Plan.

- (f) <u>Implementation of Plan (11 U.S.C. § 1123(a)(5))</u>. The Plan provides adequate and proper means for implementation of the Plan, thereby satisfying Section 1123(a)(5) of the Bankruptcy Code.
- (g) Non-Voting Equity Securities (11 U.S.C. § 1123(a)(6)). The Plan provides that the certificates of incorporation of the Reorganized Debtors that are corporations shall prohibit the issuance of non-voting equity securities to the extent required by Section 1123(a)(6) of the Bankruptcy Code, thereby satisfying Section 1123(a)(6) of the Bankruptcy Code.
- (h) Selection of Directors and Officers (11 U.S.C. § 1123(a)(7)). Pursuant to Section 5.4(b) of the Plan, the initial officers of Reorganized Bally shall be the officers of Bally existing immediately prior to the Effective Date. The board of directors of Reorganized Bally were identified by the Debtors at or prior to the Confirmation Hearing. At or prior to the Confirmation Hearing, the Debtors disclosed the identity and affiliations of any Person (each, a "New Director") proposed to serve on the initial board of directors of Reorganized Bally (the "Reorganized Board"), and, to the extent such Person is an insider other than by virtue of being a director, the nature of any compensation for such Person. At the Effective Date, the board of directors of Reorganized Bally shall be reconstituted and declassified as provided in the Amended Certificate of Incorporation of Reorganized Bally, and the New Directors shall be appointed as the directors of the Reorganized Bally. The boards of directors and initial officers of the Reorganized Affiliate Debtors on the Effective Date shall be comprised of the same individuals who currently serve in such capacities (other than those Persons that are deemed to have resigned on the Effective Date pursuant to the Plan). The directors

and officers of the Reorganized Debtors were selected in a manner consistent with the interests of creditors and with public policy, thereby satisfying Section 1123(a)(7) of the Bankruptcy Code. No action required by the Plan (including, but not limited to the appointment of new directors to the boards of any of the Debtors) shall be a "Change of Control" under the DIP Credit Agreement, the Prepetition Senior Notes Indenture, the Prepetition Subordinated Notes Indenture, the New Second Lien Notes Indenture or the New Harbinger Subordinated Notes Indenture.

- (i) Additional Plan Provisions (11 U.S.C. § 1123(b)). The Plan's provisions are appropriate, in the best interests of the Debtors and their Estates and not inconsistent with the applicable provisions of the Bankruptcy Code, including, without limitation, provisions for (i) the assumption or rejection of executory contracts and unexpired leases; (ii) the Reorganized Debtors' retention of certain Litigation Claims that the Debtors had or had power to assert immediately prior to the Effective Date, whether directly or derivatively; and (iii) releases of various persons and entities, exculpation of various persons and entities with respect to actions related to or taken in furtherance of the Chapter 11 Cases and preliminary and permanent injunctions against certain actions against the Debtors, their Estates and their properties.
- (j) <u>Identification of Plan (Bankruptcy Rule 3016(a))</u>. The Plan is dated and identifies the entity submitting it, thereby satisfying Bankruptcy Rule 3016(a).
- I. The Debtors' Compliance with Bankruptcy Code (11 U.S.C. § 1129(a)(2)). The Debtors have complied with the applicable provisions of the Bankruptcy Code, thereby satisfying Section 1129(a)(2) of the Bankruptcy Code. Specifically:

- (a) The Debtors are proper debtors under Section 109 of the Bankruptcy Code and proper proponents of the Plan under Section 1121(a) of the Bankruptcy Code.
- (b) The Debtors have complied with the applicable provisions of the Bankruptcy Code, except as otherwise provided or permitted by orders of this Court.
- (c) The Debtors have complied with the applicable provisions of the Bankruptcy Code, the Bankruptcy Rules, the local bankruptcy rules of this Court, the Prepack Guidelines and the Solicitation Procedures Order in transmitting the Original Plan, the Plan, the Disclosure Statement, the Ballots and related documents and notices and in soliciting and tabulating votes on the Original Plan.
- J. Plan Proposed in Good Faith (11 U.S.C. § 1129(a)(3)). The Debtors have proposed the Plan in good faith and not by any means forbidden by law, thereby satisfying Section 1129(a)(3) of the Bankruptcy Code. In determining that the Plan has been proposed in good faith, this Court has examined the totality of the circumstances surrounding the filing of the Chapter 11 Cases and the formulation and confirmation of the Plan. The Debtors, the Prepetition Noteholders Committee, the New Investors and the Backstop Parties (and each of their respective professionals) have negotiated the Plan and participated in the Plan formulation process in good faith. The Chapter 11 Cases were filed and the Plan was proposed with the legitimate and honest purpose of reorganizing the Debtors and maximizing the value of the Debtors' assets and expeditiously distributing the New Securities and Documents and other consideration to the Debtors' creditors and interest holders pursuant to the Plan.
- K. Payments for Services or Costs and Expenses (11 U.S.C. § 1129(a)(4)). Any payment made or to be made by the Debtors for services or for costs and expenses in or in connection with the Chapter 11 Cases, or in connection with the Plan and incident to the Chapter 11 Cases requiring approval, has been approved by, or is subject to the approval of, this Court as reasonable, thereby satisfying Section 1129(a)(4) of the Bankruptcy Code.

- L. <u>Directors, Officers and Insiders (11 U.S.C. § 1129(a)(5))</u>. The Debtors have complied with Section 1129(a)(5) of the Bankruptcy Code. The identity and affiliations of the persons proposed to serve as the initial directors and officers of the Reorganized Debtors after confirmation of the Plan have been fully disclosed to the extent such information is available, and the appointment to, or continuance in, such offices of such persons is consistent with the interests of Holders of Claims against and Interests in the Debtors and with public policy. To the extent available, the identity of any insider that will be employed or retained by the Reorganized Debtors and the nature of such insider's compensation have also been fully disclosed.
- M. <u>No Rate Changes (11 U.S.C. § 1129(a)(6))</u>. The Debtors are not subject to any governmental regulatory commission with jurisdiction, after confirmation of the Plan, over the rates of the Debtors. Thus, Section 1129(a)(6) of the Bankruptcy Code is not applicable in the Chapter 11 Cases.
- N. <u>Best Interests of Creditors Test (11 U.S.C. § 1129(a)(7))</u>. The Plan satisfies Section 1129(a)(7) of the Bankruptcy Code. The liquidation analysis set forth in Exhibit 4 to the Disclosure Statement and other evidence proffered or adduced at the Confirmation Hearing (a) are persuasive and credible, (b) have not been controverted by other evidence and (c) establish that each Holder of a Claim or Interest in an Impaired Class either (i) has accepted the Plan or (ii) will receive or retain under the Plan, on account of such Claim or Interest, property of a value, as of the Effective Date of the Plan, that is not less than the amount that it would receive if the Debtors were liquidated under Chapter 7 of the Bankruptcy Code on such date.
- O. <u>Acceptance by Classes (11 U.S.C. § 1129(a)(8))</u>. Class 1 Non-Tax Priority Claims, Class 2 Other Secured Claims, Class 3 Unimpaired Unsecured Claims, Class 4 Prepetition Lenders Claims and Class 9 Old Equity Interests in Affiliate Debtors are Unimpaired

by the Plan. Under Section 1126(f) of the Bankruptcy Code, such Holders of Claims and Interests are conclusively presumed to have accepted the Plan. Class 5 Prepetition Senior Notes Claims and Class 6-A Prepetition Senior Subordinated Notes Claims have voted to accept the Plan in accordance with Sections 1126(b) and (c) of the Bankruptcy Code and the Plan Modification Approval Order. Class 6-B-1 Rejection Claims Against Only Bally, Class 6-B-2 Rejection Claims Against Any Affiliate Debtor, Class 7 Old Common Stock and Class 8 Old Unexercised Equity Interests are deemed to have rejected the Plan. Although Section 1129(a)(8) of the Bankruptcy Code is not satisfied with respect to such rejecting Classes identified above, the Plan may nevertheless be confirmed because the Plan satisfies Section 1129(b) of the Bankruptcy Code with respect to such rejecting Classes. Section 4.5 of the Plan contemplates non-consensual confirmation of the Plan.

- P. Treatment of Administrative Claims, Non-Tax Priority Claims and Priority Tax Claims (11 U.S.C. § 1129(a)(9)). The treatment of Allowed Administrative Claims and Allowed Non-Tax Priority Claims under Section 3.1(a) and Section 3.2(a) of the Plan, respectively, satisfies the requirements of Section 1129(a)(9)(A) and (B) of the Bankruptcy Code, and the treatment of Allowed Priority Tax Claims under Section 3.1(b) of the Plan satisfies the requirements of Section 1129(a)(9)(C) of the Bankruptcy Code, thereby satisfying Section 1129(a)(9) of the Bankruptcy Code.
- Q. Acceptance by At Least One Impaired Class of Claims (11 U.S.C. § 1129(a)(10)). Class 5 Prepetition Senior Notes Claims and Class 6-A Prepetition Senior Subordinated Notes Claims are impaired classes of Claims that have voted to accept the Plan in accordance with Sections 1126(b) and (c) of the Bankruptcy Code, determined without including any acceptance of the Plan by "insiders," thereby satisfying Section 1129(a)(10) of the Bankruptcy Code.

- R. Feasibility (11 U.S.C. § 1129(a)(11)). The evidence proffered or adduced at the Confirmation Hearing (i) is persuasive and credible, (ii) has not been controverted by other evidence, and (iii) establishes that the Plan is feasible and that there is a reasonable prospect of the Reorganized Debtors being able to meet their financial obligations under the Plan and their business in the ordinary course and that confirmation of the Plan is not likely to be followed by the liquidation or the need for further financial reorganization of the Reorganized Debtors, thereby satisfying the requirements of section 1129(a)(11) of the Bankruptcy Code. This Court has made this finding after taking into account, and fully considering, pending litigation against the Debtors and Reorganized Debtors.
- S. Payment of Fees (11 U.S.C. § 1129(a)(12)). All fees payable under 28 U.S.C. § 1930 have been paid or will be paid pursuant to Section 12.5 of the Plan, thereby satisfying Section 1129(a)(12) of the Bankruptcy Code.
- T. Continuation of Retiree Benefits (11 U.S.C. § 1129(a)(13)). Pursuant to Section 7.4 of the Plan, the Reorganized Debtors shall continue to pay all retiree benefits of the Debtors for the duration of the period for which the Debtors have obligated themselves to provide such benefits, thereby satisfying Section 1129(a)(13) of the Bankruptcy Code to the extent such section is applicable to the Debtors.
- U. <u>No Unfair Discrimination; Fair and Equitable (11 U.S.C. § 1129(b))</u>. Class 6-B-1 Rejection Claims Against Only Bally, Class 6-B-2 Rejection Claims Against Any Affiliate Debtor, Class 7 Old Common Stock and Class 8 Old Unexercised Equity Interests (collectively, the "<u>Rejecting Classes</u>") are deemed to have rejected the Plan. The evidence proffered or adduced at the Confirmation Hearing (i) is persuasive and credible, (ii) has not been controverted by other evidence, and (iii) establishes that the Plan does not discriminate unfairly, and is fair

and equitable, with respect to the Rejecting Classes, as required by sections 1129(b)(1) and (b)(2) of the Bankruptcy Code, because either (a) the Holders of such Impaired Claims represented by the respective Rejecting Class will receive or retain on account of such Claim property of a value, as of the effective date of the plan, equal to the allowed amount of such Claim or (b) no Holder of any interest that is junior to the Impaired Interest represented by the respective Rejecting Class will receive or retain any property under the Plan on account of such junior interest, and no Holder of a Claim in a Class senior to the Rejecting Classes is receiving more than 100% recovery on account of its Claim. Thus, the Plan may be confirmed notwithstanding the deemed rejection of the Plan by the Rejecting Classes.

- V. <u>Principal Purpose of Plan (11 U.S.C. § 1129(d))</u>. The principal purpose of the Plan is not the avoidance of taxes or the avoidance of the application of Section 5 of the Securities Act of 1933 (15 U.S.C. § 77e).
- W. Good Faith Solicitation (11 U.S.C. § 1125(e)). The evidence proffered or adduced at the Confirmation Hearing (i) is persuasive and credible, (ii) has not been controverted by other evidence, and (iii) establishes that the Debtors, the Reorganized Debtors, the Prepetition Noteholders Committee, the Prepetition Senior Noteholders, the Prepetition Senior Subordinated Noteholders, the Prepetition Senior Notes Indenture Trustee, the Prepetition Senior Subordinated Notes Indenture Trustee, the Backstop Parties, the New Investors (if the Harbinger Investment Effective Date Condition has been satisfied) and Liberation (if the Harbinger Investment Effective Date Condition has been satisfied) and their respective Related Persons (collectively, the "Protected Parties") have (a) solicited acceptances or rejections of the Original Plan and subscriptions to the Rights Offering in good faith and in compliance with the applicable provisions of the Bankruptcy Code, including without limitation, sections 1125 and 1126(b) of

the Bankruptcy Code, the Bankruptcy Rules, the local bankruptcy rules of this Court, the Prepack Guidelines and any applicable non-bankruptcy law, rule or regulation governing the adequacy of disclosure in connection with such solicitation and (b) participated in good faith and in compliance with the applicable provisions of the Bankruptcy Code in the offer and issuance of any securities under the Plan. Accordingly, each of the Protected Parties is entitled to the protections afforded by Section 1125(e) of the Bankruptcy Code and the exculpation, release and limitation of liability provisions and protections set forth in Article X of the Plan.

- X. <u>Satisfaction of Confirmation Requirements</u>. Based on the foregoing, the Plan satisfies the requirements for confirmation set forth in Section 1129 of the Bankruptcy Code.
- Y. <u>Retention of Jurisdiction</u>. This Court may properly retain jurisdiction over the matters set forth in Article Eleven of the Plan and as contemplated herein.
- Z. <u>Findings Regarding Third-Party Releases and Related Provisions</u>. The release, exculpation and injunction provisions contained in the Plan, including, without limitation, those contained in Article Ten of the Plan, (i) have been negotiated in good faith and at arms' length, (ii) are consistent with sections 105, 1123(b)(6), 1129, and 1142 of the Bankruptcy Code, (iii) are each necessary for the Debtors' successful reorganization and are integral to the structure of the Plan and formed part of the agreement among all parties in interest embodied therein, and (iv) satisfy the applicable standards contained in <u>In re Metromedia Fiber Network, Inc.</u>, 416 F.3d 136 (2d Cir. 2005).

NOW THEREFORE, IT IS HEREBY ORDERED, ADJUDGED AND DECREED THAT:

1. <u>Approval of Disclosure Statement</u>. The Disclosure Statement (a) contains sufficient information of a kind necessary to satisfy the disclosure requirements of applicable

non-bankruptcy law, including, without limitation, the Securities Act, (b) contains "adequate information" (as such term is defined in section 1125(a)(1) and used in section 1126(b)(2) of the Bankruptcy Code) with respect to the Debtors, the Plan, and the transactions contemplated therein, and (c) is approved in all respects.

- 2. <u>Approval of Solicitation and Notice Procedures</u>. The notice and solicitation procedures utilized by the Debtors in soliciting acceptances and rejections of the Original Plan and to provide notice of this Confirmation Hearing are approved in all respects. The form of Ballots used to solicit votes on the Original Plan are approved in all respects.
- 3. <u>Confirmation of the Plan</u>. The Plan is approved and confirmed under Section 1129 of the Bankruptcy Code. Each of the terms and conditions of the Plan, and the exhibits and schedules thereto, are an integral part of the Plan and are incorporated by reference into this Confirmation Order. The Plan complies with all applicable provisions of the Bankruptcy Code, the Bankruptcy Rules, the local bankruptcy rules of this Court and the Prepack Guidelines relating to and regarding confirmation.
- 4. <u>Objections</u>. All Objections to approval of the Disclosure Statement and confirmation of the Plan that have not been withdrawn, waived, or settled and all reservations of rights included therein, are overruled on the merits and for the reasons set forth on the record at the Confirmation Hearing. All withdrawn objections are deemed withdrawn with prejudice.
- 5. <u>Provisions of Plan and Confirmation Order Nonseverable and Mutually Dependent</u>. The provisions of the Plan and this Confirmation Order, including the findings of fact and conclusions of law set forth herein, are nonseverable and mutually dependent.
- 6. <u>Plan Classification Controlling</u>. The classification of Claims and Interests for purposes of the distributions to be made under the Plan shall be governed solely by the terms of

the Plan. The classifications set forth on the Ballots tendered to or returned by the Debtors' creditors in connection with voting on the Plan (a) were set forth on the Ballots solely for purposes of voting to accept or reject the Plan, (b) do not necessarily represent, and in no event shall be deemed to modify or otherwise affect, the actual classification of such Claims or Interests under the Plan for distribution purposes, and (c) may not be relied upon by any creditor or interest holder as representing the actual classification of such Claims or Interests under the Plan for distribution or any other purpose (other than for evidencing the vote of such party on the Original Plan).

- Binding Effect. Pursuant to Section 1141 of the Bankruptcy Code, effective as of the Effective Date, the provisions of the Plan (including the exhibits and schedules to, and all documents and agreements executed pursuant to or in connection with, the Plan) and this Confirmation Order shall be binding on (a) the Debtors, (b) all Holders of Claims against and Interests in any of the Debtors, whether or not Impaired under the Plan and whether or not such Holders have accepted or rejected the Plan, (c) each Person or Entity receiving, retaining or otherwise acquiring property under the Plan, (d) any non-Debtor party to an executory contract or unexpired lease with a Debtor, (e) any Person or Entity making an appearance in the Chapter 11 Cases or any other party-in-interest in these Chapter 11 Cases, and (f) each of the foregoing's respective heirs, successors, assigns, trustees, executors, administrators, affiliates, officers, directors, agents, representatives, attorneys, beneficiaries, or guardians.
- 8. <u>Claims Arising Under the DIP Credit Agreement</u>. On the Effective Date, any and all DIP Lenders Claims shall be (A) paid in full in Cash or (B) assumed by the applicable Reorganized Debtors pursuant to the terms and conditions of the New Credit Agreement. On the full payment or other satisfaction of such Claims, unless such DIP Lenders Liens have been

continued as part of the New Credit Agreement pursuant to clause (B) above, the DIP Lenders Liens shall be deemed released, terminated and extinguished, in each case without further notice to or order of this Court, act or action under applicable law, regulation, order, or rule or the vote, consent, authorization or approval of any Person.

9. Revesting of Assets. Subject to the Restructuring Transactions permitted by Section 5.2 of the Plan, after the Effective Date, the Reorganized Debtors shall continue to exist as separate legal entities in accordance with the applicable law in the respective jurisdiction in which they are incorporated or formed and pursuant to their respective certificates or articles of incorporation and by-laws, or other applicable organizational documents, in effect immediately prior to the Effective Date, except to the extent such certificates or articles of incorporation and by-laws, or other applicable organizational documents, are amended, amended and restated or otherwise modified under the Plan. Notwithstanding anything to the contrary in the Plan, including Section 5.1 thereof as to the limited substantive consolidation provided therein, the Claims of a particular Debtor or Reorganized Debtor shall remain the obligations solely of such Debtor or Reorganized Debtor and shall not become obligations of any other Debtor or Reorganized Debtor solely by virtue of the Plan or the Chapter 11 Cases. Except as otherwise provided in the Plan or this Confirmation Order, on and after the Effective Date, all property and assets of the Estates of the Debtors, including all claims, rights, and Litigation Claims of the Debtors, and any other assets or property acquired by the Debtors or the Reorganized Debtors under or in connection with the Plan, shall vest in the Reorganized Debtors free and clear of all Claims, Liens, charges, other encumbrances, and Interests, subject to the Restructuring Transactions and Liens which survive the occurrence of the Effective Date as described in Article Three of the Plan. On and after the Effective Date, the Reorganized Debtors may (i)

operate their businesses, (ii) use, acquire, transfer and dispose of property and (iii) compromise or settle any Claims, in each case without notice to, hearing before, supervision of or approval by this Court and free and clear of any restrictions of the Bankruptcy Code, the Bankruptcy Rules or the local bankruptcy rules of this Court, other than restrictions expressly imposed by the Plan or this Confirmation Order.

- 10. <u>Distributions are Fair</u>. The distribution of Cash, New Common Stock and the other New Securities and Documents to the Holders of Allowed Claims and Allowed Interests in exchange for their Allowed Claims and Allowed Interests, as applicable, is fair and for reasonably equivalent value.
- Distributions Exempt from Securities Laws. The New Common Stock, the New Senior Second Lien Notes, any New Harbinger Subordinated Notes, any New Subordinated Notes, any New Junior Subordinated Notes, any Rights Offering Senior Subordinated Notes (and offer of the Rights for the Rights Offering Period to Holders of Allowed Class 6-A and 6-B-1 Claims), and any other New Securities and Documents issued or deemed issued under the Plan are exempt from registration under the Securities Act of 1933, and any state or local law requiring registration, to the fullest extent permitted pursuant to section 1145 of the Bankruptcy Code, except to the extent that: (i) the Holders of any of the foregoing are "underwriters," as that term is defined in section 1145 of the Bankruptcy Code, (ii) the New Common Stock is issued to the New Investors, in which case such New Common Stock is exempt from such registration pursuant to Section 4(2) of the Securities Act, and/or (iii) in the event that the Backstop Rights Offering Effective Date Condition is satisfied, certain of the New Securities and Documents are issued to the Backstop Parties, in which case such New Securities and Documents are exempt from such registration pursuant to Section 4(2) of the Securities Act.

- 12. Exit Financing. On the Effective Date, the Reorganized Debtors are authorized to enter into the New Credit Agreement, as well as execute, deliver, file, record and issue any notes, documents, instruments or agreements in connection therewith and perform their obligations under the New Credit Agreement and any such notes, documents, instruments or agreements in connection therewith (collectively with the New Credit Agreement, the "New Credit Agreement Documents"), in each case without further notice to or order of this Court, act or action under applicable law, regulation, order, or rule or the vote, consent, authorization or approval of any Person (other than as expressly required by the New Credit Agreement). The Liens and security interests to be granted by the Reorganized Debtors pursuant to the terms of the New Credit Agreement and the New Senior Second Lien Notes Indenture shall be deemed perfected on the Effective Date, subject only to such Liens and security interests as may be permitted under the New Credit Agreement and the New Senior Second Lien Notes Indenture, as applicable. The New Credit Agreement Documents shall be in form and substance acceptable to the Debtors, the New Agent and the New Lenders.
- 13. New Stockholders Agreement and Registration Rights Agreement. In the event that the Backstop Rights Offering Effective Date Condition is satisfied, Reorganized Bally is authorized to enter into and consummate the transactions contemplated by the New Stockholders Agreement and Registration Rights Agreement and such documents, and any agreement or document entered into in connection therewith, shall become effective and binding in accordance with their respective terms and conditions upon the parties thereto, in each case without further notice to or order of this Court, act or action under applicable law, regulation, order, or rule or the vote, consent, authorization or approval of any Person (other than as expressly required by the New Stockholders Agreement and Registration Rights Agreement, as applicable). In the

event that the Harbinger Investment Effective Date Condition is satisfied, there shall be no New Stockholders Agreement or Registration Rights Agreement in connection with the Plan.

- 14. Releases, Exculpation and Limitation of Liability. The releases, exculpation and limitation of liability provisions contained in the Plan, including, but not limited to, those provided in Article Ten of the Plan, are fair and equitable and given for valuable consideration and are in the best interest of the Debtors and all parties-in-interest, and, accordingly, are hereby authorized, approved and binding on all Persons and Entities described therein. To the extent that a release or other provision in the Plan constitutes a compromise of a controversy, this Confirmation Order shall constitute an order under Bankruptcy Rule 9019 approving such compromise.
- 15. Injunctions. The injunctions contained in the Plan, including, but not limited to, those provided in Section 10.2(d) and Section 10.6 of the Plan, are hereby authorized, approved and binding on all Persons and Entities described therein. Except as otherwise provided in the Plan, this Confirmation Order or in any document, instrument, release, or other agreement entered into in connection with the Plan or approved by order of this Court, this Confirmation Order constitutes an injunction from and after the Effective Date restraining all Persons or Entities who have held, hold, or may hold Claims against or Interests in the Debtors from taking any of the following actions against any of the Debtors, the Reorganized Debtors, the Estate(s), or any of their respective assets or property: (A) commencing or continuing, in any manner or in any place, any action or other proceeding with respect to any such Claims or Interests; (B) enforcing, attaching, collecting, or recovering in any manner any judgment, award, decree or order with respect to any such Claims or Interests; (C) creating, perfecting, or enforcing any Lien or encumbrance with respect to any such Claims or Interests; (D) asserting a setoff or right of

subrogation of any kind against any debt, liability or obligation due to the Debtors with respect to any such Claims or Interests; and (E) commencing or continuing, in any manner or in any place, any action that does not comply with or is inconsistent with the provisions of the Plan with respect to any such Claims or Interests; provided, however, that nothing contained herein shall preclude such Persons or Entities from exercising their rights pursuant to and consistent with the terms of the Plan, this Confirmation Order or in any document, instrument, release, or other agreement entered into in connection with the Plan or approved by order of this Court. Pursuant to Section 10.7 of the Plan, all injunctions or stays provided for in the Chapter 11 Cases under sections 105 or 362 of the Bankruptcy Code, or otherwise, and in existence on the Confirmation Date, shall remain in full force and effect until the Effective Date.

16. SEC Issues. Except with respect to the Plan-Related Claims (as defined below), nothing in this Confirmation Order or the Plan (i) shall effect a release of any Claim by the United States Government or any of its agencies or any state and local authority whatsoever, including, without limitation, any Claim arising under the Internal Revenue Code, the environmental laws or any criminal laws of the United States or any state and local authority against the Released Parties, but excluding any Plan-Related Claims, (ii) enjoin the United States or any state or local authority from bringing any Claim, suit, action or other proceedings against the Released Parties for any liability whatsoever, including, without limitation, any Claim, suit or action arising under the Internal Revenue Code, the environmental laws or any criminal laws of the United States or any state or local authority, but excluding any Plan-Related Claims, or (iii) exculpate any Released Party from any liability to the United States Government or any of its agencies or any state and local authority whatsoever, including any liabilities arising under the Internal Revenue Code, the environmental laws or any criminal laws of the United States or any

state and local authority against the Released Parties, but excluding any liabilities in respect of any Plan-Related Claims. Notwithstanding anything to the contrary in this Confirmation Order or the Plan, the Securities and Exchange Commission ("SEC") expressly reserves its right to continue to investigate, and, in its sole discretion, prosecute and enforce any and all Claims against any or all of the Debtors or the Reorganized Debtors arising from any prepetition violations by any Debtor of any of the U.S. securities laws other than Plan-Related Claims (collectively, the "Reserved SEC Claims"), including, without limitation, any claims for disgorgement of any benefits received by any Debtor as a result of any such violations and any Claims for penalties imposed by the SEC in respect of any such violations. For the avoidance of doubt, pursuant to Section 10.2(b) of the Plan, all Plan-Related Claims of the United States Government or any of its agencies or any state and local authority whatsoever are released, waived and discharged. Nothing in this Confirmation Order or the Plan shall result in the discharge of any Reserved SEC Claims, and the SEC expressly reserves its rights to assert that any and all Reserved SEC Claims are non-dischargeable as against the Reorganized Debtors pursuant to Sections 1141(d)(6)(a) and 523(a)(2)(A) of the Bankruptcy Code. The SEC has advised this Court and the Debtors that, as of the entry of this Confirmation Order, it has not yet determined whether to assert any Reserved SEC Claims against any or all of the Debtors or Reorganized Debtors.

17. Assumed Contracts and Leases.

(a) <u>Generally</u>. All of the executory contracts and unexpired leases of the Debtors are hereby assumed in accordance with the provisions and requirements of Sections 365 and 1123 of the Bankruptcy Code, effective as of the Effective Date (the "<u>Assumed Contracts</u> <u>and/or Leases</u>"), except those executory contracts (including, without limitation, employment

agreements) and unexpired leases that (i) have been previously assumed or rejected by order of this Court, (ii) are the subject of a motion to reject pending on the Effective Date, (iii) are identified on Exhibit E-1 or Exhibit E-2 to the Plan, (iv) are rejected pursuant to the terms of the Plan or this Confirmation Order, or (v) previously expired or terminated pursuant to its own terms.

- (b) Compensation and Benefit Programs. Except as otherwise expressly provided in the Plan, this Confirmation Order or listed on Exhibit E-1 or Exhibit E-2 to the Plan, all employment and severance policies, and all compensation and benefit plans, policies, and programs of the Debtors applicable to their employees, retirees, and non-employee directors and the employees and retirees of its subsidiaries, including, without limitation, all savings plans, retirement plans, healthcare plans, disability plans, severance benefit plans, incentive plans (including, without limitation, the Prepetition Management Incentive Plan), life, and accidental death and dismemberment insurance plans, are treated as executory contracts under the Plan and included in the defined term "Assumed Contracts and/or Leases". Any payment obligations under any assumed employment contracts and benefit plans that have been or purport to have been accelerated as a result of the commencement of any Chapter 11 Case or the consummation of any transactions contemplated by the Plan or this Confirmation Order shall be reinstated and such acceleration shall be rescinded and deemed not to have occurred.
- (c) <u>Workers' Compensation Programs</u>. Except as otherwise expressly provided in the Plan or this Confirmation Order, as of the Effective Date, the Debtors and the Reorganized Debtors shall continue to honor their obligations under: (i) all applicable workers' compensation laws in states in which the Reorganized Debtors operate; and (ii) the Debtors' written contracts, agreements, agreements of indemnity, self-insured workers' compensation

bonds, and any other policies, programs, and plans regarding or relating to workers' compensation and workers' compensation insurance. All such contracts and agreements are treated as executory contracts under the Plan and included in the defined term "Assumed Contracts and/or Leases".

18. Approval of Assumed Contracts and Leases.

- (a) <u>Court Approval</u>. The Debtors' assumption of the Assumed Contracts and/or Leases is hereby approved. The Assumed Contracts and/or Leases shall remain in full force and effect for the benefit of the Reorganized Debtors, notwithstanding any provision in such Assumed Contracts and/or Leases (including, without limitation, those described in Sections 365(b), (c), (e) and (f) of the Bankruptcy Code) or under applicable non-bankruptcy law that purports to (a) terminate, modify, or restrict, or permit a party other than the applicable Debtor party to terminate, modify or restrict, such contract or lease or the applicable Debtor party's rights, benefits and privileges thereunder; (b) create or impose, or permit a party other than the applicable Debtor party to create or impose, any additional duties, obligations, penalties, default rates of interest or payments (monetary and non-monetary) upon the applicable Debtor party, in either case as a result of or in connection with (i) the filing of a petition for relief under Chapter 11 of the Bankruptcy Code by the applicable Debtor party or (ii) the applicable Debtor party's insolvency or financial condition at any time before its applicable Chapter 11 Case is closed, and/or (c) prohibit, condition, or restrict assignment or transfer of such contract or lease by the Debtors.
- (b) <u>Cure Disputes</u>. The amounts, if any, due by the Debtors pursuant to each Assumed Contract and/or Lease that is in default shall be satisfied by payment of such amount in Cash on the Effective Date or on such other terms as the parties to such Assumed Contract

and/or Lease may otherwise agree in writing. In the event of a dispute regarding the amount and timing of any cure payments, the Debtors and applicable non-Debtor parties shall promptly confer after entry of this Confirmation Order to attempt to resolve any such dispute consensually without further order of this Court. In the event such dispute cannot be resolved consensually by the applicable parties, then the Debtors shall, within thirty (30) days after the Effective Date, file a notice of dispute with this Court (and promptly serve such notice on the applicable landlord) and such dispute shall be set for a status conference at the next scheduled omnibus hearing in these Chapter 11 Cases, with subsequent evidentiary hearings to be established by this Court as and if necessary. The payments, if any, or other actions, if any, that this Court determines the Debtors are required to pay or otherwise perform to assume the applicable Assumed Contract and/or Lease pursuant to Section 365(b)(1) of the Bankruptcy Code shall be promptly paid or undertaken as required by Final Order resolving the applicable dispute (and the undisputed portion of any cure claim shall be paid on the later of the Effective Date or the date required under the applicable Assumed Contract and/or Lease). Unless a dispute is otherwise litigated to Final Order, nothing in this Confirmation Order, the Plan or the assumption of the Assumed Contracts and/or Leases shall, or shall be deemed to, operate as a discharge, release, injunction or waiver of any Claims or Causes of Action that any of the non-Debtor parties may have against the Debtors (or that the Debtors may have against the non-Debtor parties) under the applicable Assumed Contract and/or Lease (whether such Claims or Causes of Action arose or accrued prior to or after the applicable assumption date, were known or unknown, or relate to setoff or recoupment), and the Debtors and Reorganized Debtors reserve all of their respective rights and defenses related thereto (and the non-Debtor parties reserve all of their respective rights and defenses related to any Claims or Causes of Action that the Debtors may have against them).

Attached hereto as <u>Exhibit 3</u> is a non-exclusive list of the unexpired leases of non-residential real property that the Debtors assumed on the Effective Date.

- 19. Approval of Rejected Contracts and Leases. All of the executory contracts and unexpired leases of the Debtors that are identified on Exhibit E-1 or Exhibit E-2 to the Plan, or that are otherwise rejected pursuant to the terms of the Plan or this Confirmation Order, (collectively, the "Rejected Contracts and Leases") are rejected by the Debtors and such rejection is hereby approved by this Court effective as of the Confirmation Date (the "Rejection Date"); provided that, with respect to the rejection of any unexpired lease of non-residential real property, such Rejection Date shall occur on the later of the Confirmation Date or the date on which the Debtors have vacated the applicable leased premises. All proofs of claim with respect to Claims arising from or in connection with the Rejected Contracts and Leases, if any, must be filed with this Court within thirty (30) days after the applicable Rejection Date. Any and all Claims arising from or in connection with the rejection of the Rejected Contracts and Leases not filed within such time will be forever barred from assertion against the Debtors or Reorganized Debtors, their Estates, or property unless otherwise ordered by this Court or provided for in the Plan.
- 20. <u>General Authorizations</u>. Pursuant to Section 1142(b) of the Bankruptcy Code and in each case without further notice to, hearing before or order of this Court, act or action under applicable law, regulation, order, or rule or the vote, consent, authorization or approval of any Person, (a) the Debtors, the Reorganized Debtors and their respective officers and directors and (b) all other necessary parties are authorized and empowered to:
 - (a) take any and all actions necessary or appropriate to enter into, implement, and consummate the contracts, instruments, releases, leases, indentures, notes, and other

agreements, documents and transactions contemplated by or described in the Plan or in this Confirmation Order;

- (b) issue, as applicable, (A) the New Common Stock, (B) the New Senior Second Lien Notes, (C) subject to satisfaction of the Harbinger Investment Effective Date Condition, the New Harbinger Senior Subordinated Notes, (D) subject to satisfaction of the Backstop Rights Offering Effective Date Condition, the New Subordinated Notes, the New Junior Subordinated Notes, and the Rights Offering Senior Subordinated Notes, and (E) any other New Securities and Documents, and enter into the New Credit Agreement; and
- (c) perform any and all other acts that are necessary, appropriate, or required to comply with or carry out the terms and conditions of the Plan or this Confirmation Order.
- 21. Authority to Act. The Debtors, the Reorganized Debtors and their respective officers and directors are authorized and empowered pursuant to Section 303 of the Delaware General Corporation Law and other applicable corporation, limited liability company and limited partnership laws of jurisdictions in which the Reorganized Debtors are incorporated, organized or formed, to take any and all actions necessary or desirable to implement the transactions contemplated by the Plan and this Confirmation Order, in each case without any requirement of further vote, consent, approval, authorization or other action by the stockholders, security holders, officers, directors, partners, managers, members or other owners of the Reorganized Debtors or notice to, order of, or hearing before this Court. Each federal, state, and local governmental agency or department is hereby authorized and directed to accept any and all documents and instruments necessary and appropriate to consummate the Plan and the transactions contemplated thereby.
- 22. <u>Exemption From Transfer Taxes</u>. Pursuant to section 1146(a) of the Bankruptcy Code, (a) the issuance, transfer, or exchange (or deemed issuance, transfer or exchange) of notes or equity securities under the Plan or this Confirmation Order, including, without limitation, the

New Senior Second Lien Notes, the New Harbinger Subordinated Notes, the New Subordinated Notes, the New Junior Subordinated Notes, the New Common Stock, the Rights, the Rights Offering Senior Subordinated Notes and the other New Securities and Documents; (b) the creation of any mortgage, deed of trust, Lien, pledge, or other security interest; (c) the making or assignment of any lease or sublease; or (d) the making or delivery of any deed or other instrument of transfer under, in furtherance of, or in connection with, the Plan or this Confirmation Order (including, without limitation, any merger agreements, agreements of consolidation, restructuring, disposition, liquidation, or dissolution, deeds, bills of sale, and transfers of tangible property) will not be subject to, and are hereby exempted from, any stamp tax, recording tax, personal property tax, real estate transfer tax, sales tax, use tax, transaction privilege tax (including, without limitation such taxes on prime contracting and owner-builder sales), privilege taxes (including, without limitation, privilege taxes on construction contracting with regard to speculative builders and owner builders), and other similar taxes. Except as otherwise ordered by this Court, all sales, transfers, and assignments of owned and leased property approved by this Court on or prior to the Effective Date, shall be deemed to have been in furtherance of, or in connection with, the Plan. Each federal, state, commonwealth, local, foreign, or other governmental agency is hereby authorized and directed to accept any and all documents, mortgages, and instruments necessary or appropriate to effectuate, implement or consummate the transactions contemplated by the Plan and this Confirmation Order, including, without limitation, the New Credit Agreement.

23. Professional Fee Claims.

(a) <u>Professional Fees Bar Date</u>. All final fee applications for Professional Fees incurred prior to the Effective Date and for services rendered during or in connection with

the Chapter 11 Cases shall be filed with this Court no later than the Business Day that is sixty (60) days after the Effective Date, or such other date as approved by order of this Court (the "Professional Fees Bar Date"); provided, however, that the reasonable fees and expenses incurred on or after the Petition Date by the Prepetition Noteholders Committee Professionals pursuant to agreements with the Debtors entered into prior to, on, or subsequent to the Petition Date, or by the Prepetition Senior Notes Indenture Trustee or the Prepetition Senior Subordinated Notes Indenture Trustee, shall be paid by the Debtors or Reorganized Debtors as Administrative Claims in the ordinary course of the Debtors' business, without application by or on behalf of any such parties to this Court, and without notice and a hearing, unless specifically required by this Court. For the avoidance of doubt, pursuant to the terms of the Investment Agreement, the reasonable fees and expenses incurred by the New Investors pursuant to the Investment Agreement shall be payable on the Effective Date without application by or on behalf of the New Investors to this Court and without notice and a hearing in accordance with the terms of the Investment Agreement. If the Debtors or Reorganized Debtors and any such applicable party cannot agree on the amount of fees and expenses to be paid to such party, the reasonableness of any such fees and expenses shall be determined by this Court.

(b) <u>Service of Final Fee Applications</u>. All final fee applications of Professionals shall be filed with this Court and actually served on or prior to the Professional Fees Bar Date upon the following parties (collectively, the "<u>Notice Parties</u>") (i) Latham & Watkins LLP, Counsel for the Debtors, Sears Tower, Suite 5800, 233 S. Wacker Drive, Chicago, IL 60606, Attn: David S. Heller; (ii) Office of the United States Trustee for the Southern District of New York, 33 Whitehall Street, 21st Floor, New York, NY 10004, Attn: Tracey Hope Davis; (iii) Akin Gump Strauss Hauer & Feld LLP, Counsel for the Prepetition

Noteholders Committee, 590 Madison Avenue, New York, NY 10022, Attn: David H. Botter and (iv) Kasowitz, Benson, Torres & Friedman LLP, Counsel for the New Investors, 1633 Broadway, New York, New York 10019, Attn: Andrew K. Glenn.

- Objections to and Hearing to Approve Final Fee Applications. (c) objection to any final fee application shall be filed with this Court, together with proof of service thereof, and served upon the applicable Professional and the other Notice Parties, so as to be actually received not later than 4:00 p.m. (prevailing Eastern Time) on the date that is twenty (20) days after such final fee application is filed with this Court and served upon the Notice Parties (the "Professional Fees Objection Deadline"). Only those objections made in writing and timely filed and received by the Professional Fees Objection Deadline will be considered by this Court. If no objection to a final fee application is timely filed and served in accordance with the procedures set forth herein, then this Court may enter a final order approving such uncontested final fee application without further notice and the Reorganized Debtors may pay the amounts described in such uncontested final fee application (or if any final fee application is the subject of an objection, the Reorganized Debtors may pay the undisputed amounts described in such final fee application). The hearing to consider approval of the final fee applications, if necessary, will be held as soon as reasonably practicable after the expiration of the Professional Fees Objection Deadline and the date of such hearing will be promptly provided to the applicable Professional and Notice Parties and posted on the Debtors' restructuring website.
- 24. <u>Resolution of Disputed Claims</u>. Except as otherwise ordered by this Court, any Claim or Interest that is not an Allowed Claim or Allowed Interest, as applicable, shall be determined, resolved, or adjudicated in accordance with the terms of this Confirmation Order and the Plan, including, without limitation, Section 8.1 of the Plan. Notwithstanding anything in

the Bankruptcy Code, the Bankruptcy Rules, the local bankruptcy rules of this Court, the Plan or this Confirmation Order to the contrary, to the extent that any Holder of a Claim or Interest files a proof of claim or proof of interest with respect thereto with this Court but is not required by Section 8.1(a) of the Plan to file such proof of claim or proof of interest, then such Claim or Interest shall be deemed to be a Disputed Claim or Disputed Interest, as applicable, unless and until either (i) the Debtors or Reorganized Debtors provide written notice to the Holder of such Claim or Interest that the Debtors or Reorganized Debtors have no objection to such Claim or Interest based on their books and records or (ii) such Claim or Interest becomes an Allowed Claim or Allowed Interest, as applicable, pursuant to Final Order entered by this Court (or any other court of competent jurisdiction) after the Effective Date expressly Allowing such Disputed Claim or Disputed Interest, as applicable.

- 25. <u>No Distributions Pending Allowance</u>. Notwithstanding any other provision of the Plan or this Confirmation Order to the contrary, no payments or distributions of any kind or nature shall be made with respect to all or any portion of a Disputed Claim or Disputed Interest unless and until all objections to such Disputed Claim or Disputed Interest have been settled or withdrawn or have been determined by Final Order, and the Disputed Claim has become an Allowed Claim or the Disputed Interest has become an Allowed Interest.
- 26. Old Common Stock Cash Amount. Notwithstanding any other provision of the Plan or this Confirmation Order to the contrary, the Debtors and Reorganized Debtors shall only be obligated to satisfy Allowed 510(b) Equity Claims and Allowed Interests arising from Old Common Stock from the Old Common Stock Cash Amount reserve established pursuant to Section 8.4 of the Plan and no other asset or property of the Debtors, the Reorganized Debtors or

their respective Estates shall be required to be used or otherwise monetized to pay or otherwise fund such Allowed Claims and/or Allowed Interests.

- 27. <u>Payment of Fees</u>. All fees payable pursuant to section 1930 of title 28, United States Code shall be paid on, or as soon as reasonably practicable after, the Effective Date.
- 28. Discharge of the Debtors. To the fullest extent provided under section 1141(d)(1)(A) and other applicable provisions of the Bankruptcy Code, except as otherwise expressly provided by the Plan or this Confirmation Order, all consideration distributed under the Plan shall be in exchange for, and in complete satisfaction, settlement, discharge, and release of, all Claims of any kind or nature whatsoever against the Debtors, the Estates or any of their respective assets or properties, and regardless of whether any property shall have been distributed or retained pursuant to the Plan on account of such Claims. Except as otherwise expressly provided by the Plan or this Confirmation Order, upon the Effective Date, the Debtors, and each of them, shall be deemed discharged and released under and to the fullest extent provided under section 1141(d)(1)(A) of the Bankruptcy Code from any and all Claims of any kind or nature whatsoever, including, but not limited to, demands and liabilities that arose before the Confirmation Date, and all debts of the kind specified in section 502(g), 502(h), or 502(i) of the Bankruptcy Code, whether or not (i) a proof of the claim based on such debt is filed or deemed filed under section 501 of the Bankruptcy Code; (ii) such claim is allowed under section 502 of the Bankruptcy Code; or (iii) the holder of such claim has accepted or rejected the Plan.
- 29. <u>Termination of Interests</u>. On the Effective Date, except as otherwise provided in the Plan or this Confirmation Order, the Old Common Stock and the Old Unexercised Equity Interests in Bally shall be terminated, cancelled and extinguished. Notwithstanding the foregoing, on the Effective Date, the Old Affiliate Interests shall remain effective and

outstanding and be owned and held by the same applicable Person(s) that held and/or owned such Interests immediately prior to the Effective Date. Each Affiliate Debtor shall continue to be governed by the terms and conditions of its applicable organizational documents as in effect immediately prior to the Effective Date, as amended or modified by the Plan or this Confirmation Order.

- 30. Notice of Confirmed Plan. In accordance with Bankruptcy Rules 2002(f)(7), 2002(k) and 3020(c), as soon as reasonably practicable after the Confirmation Date, the Debtors shall serve notice of the entry of this Confirmation Order, substantially in the form annexed hereto as Exhibit 4 (the "Notice of Confirmed Plan"), by first-class mail, postage prepaid on all known creditors, equity security holders, and other parties in interest in these Chapter 11 Cases; provided, however, that such notice need not be given or served under or pursuant to the Bankruptcy Code, the Bankruptcy Rules, the local bankruptcy rules of this Court or this Confirmation Order to any Person or Entity to whom the Debtors mailed a notice of the Confirmation Hearing, but received such notice returned marked "undeliverable as addressed," "moved-left no forwarding address" or "forwarding order expired," or similar reason, unless the Debtors have been informed in writing by such Person or Entity of that Person's or Entity's new mailing address. The notice described herein is adequate and appropriate under the particular circumstances and no other or further notice is necessary or required.
- 31. <u>Publication/Notice</u>. As soon as reasonably practicable after the Confirmation Date, the Debtors shall cause the Notice of Confirmed Plan to be published one time in <u>The Wall Street Journal</u>. The Debtors may, but are not required to, publish the Notice of Confirmed Plan in such other papers or publications as they deem appropriate.

32. Limited Substantive Consolidation. In accordance with Section 5.1 of the Plan, the Debtors are substantively consolidated solely for the limited purposes of treating Class 5 Claims and Class 6-B-2 Claims, including, without limitation, for voting, confirmation and distribution purposes. Notwithstanding anything in the Plan or this Confirmation Order to the contrary, the Debtors are not substantively consolidated with respect to the other Classes of Claims or Interests set forth in the Plan, or for any other purpose. Accordingly, for voting, confirmation and distribution purposes, (i) any obligation of any Debtor and all guarantees with respect to Class 5 Claims and Class 6-B-2 Claims thereof executed by one or more of the other Debtors shall be treated as a single obligation and any obligation of two or more Debtors, and all multiple Impaired Claims against such entities on account of such joint obligations, shall be treated and Allowed only as a single Impaired Claim against the consolidated Debtors, and (ii) each Class 5 Claim and Class 6-B-2 Claim filed or to be filed against any Debtor shall be deemed filed against the consolidated Debtors and shall be deemed a single Class 5 Claim and Class 6-B-2 Claim, as applicable, against and a single obligation of the consolidated Debtors. Except as set forth in this paragraph, such limited substantive consolidation shall not and shall not be deemed to (other than for purposes related to the Plan): (i) affect the legal and corporate structures of the Reorganized Debtors, subject to the right of the Debtors or Reorganized Debtors to effect Restructuring Transactions as provided in Section 5.2 of the Plan, (ii) cause any Debtor to be liable for any Impaired Claim or Unimpaired Claim under the Plan for which it otherwise is not liable, and the liability for any such Claim shall not be affected by such substantive consolidation, (iii) affect Intercompany Claims of Debtors against Debtors, (iv) modify, affect or otherwise alter the Old Affiliate Interests, (v) affect any obligations under any leases or contracts assumed under the Plan or this Confirmation Order or otherwise arising subsequent to the filing

of the Chapter 11 Cases, or (vi) affect any obligations to pay quarterly fees to the United States Trustee. From and after the Effective Date, each of the Reorganized Debtors will be deemed a separate and distinct entity, properly capitalized, vested with all of the assets of such Debtor as they existed immediately prior to the Effective Date and having the liabilities and obligations provided for under the Plan.

- 33. No Liability for Protected Parties. Based on the factual findings described in this Confirmation Order, the Protected Parties are not, and on account of or with respect to the offer or issuance of any security under the Original Plan or Plan, and/or solicitation of votes on the Original Plan, will not be, liable at any time for any violation of any applicable law, rule, or regulation governing the solicitation of acceptances or rejections of the Original Plan or the offer and issuance of any securities under the Original Plan or Plan, including pursuant to the Subscription and Backstop Purchase Agreement and/or the Investment Agreement. The Protected Parties have solicited votes on the Original Plan in good faith and in compliance with the applicable provisions of the Bankruptcy Code, the Bankruptcy Rules, the local bankruptcy rules of this Court and the Prepack Guidelines and all other applicable rules, laws, and regulations and are, therefore, entitled to, and are hereby granted, the protections afforded by Section 1125(e) of the Bankruptcy Code and the exculpation, release and limitation of liability provisions set forth in Article Ten of the Plan.
- 34. <u>Pending Litigation</u>. Except as set forth in the proviso of this sentence, and subject to the occurrence of the Effective Date, nothing in the Plan or this Confirmation Order shall impair the respective rights, if any, of any of the Debtors (or the Reorganized Debtors) or non-Debtor Persons or Entities (other than Holders of Rejection Claims and/or 510(b) Equity Claims) that are party to any action pending in any federal or state court as of the Petition Date to proceed

with such action in such federal or state court, to the extent permitted by, and subject to the provisions of, applicable non-bankruptcy law, without serving the Debtors the writing referred to in Section 8.1(b) of the Plan or filing any pleading with this Court; provided however that (i) pursuant to Section 8.1(b) of the Plan, the Debtors and the Reorganized Debtors reserve their rights to request that any such action be removed to, and/or determined by, this Court, (ii) the non-Debtor parties to such action reserve all of their respective rights to oppose any such request, and (iii) the Debtors, the Reorganized Debtors and the non-Debtor parties to such action expressly reserve their respective rights, if any, to arbitration of any and all Claims asserted or assertable in such action (including, in the case of a putative class action, any rights to arbitration of the underlying Claims on a non-class and non-collective basis); provided further that the Debtors and Reorganized Debtors shall have until ninety (90) days after the Effective Date (or such later date as approved by this Court for cause shown after notice and hearing) to file with this Court an objection to the alleged Claims arising from such pending action and notice requesting that any such pending action be removed to, and/or determined by, this Court, and such removal deadline is hereby extended and approved pursuant to Bankruptcy Rule 9027.

- 35. <u>Substantial Consummation</u>. "Substantial Consummation" of the Plan, as defined in 11 U.S.C. § 1101(2), shall be deemed to occur on the Effective Date.
- 36. <u>Estimation Proceedings</u>. Any and all rights of the Debtors and Reorganized Debtors under Section 502(c) of the Bankruptcy Code are reserved.
- 37. Other Rights. Any and all rights of the Debtors and Reorganized Debtors under Section 502(e) of the Bankruptcy Code are reserved.
- 38. <u>Reversal or Modification of Confirmation Order</u>. Except as otherwise provided in this Confirmation Order, if any or all of the provisions of this Confirmation Order are hereafter

reversed, modified, vacated or stayed by subsequent order of this Court, or any other court of competent jurisdiction, such reversal, stay, modification or vacatur shall not affect the validity or enforceability of any act, obligation, indebtedness, liability, priority or Lien incurred or undertaken by the Debtors or the Reorganized Debtors, as applicable, prior to the date that the Debtors received actual written notice of the effective date of such reversal, stay, modification or vacatur. Notwithstanding any such reversal, stay, modification or vacatur of this Confirmation Order, any such act or obligation incurred or undertaken pursuant to, or in reliance on, this Confirmation Order prior to the date that the Debtors received actual written notice of the effective date of such reversal, stay, modification or vacatur shall be governed in all respects by the provisions of this Confirmation Order and the Plan, or any amendments or modifications thereto, in effect prior to the date that the Debtors received such actual written notice.

39. <u>Failure to Consummate Plan</u>. If each of the conditions to consummation and the occurrence of the Effective Date contained in Section 9.2 of the Plan has not been satisfied or duly waived on or before the first Business Day that is thirty (30) days after the Confirmation Date, or such later date as shall be consented to by the Debtors and (i) the New Investors, if the Investment Agreement is then in effect, and/or (ii) the Majority Backstop Parties, if the Subscription and Backstop Purchase Agreement is then in effect, then upon motion by the Debtors and upon notice to such parties in interest as this Court may direct, this Confirmation Order shall be vacated by this Court. If this Confirmation Order is so vacated, the Plan shall be null and void in all respects, and nothing contained in the Plan or this Confirmation Order shall constitute a waiver or release of any Claims by, or against, any of the Debtors or the allowance of any Administrative Expense or Claim.

- 40. <u>Break-Up Fee and Expense Reimbursement to New Investors</u>. In the event that the Debtors' obligations under Section 8.2 of the Investment Agreement become due and owing, then the Debtors shall immediately pay the New Investors the Break-Up Fee and Expense Reimbursement owing under Section 8.2 of the Investment Agreement to the extent provided therein.
- 41. Retention of Jurisdiction. Pursuant to Sections 105(a) and 1142 of the Bankruptcy Code and notwithstanding the entry of this Confirmation Order or the occurrence of the Effective Date, this Court shall retain jurisdiction over the Chapter 11 Cases and all matters arising under, arising in, or related to, the Chapter 11 Cases and the Plan to the fullest extent permitted by law, including, among other things, jurisdiction over the matters set forth in Article Eleven of the Plan. This Court shall retain jurisdiction to hear and determine all matters arising from the implementation of this Confirmation Order.
- 42. <u>Guarantors of New Senior Second Lien Notes</u>. Attached hereto as <u>Exhibit 5</u> is the list of Affiliate Debtors that are guarantors of the New Senior Second Lien Notes.
- 43. <u>Non-Voting Equity Securities</u>. Notwithstanding anything to the contrary in the organizational documents of the Reorganized Debtors, the Reorganized Debtors shall not have the authority to issue any class of non-voting equity securities on the Effective Date.
- 44. <u>Headings</u>. The headings contained within this Confirmation Order are used for the convenience of the parties and shall not alter or affect the meaning of the text of this Confirmation Order.
- 45. <u>Objecting Former Officers and Directors</u>. Nothing in this Confirmation Order or in the Plan shall prejudice the respective rights of John W. Dwyer and Lee S. Hillman (collectively, the "**Objecting Former Officers and Directors**") and the Debtors with respect to

the classification, estimation or allowance or disallowance of any of the Claims of the Objecting Former Officers and Directors or the executory or non-executory nature of any of the Debtors' organizational documents or any contract (including any option or warrant agreement) between any Objecting Former Officers and Directors and any Debtors, including, without limitation, (a) the respective rights of the Objecting Former Officers and Directors (i) to contest the treatment of any of the Debtors' organizational documents or any of the contracts between any Debtor and any Objecting Former Officers and Directors as an executory contract or to contest the right of any Debtor to reject any such organizational documents or contracts pursuant to Section 365 of the Bankruptcy Code or (ii) to seek to have their Claims classified and Allowed as Class 3 Claims or to contest any other classification of their Claims or (b) the rights of the Debtors (i) to seek to treat any of the organizational documents of any Debtor or any contracts between any Objecting Former Officers and Directors and any Debtor as an executory contract and/or to reject any such organizational documents or contracts pursuant to Section 365 of the Bankruptcy Code, (ii) to seek to classify any Claim of any Objecting Former Officers and Directors as a Class 6-B-1 or Class 6-B-2 Claim or as a 510(b) Claim (that is part of Class 7) or (iii) to seek the disallowance of any Claim of any Objecting Former Officers and Directors pursuant to Section 502(e)(i) of the Bankruptcy Code or to seek estimation of any such Claim pursuant to Section 502(c) of the Bankruptcy Code. Notwithstanding the foregoing, the Reorganized Debtors shall be authorized to issue and file with the appropriate governmental authorities their respective amended by-laws and articles of incorporation (or similar organization documents) as contemplated by the Plan, and the foregoing reservation of rights shall not impair the validity, effectiveness or enforceability of such organizational documents; provided, however, that the issuance and filing of any amended by-laws or articles of incorporation (or similar organization

documents) by any Reorganized Debtor shall not affect any Claims of any Objecting Former Officers and Directors under any by-laws or articles of incorporation of any Debtors in effect at any time prior to the Effective Date.

46. <u>References to Plan Provisions</u>. The failure specifically to include or reference any particular provision of the Plan in this Confirmation Order shall not diminish or impair the effectiveness of such provision, it being the intent of this Court that the Plan (and the exhibits and schedules thereto) be confirmed in its entirety and incorporated herein by reference.

47. Confirmation Order Controlling. If there is any conflict or inconsistency between

the Plan and this Confirmation Order, the terms of this Confirmation Order shall control and

govern.

48. Immediate Effectiveness of this Confirmation Order. Pursuant to Bankruptcy

Rule 3020(e), the ten day stay of this Confirmation Order imposed thereby is waived and the

Debtors are hereby authorized to consummate the Plan and the transactions contemplated thereby

immediately upon the entry of this Confirmation Order upon the docket and upon the satisfaction

or waiver of the conditions set forth in Article Nine of the Plan.

Dated: September 17, 2007

New York, New York

/s/Burton R. Lifland

United States Bankruptcy Judge

43

Exhibit 1

[Plan of Reorganization]

IN THE UNITED STATES BANKRUPTCY COURT FOR THE SOUTHERN DISTRICT OF NEW YORK

In re)	Chapter 11
)	Case No. 07-12395 (BRL)
BALLY TOTAL FITNESS OF GREATER)	(Jointly Administered)
NEW YORK, INC., et al., 1)	•
)	Honorable Burton R. Lifland
Debtors.)	
)	

FIRST AMENDED JOINT PREPACKAGED CHAPTER 11 PLAN OF REORGANIZATION OF BALLY TOTAL FITNESS HOLDING CORPORATION AND ITS AFFILIATE DEBTORS

Dated: September 17, 2007

New York, New York

-

The Debtors in these proceedings are: Bally Total Fitness Holding Corporation, Bally Total Fitness Corporation, Bally ARA Corporation, Bally Fitness Franchising, Inc., Bally Franchise RSC, Inc., Bally Franchising Holdings, Inc., Bally Real Estate I LLC, Bally REFS West Hartford, LLC, Bally Sports Clubs, Inc., Bally Total Fitness Franchising, Inc., Bally Total Fitness International, Inc., Bally Total Fitness of California, Inc., Bally Total Fitness of Colorado, Inc., Bally Total Fitness of Connecticut Coast, Inc., Bally Total Fitness of Greater New York, Inc., Bally Total Fitness of Minnesota, Inc., Bally Total Fitness of Missouri, Inc., Bally Total Fitness of Philadelphia, Inc., Bally Total Fitness of Rhode Island, Inc., Bally Total Fitness of the Mid-Atlantic, Inc., Bally Total Fitness of the Midwest, Inc., Bally Total Fitness of the Southeast, Inc., Bally Total Fitness of Toledo, Inc., Bally Total Fitness of Upstate New York, Inc., BTF Cincinnati Corporation, BTF Europe Corporation, BTF Indianapolis Corporation, BTF Minneapolis Corporation, BTF/CFI, Inc., BTFCC, Inc., BTFF Corporation, Greater Philly No. 1 Holding Company, Greater Philly No. 2 Holding Company, Health & Tennis Corporation of New York, Holiday Health Clubs of the East Coast, Inc., Holiday/Southeast Holding Corp., Jack LaLanne Holding Corp., New Fitness Holding Co., Inc., Nycon Holding Co., Inc., Rhode Island Holding Company, Tidelands Holiday Health Clubs, Inc., and U.S. Health, Inc.

TABLE OF CONTENTS

ARTICLE C	ONE DEFINED TERMS AND RULES OF INTERPRETATION	1
1.1	Defined Terms	1
1.2	Exhibits and Plan Schedules	
1.3	Rules of Interpretation and Computation of Time	
ARTICLE T	WO CLASSIFICATION OF CLAIMS AND INTERESTS	21
2.1	Unclassified Claims	22
2.2	Unimpaired Classes of Claims and Interests	
2.3	Impaired Classes of Claims and Interests	
2.4	Impaired Classes of Claims and Interests	
ARTICLE T	THREE TREATMENT OF CLAIMS AND INTERESTS	
3.1	Unclassified Claims	22
3.1		
	Unimpaired Classes of Claims	
3.3 3.4	Impaired Classes of Claims and Interests	
	Unimpaired Class of Interests	
3.5	Special Provision Regarding Unimpaired Claims	29
ARTICLE F	OUR ACCEPTANCE OR REJECTION OF THE PLAN	29
4.1	Impaired Classes of Claims Entitled to Vote	29
4.2	Acceptance by an Impaired Class	
4.3	Presumed Acceptances by Unimpaired Classes	
4.4	Presumed Rejection by Certain Impaired Classes	
4.5	Confirmation Pursuant to Section 1129(b) of the Bankruptcy Code	
4.6	Elimination of Vacant Classes	
ARTICLE F	TVE MEANS FOR IMPLEMENTATION OF THE PLAN	31
5.1	Limited Substantive Consolidation for Purposes of Treating Impaired	
3.1	Claims Other Than Impaired Claims Against Only Bally	21
5.2		
	Restructuring Transactions	32
5.3	Continued Legal Existence and Vesting of Assets in the Reorganized Debtors	22
5.4	Corporate Governance, Directors, Officers, and Corporate Action	
		33
5.5	Cancellation of Notes, Instruments, Debentures, Preferred Stock and Common Stock	35
5.6	Issuance of New Securities and Related Documentation	
5.0 5.7	Exit Financing	
5.8	Sources of Cash for Plan Distributions	
J.O	DOMESON OF CASH TOLE IMILIDISH TOUR TOUR TOUR TOUR TOUR TOUR TOUR TOUR	/ /

	5.9	New Stockholders Agreement	37
	5.10	Investment Agreement	
	5.11	Old Affiliate Interests	
	5.12	Intercompany Claims	
	5.13	The Rights Offering and Subscription and Backstop Purchase Agreement	
ARTI	CLE SI	X PROVISIONS GOVERNING DISTRIBUTIONS	40
	6.1	Distributions for Claims and Interests Allowed as of the Effective Date	40
	6.2	No Postpetition Interest on Claims	41
	6.3	Distributions by Reorganized Debtors	41
	6.4	Delivery of Distributions and Undeliverable or Unclaimed Distributions	41
	6.5	Record Date for Distributions	
	6.6	Allocation of Plan Distributions Between Principal and Interest	42
	6.7	Means of Cash Payment	42
	6.8	Withholding and Reporting Requirements	42
	6.9	Setoffs	43
	6.10	Fractional Shares	43
	6.11	Surrender of Canceled Notes and Canceled Instruments of Securities	43
	6.12	Lost, Stolen, Mutilated, or Destroyed Securities	44
ARTI		EVEN TREATMENT OF EXECUTORY CONTRACTS AND	
	UNE	KPIRED LEASES	45
	7.1	Assumption of Executory Contracts and Unexpired Leases	45
	7.2	Claims Based on Rejection of Executory Contracts or Unexpired Leases	
	7.3	Cure of Defaults of Assumed Executory Contracts and Unexpired Leases	
	7.4	Compensation and Benefit Programs	
	7.5	Workers' Compensation Programs	
	,	Workers Compensation 110 grams	
ARTI	CLE EI	GHT PROCEDURES FOR RESOLVING DISPUTED, CONTINGENT	
	AND	UNLIQUIDATED CLAIMS	47
	0.1		477
		Resolution of Disputed Claims	
	8.2	No Distributions Pending Allowance	47
	8.3	Distributions on Account of Disputed Claims Once They Are Allowed and	40
	0.4	Additional Distributions on Account of Previously Allowed Claims	
	8.4	Reserve for Disputed 510(b) Equity Claims and Old Common Stock	48
ARTI	CLE N	NE CONFIRMATION AND CONSUMMATION OF THE PLAN	48
	9.1	Conditions to Confirmation	48
	9.2	Conditions to Effective Date	
	9.3	Waiver of Conditions	
	9.3 9.4	Consequences of Non-Occurrence of Effective Date	51 51

ARTICLE TE	N EFFECT OF PLAN CONFIRMATION	52
10.1	Binding Effect; Plan Binds All Holders of Claims and Interests	52
10.1	Releases and Related Injunctions.	
10.2	Discharge of Claims	
10.3	Preservation of Rights of Action; Settlement of Litigation Claims	
10.4	Exculpation and Limitation of Liability	
10.5	Injunctions	
10.0	Term of Bankruptcy Injunction or Stays	
10.7	Termination of Subordination Rights and Settlement of Related Claims	
10.0	Termination of Subordination Rights and Settlement of Related Claims	50
ARTICLE EL	EVEN RETENTION OF JURISDICTION	57
A DELCT E EV	WELVE MICCELL ANEOLIC DROVIGIONS	50
ARTICLE IV	VELVE MISCELLANEOUS PROVISIONS	39
12.1	Effectuating Documents and Further Transactions	59
12.2	Authority to Act	
12.3	Exemption from Transfer Taxes	
12.4	Bar Dates for Administrative Claims	
12.5	Payment of Statutory Fees	
12.6	Amendment or Modification of the Plan	
12.7	Severability of Plan Provisions	
12.8	Successors and Assigns	
12.9	Revocation, Withdrawal, or Non-Consummation	
12.10	Notice	
12.11	Governing Law	63
12.12	Tax Reporting and Compliance	63
12.13	Schedules	
12.14	Filing of Additional Documents	63
12.15	No Strict Construction	
12.16	Conflicts	64
12.17	Dissolution of Committee	64
	Fees and Expenses	

EXHIBITS

Exhibit A	Amended Certificate of Incorporation of Reorganized Bally
Exhibit B	Amended By-Laws of Reorganized Bally
Exhibit C	Subscription and Backstop Purchase Agreement
Exhibit D	DIP Credit Agreement
Exhibit E-1	Rejection Claims List For Bally
Exhibit E-2	Rejection Claims List For Affiliate Debtors
Exhibit F	Form of New Credit Agreement
Exhibit G-1	New Senior Second Lien Notes Indenture (if the Harbinger Investment Effective Date Condition is satisfied)
Exhibit G-2	New Senior Second Lien Notes Indenture (if the Backstop Rights Offering Condition is satisfied)
Exhibit H	Form of Rights Offering Senior Subordinated Notes Indenture, New Subordinated Notes Indenture and New Junior Subordinated Notes Indenture
Exhibit I	New Stockholders Agreement
Exhibit J	Prepetition Management Incentive Plan
Exhibit K	Registration Rights Agreement
Exhibit L	First Restructuring Support Agreement
Exhibits M-1 and M-2	Second Restructuring Support Agreements
Exhibit N	Investment Agreement
Exhibit O	New Harbinger Subordinated Notes Indenture

PLAN SCHEDULES

Plan Schedule 1.1(a) Non-Exclusive List of Litigation Claims, including derivative actions

Plan Schedule 1.1(b) Non-Exclusive List of 510(b) Equity Claims

INTRODUCTION

Bally Total Fitness Holding Corporation ("Bally") and the other above-captioned debtors and debtors-in-possession (collectively, the "Debtors") propose the following first amended prepackaged joint plan of reorganization for the resolution of the outstanding claims against and interests in the Debtors. Reference is made to the Disclosure Statement (as that term is defined herein and distributed contemporaneously herewith) for a discussion of the Debtors' history, business, properties and operations, projections for those operations, risk factors, a summary and analysis of this Plan (as that term is defined herein), and certain related matters including, among other things, certain tax matters, the securities to be issued under this Plan and the proposed substantive consolidation of the Debtors' cases for certain limited purposes. Subject to certain restrictions and requirements set forth in 11 U.S.C. § 1127 and Fed. R. Bankr. P. 3019, the Debtors reserve the right to alter, amend, modify, revoke or withdraw this Plan prior to its substantial consummation.

ARTICLE ONE DEFINED TERMS AND RULES OF INTERPRETATION

1.1 **Defined Terms**. Capitalized terms used in this Plan shall have the meanings set forth below. Any term that is not otherwise defined herein, but that is used in the Bankruptcy Code or the Bankruptcy Rules, will have the meaning given to that term in the Bankruptcy Code or the Bankruptcy Rules, as applicable.

510(b) Equity Claims means any Claim subordinated pursuant to section 510(b) of the Bankruptcy Code, including, without limitation, those claims arising from the rescission of a purchase or sale of Old Common Stock or rights relating to such Old Common Stock, or any Claim for damages arising from the purchase or sale of Old Common Stock or any Claim for reimbursement, contribution, or indemnification arising from or relating to any such Claims. A non-exclusive list of the 510(b) Equity Claims is attached to this Plan as Plan Schedule 1.1(b).

Administrative Claim means a Claim for costs and expenses of administration of the Chapter 11 Cases that are Allowed under section 503(b), 507(b), or 1114(e)(2) of the Bankruptcy Code, including, without limitation: (a) any actual and necessary costs and expenses incurred after the Petition Date of preserving the Debtors' Estates and operating the businesses of the Debtors (such as wages, salaries, and commissions for services and payments for inventory, leased equipment, and premises); (b) compensation for legal, financial, advisory, accounting, and other services and reimbursement of expenses Allowed by the Bankruptcy Court under section 327, 330, 331, 363, or 503(b) of the Bankruptcy Code to the extent incurred prior to the Effective Date; (c) all fees and charges assessed against the Debtors' Estates under section 1930, chapter 123, of title 28, United States Code; (d) any amounts and obligations owed and outstanding under the DIP Credit Agreement; (e) the Backstop Commitment Fee, to the extent payable, and, as set forth more fully in, and in accordance with, the Subscription and Backstop Purchase Agreement or the Investment Agreement, as applicable, the reasonable fees and expenses of the Backstop Parties, the New Investors and Liberation in connection with the Chapter 11 Cases and the negotiation, confirmation and implementation of this Plan and the transactions contemplated hereby; (f) the Prepetition Senior Notes Indenture Amendment Fee; (g) the Prepetition Senior Notes Indenture Trustee Fees, (h) the Prepetition Senior Subordinated Notes Indenture Trustee

Fees; and (i) the reasonable fees and expenses of the Prepetition Noteholders Committee Professionals.

Administrative Claims Bar Date means the Business Day which is thirty (30) days after the Effective Date or such other date as approved by order of the Bankruptcy Court.

Affiliate Debtor(s) means, individually or collectively, a Debtor or Debtors other than Bally, as applicable.

Allowed means, with respect to a Claim or Interest, an Allowed Claim or an Allowed Interest in a particular Class or category specified. Any reference herein to a particular Allowed Claim includes both the secured and unsecured portions of such Claim.

Allowed Claim means any Claim or portion thereof that is not a Disputed Claim and (a) that has been listed by a Debtor in its Schedules (if such Schedules are required by order of the Bankruptcy Court) as liquidated in amount and not disputed or contingent and for which no contrary proof of claim has been filed; (b) as to which no objection to allowance has been timely interposed in accordance with section 502 of the Bankruptcy Code and Bankruptcy Rule 2007 or such other applicable period of limitation fixed by the Bankruptcy Code, the Bankruptcy Rules or the Bankruptcy Court, or as to which any objection has been determined by a Final Order to the extent such objection is determined in favor of the respective holder; (c) as to which, upon the lifting of the automatic stay pursuant to section 362 of the Bankruptcy Code, the liability of the Debtors (allowance and the amount thereof) is determined by Final Order of a court of competent jurisdiction other than the Bankruptcy Court; or (d) that is expressly allowed by this Plan.

The term "Allowed Claim" shall not, for purposes of computing distributions under this Plan, include interest on such Claim from and after the Petition Date, except as provided in section 506(b) of the Bankruptcy Code or as otherwise expressly set forth in this Plan.

Allowed	_ <i>Claim</i> means an A	Allowed Cla	im of the type	described.
---------	---------------------------	-------------	----------------	------------

Allowed Interest means any Interest or portion thereof that is not a Disputed Interest and (a) as to which no objection to allowance has been timely interposed in accordance with section 502 of the Bankruptcy Code or such other applicable period of limitation fixed by the Bankruptcy Code, the Bankruptcy Rules or the Bankruptcy Court, or as to which any objection has been determined by a Final Order to the extent such objection is determined in favor of the respective holder; or (b) that is expressly allowed by this Plan.

Allowed	_ Interest	means	an A	Allowed	Interest	Oİ.	the	type	describ	ed.	,
---------	------------	-------	------	---------	----------	-----	-----	------	---------	-----	---

Amended Certificate of Incorporation and By-Laws means the amended and restated certificate of incorporation and by-laws of Reorganized Bally in substantially the form attached to this Plan as Exhibit A and Exhibit B, respectively.

Applicable Outside Date means (i) if the Section 1127(a) Order is entered and remains in full force and effect, October 15, 2007, and (ii) if the Section 1127(a) Order has not been entered or otherwise is not in full force and effect, then November 30, 2007.

Avoidance and Other Actions means any and all avoidance, recovery, subordination or other actions or remedies that may be brought by and on behalf of the Debtors or their Estates under the Bankruptcy Code or applicable non-bankruptcy law, including, without limitation, actions or remedies arising under sections 510 and 542-553 of the Bankruptcy Code.

Backstop Commitment means \$90,000,000 in the aggregate, and with respect to each Backstop Party, the amount listed on Schedule 1 to the Subscription and Backstop Purchase Agreement with respect to such Backstop Party.

Backstop Commitment Fee means the fee to be paid by Bally (and guaranteed by each of the Affiliate Debtors) to each Backstop Party in consideration for its respective Backstop Commitment, as more fully described in the Subscription and Backstop Purchase Agreement attached to this Plan as Exhibit C.

Backstop Parties means those Prepetition Senior Subordinated Noteholders that provided the Backstop Commitment, which Prepetition Senior Subordinated Noteholders are parties to the Subscription and Backstop Purchase Agreement.

Backstop Rights Offering Effective Date Condition shall have the meaning set forth in Section 9.2(d) of the Plan.

Ballot means each of the ballot forms distributed to each Holder of an Impaired Claim that is entitled to vote to accept or reject this Plan and on which the Holder is to indicate, among other things, acceptance or rejection of this Plan.

Bally means Bally Total Fitness Holding Corporation, a Delaware corporation, debtor-in-possession in these Chapter 11 Cases pending in the Bankruptcy Court.

Bankruptcy Code means title 11 of the United States Code, as now in effect or hereafter amended so as to be applicable in the Chapter 11 Cases.

Bankruptcy Court means the United States Bankruptcy Court for the Southern District of New York, or any other court with jurisdiction over the Chapter 11 Cases.

Bankruptcy Rules means the Federal Rules of Bankruptcy Procedure and the local rules of the Bankruptcy Court, as now in effect or hereafter amended.

Bar Date means, (i) with respect to the Holders of Rejection Claims, the bar date for Filing proofs of Claim in the Chapter 11 Cases as set forth in Section 7.2 of this Plan and (ii) with respect to the Holders of 510(b) Equity Claims, such date ordered by the Bankruptcy Court; provided that this clause (ii) shall only be applicable in the event that the Harbinger Investment Effective Date Condition is satisfied.

Business Day means any day, other than a Saturday, Sunday, or "legal holiday" (as defined in Bankruptcy Rule 9006(a)).

Cash means legal tender of the United States of America and equivalents thereof.

Causes of Action means, without limitation, any and all claims, causes of action, demands, rights, actions, suits, damages, injuries, remedies, obligations, liabilities, accounts, defenses, offsets, powers, privileges, licenses and franchises of any kind or character whatsoever, known, unknown, accrued or to accrue, contingent or non-contingent, matured or unmatured, suspected or unsuspected, foreseen or unforeseen, whether arising before, on or after the Petition Date, in contract or in tort, in law or in equity, or under any other theory of law, whether asserted or assertable directly or derivatively in law or equity or otherwise by way of claim, counterclaim, cross-claim, third party action, action for indemnity or contribution or otherwise, including, without limitation, the Avoidance and Other Actions.

Chapter 11 Cases means (a) when used with reference to a particular Debtor, the case under chapter 11 of the Bankruptcy Code commenced by the Debtor in the Bankruptcy Court, and (b) when used with reference to all Debtors, the cases under chapter 11 of the Bankruptcy Code commenced by the Debtors in the Bankruptcy Court.

Claim means a "claim" as defined in section 101(5) of the Bankruptcy Code.

Claims Objection Deadline means the last day for Filing objections to Rejection Claims and, in the event that the Harbinger Investment Effective Date Condition is satisfied, the 510(b) Equity Claims, which day shall be (i) the later of (a) thirty (30) days after the Effective Date or (b) sixty (60) days after the Filing of a proof of claim for, or request for payment of, such Claim, or (ii) such other date as the Bankruptcy Court may order.

Class means a category of Holders of Claims or Interests, as described in Article II hereof.

Collateral means any property or interest in property of the Debtors' Estates that is subject to a valid and enforceable Lien to secure a Claim.

Confirmation Date means the date on which the clerk of the Bankruptcy Court enters the Confirmation Order on the docket of the Bankruptcy Court.

Confirmation Hearing means the hearing held by the Bankruptcy Court pursuant to sections 105(d)(2)(B)(vi) and 1128 of the Bankruptcy Code to consider (i) approval of the Disclosure Statement under sections 1125 and 1126(b) of the Bankruptcy Code and (ii) confirmation of this Plan, as such hearing may be adjourned or continued from time to time.

Confirmation Order means the order of the Bankruptcy Court confirming this Plan pursuant to section 1129 of the Bankruptcy Code.

Creditors' Committee means the official committee of unsecured creditors of the Debtors, if any, appointed by the United States Trustee in the Chapter 11 Cases pursuant to section 1102 of the Bankruptcy Code.

Debtor(s) means, individually, any of the Debtors and, collectively, all of the above-captioned debtors and debtors-in-possession.

Debtor-Related Released Parties means, collectively, the Debtors, their Estates, the Reorganized Debtors, the New Investors and their respective Related Persons other than any Related Person of any Debtor or Reorganized Debtor who is not entitled to be indemnified, by contract or applicable law, from any Debtor or Reorganized Debtor from and against any claims, obligations, suits, judgments, damages, demands, debts, rights, Causes of Action and liabilities that may be asserted by any Holders of Claims or Holders of any Interests against such Related Person in its or his capacity as such.

DIP Agent means the administrative agent under the DIP Credit Agreement, and its successors and assigns.

DIP Credit Agreement means the debtor-in-possession secured credit agreement substantially in the form attached hereto as <u>Exhibit D</u>.

DIP Credit Agreement Obligations means any loans and other indebtedness and obligations of any or all of the Debtors to any or all of the DIP Agent and the DIP Lenders pursuant to the DIP Credit Agreement and the other DIP Credit Documents.

DIP Credit Documents means all of the agreements, documents and instruments entered into in connection with the DIP Credit Agreement.

DIP Lenders means each of the financial institutions party to the DIP Credit Agreement and identified as "Lenders" therein, and their respective successors and assigns.

DIP Lenders Claims means any and all "Obligations" as defined in the DIP Credit Agreement, including, without limitation, the DIP Credit Agreement Obligations and any and all other Claims of, and any other obligations and liabilities owed to, the DIP Agent and DIP Lenders arising from or related to the DIP Credit Agreement and any other DIP Credit Document.

DIP Lenders Liens means any security interests and Liens granted by any Debtor to the DIP Agent and/or any DIP Lender in order to secure the repayment of any DIP Lenders Claims.

DIP Loan Facility means the credit facility in the aggregate principal amount not to exceed \$292,000,000 to be provided to the Debtors during the Chapter 11 Cases pursuant to the DIP Credit Agreement if the Debtors determine a DIP Credit Agreement is necessary, the proceeds of which would be used to refinance the Prepetition Lenders Claims and for general corporate purposes.

Disbursing Agent means the Reorganized Debtors or any party designated by the Reorganized Debtors to serve as disbursing agent under this Plan. For purposes of distributions under this Plan to the Holders of Allowed Prepetition Lenders Claims, Allowed Prepetition Senior Notes Claims, and Allowed Prepetition Senior Subordinated Notes Claims and, in the event that the Harbinger Investment Effective Date Condition is satisfied, the Old Common Stock, the Prepetition Agent, the Prepetition Senior Notes Indenture Trustee, the Prepetition Senior Subordinated Notes Indenture Trustee and the transfer agent for the Old Common Stock (which transfer agent will a successor transfer agent to be retained by the Debtors prior to the

commencement of the Confirmation Hearing, and will not be the existing transfer agent), respectively, will be and shall act as the Disbursing Agent.

Disclosure Statement means that certain disclosure statement (including all exhibits and schedules thereto) dated as of the date hereof (and any amendments or supplements thereto), relating to this Plan, which was distributed by Bally on or about such date to the Prepetition Senior Noteholders and the Prepetition Senior Subordinated Noteholders existing as of the Voting Record Date in connection with the prepetition solicitation of their votes pursuant to section 1126(b) of the Bankruptcy Code.

Disputed Claim means a Claim, or any portion thereof, that (a) if the Debtors are required by order of the Bankruptcy Court to file Schedules, (i) has not been Scheduled by the Debtors or has been Scheduled at zero, or has been Scheduled as contingent, unliquidated or disputed and for which no proof of claim has been timely filed with the Bankruptcy Court or (ii) is in excess of the amount Scheduled as other than disputed, contingent or unliquidated, (b) is the subject of an objection or request for estimation filed in the Bankruptcy Court and which objection or request for estimation has not been withdrawn or overruled by a Final Order of the Bankruptcy Court, (c) is a 510(b) Equity Claim and/or (d) is otherwise disputed by any of the Debtors in accordance with applicable law, which dispute has not been withdrawn, resolved or overruled by Final Order.

Disputed Interest means an Interest, or any portion thereof, that (a) is the subject of an objection or request for estimation filed in the Bankruptcy Court and which objection or request for estimation has not been withdrawn or overruled by a Final Order of the Bankruptcy Court, and/or (b) is otherwise disputed by any of the Debtors in accordance with applicable law, which dispute has not been withdrawn, resolved or overruled by Final Order.

Distribution Record Date means the Effective Date.

Effective Date means the Business Day that this Plan becomes effective as provided in Article IX hereof.

Entity means an "entity" as defined in section 101(15) of the Bankruptcy Code.

Estate(s) means, individually, the estate of each of the Debtors and, collectively, the estates of all of the Debtors created under section 541 of the Bankruptcy Code.

Exchange Act means the Securities Exchange Act of 1934, 15 U.S.C. §§ 78a et seq., as amended.

Exhibit means an exhibit annexed to either this Plan or as an appendix to the Disclosure Statement (as such exhibits are amended, modified or otherwise supplemented from time to time).

File, Filed, or Filing means file, filed, or filing with the Bankruptcy Court or its authorized designee in the Chapter 11 Cases.

Final Order means an order of the Bankruptcy Court as to which the time to appeal, petition for certiorari, or move for reargument or rehearing has expired and as to which no appeal, petition for certiorari, or other proceedings for reargument or rehearing shall then be pending or as to which any right to appeal, petition for certiorari, reargue, or rehear shall have been waived in writing in form and substance satisfactory to the Debtors or the Reorganized Debtors, or, in the event that an appeal, writ of certiorari or reargument, or rehearing thereof has been sought, such order of the Bankruptcy Court shall have been determined by the highest court to which such order was appealed, or certiorari, reargument or rehearing shall have been denied and the time to take any further appeal, petition for certiorari, or move for reargument or rehearing shall have expired; provided, however, that the possibility that a motion under Rule 60 of the Federal Rules of Civil Procedure, or any analogous rule under the Bankruptcy Rules, may be filed with respect to such order shall not preclude such order from being a Final Order.

First Restructuring Support Agreement means that certain Restructuring Support Agreement, dated as of June 15, 2007, between Bally, the Prepetition Senior Subordinated Noteholders party thereto and the Prepetition Senior Noteholders party thereto, a copy of which is attached hereto as Exhibit L.

Harbinger Investment means the investment to be made in Cash by the New Investors in Reorganized Bally in the amount of \$233.6 million on or before September 30, 2007, in accordance with the provisions of the Investment Agreement and this Plan, and in exchange for 100% of the New Common Stock of Reorganized Bally, all as more fully described in the Investment Agreement.

Harbinger Investment Effective Date Condition shall have the meaning set forth in Section 9.2(d) of the Plan.

Holder means a Person or an Entity holding a Claim or Interest and, with respect to the Prepetition Senior Notes Claims, the Prepetition Senior Subordinated Notes Claims and the Old Common Stock, the beneficial holder thereof as of the applicable date of determination or any authorized agent of such Person or Entity who has completed and executed a Ballot or on whose behalf a Master Ballot has been completed and executed in accordance with the voting instructions that are attached to the Ballot or Master Ballot, as applicable.

Impaired means, when used in reference to a Claim or Interest, a Claim or Interest that is impaired within the meaning of section 1124 of the Bankruptcy Code.

Impaired Unsecured Claim means any (a) Prepetition Senior Notes Claim, (b) Prepetition Senior Subordinated Notes Claim, or (c) Rejection Claim.

Initial Distribution Date means the date as determined by the Reorganized Debtors upon which the initial distributions of property under this Plan will be made to Holders of Allowed Claims and, in the event that the Harbinger Investment Effective Date Condition is satisfied, Allowed Interests, which date shall be as soon as practicable after the Effective Date unless otherwise extended by order of the Bankruptcy Court; provided that, in the event that the Harbinger Investment Effective Date Condition is satisfied, then with respect to 510(b) Equity

Claims and Old Common Stock, the Initial Distribution Date shall not occur before the expiration of the Bar Date.

Intercompany Claim means (a) any account reflecting intercompany book entries by one Debtor with respect to any other Debtor or (b) any Claim that is not reflected in such book entries and is held by a Debtor against any other Debtor.

Interest means the legal, equitable, contractual, and other rights of the Holders of any ownership interest in any Debtor existing as of the Petition Date, including, without limitation, the Old Common Stock, which shall also include the rights of any Person or Entity to purchase or demand the issuance of any of the foregoing and shall include: (a) conversion, exchange, voting, participation, and dividend rights; (b) liquidation preferences; (c) options, warrants, and put rights; and (d) share-appreciation rights.

Investment Agreement means that certain Investment Agreement dated August 15, 2007 by and among Bally and the New Investors, a copy of which is attached to this Plan as <u>Exhibit N</u>. Pursuant to the Investment Agreement, the New Investors will acquire 100% of the New Common Stock issued on the Effective Date from Reorganized Bally in exchange for an aggregate purchase price of approximately \$233.6 million, all as more fully described in the Investment Agreement.

IRC means the Internal Revenue Code of 1986, as amended.

IRS means the Internal Revenue Service of the United States of America.

Liberation means Liberation Investment Group and its affiliates.

Lien means a "lien" as defined in section 101(37) of the Bankruptcy Code, and, with respect to any asset, includes, without limitation, any mortgage, lien, pledge, charge, security interest or other encumbrance of any kind, or any other type of preferential arrangement that has the practical effect of creating a security interest, in respect of such asset.

Litigation Claims means the claims, rights of action, suits or proceedings, whether in law or in equity, whether known or unknown, that any Debtor or Estate may hold against any Person or Entity, including, without limitation, the Causes of Action of any Debtor. A non-exclusive list of the Litigation Claims held by the Debtors as of the Petition Date is attached hereto as Plan Schedule 1.1(a), which shall be deemed to include the derivative actions filed against the Debtors listed on such Plan Schedule and any Causes of Action against any Person or Entity listed on Exhibit E-1 or Exhibit E-2 to this Plan.

Majority Backstop Parties means Backstop Parties holding in excess of 50% of the principal amount of Prepetition Senior Subordinated Notes held by all of the Backstop Parties.

Master Ballot means the ballot distributed to holders of record of the Prepetition Senior Notes and Prepetition Senior Subordinated Notes to record the votes of the beneficial holders of the Prepetition Senior Notes and Prepetition Senior Subordinated Notes, respectively, as of the Voting Record Date.

New Agent means the administrative agent under the New Credit Agreement, and its successors and assigns.

New Common Stock means the shares of common stock of Reorganized Bally authorized to be issued pursuant to this Plan and the Amended Certificate of Incorporation and By-Laws.

New Credit Agreement means that certain secured credit agreement between Reorganized Bally, as borrower, those entities identified as "Guarantors" in the New Credit Agreement, New Agent and New Lenders (as amended, modified, or supplemented from time to time, with the prior written consent of the Majority Backstop Parties (if the Subscription and Backstop Purchase Agreement is then in effect) and/or the New Investors (if the Investment Agreement is then in effect), in each case, which consent shall not be unreasonably withheld, conditioned or delayed), in an aggregate principal amount of at least \$292 million, and substantially in the form attached hereto as Exhibit F.

New Harbinger Subordinated Notes means, in the event that the Harbinger Investment Effective Date Condition is satisfied, the New Senior Subordinated Notes due 2013 to be issued by Reorganized Bally under the New Harbinger Subordinated Notes Indenture in the aggregate principal amount of \$200 million. The New Harbinger Subordinated Notes shall be subordinate in priority of payment to the New Senior Second Lien Notes or any refinancing thereof.

New Harbinger Subordinated Notes Indenture means, in the event that the Harbinger Investment Effective Date Condition is satisfied, that certain Amended and Restated Indenture, to be dated as of the Effective Date, among Reorganized Bally, as issuer, and the New Harbinger Subordinated Notes Indenture Trustee, relating to the New Harbinger Subordinated Notes, substantially in the form of Exhibit O attached to this Plan.

New Harbinger Subordinated Notes Indenture Trustee means, in the event that the Harbinger Investment Effective Date Condition is satisfied, HSBC Bank USA, National Association, as the indenture trustee under the New Harbinger Subordinated Notes Indenture, and its successors and assigns.

New Investors means Harbinger Capital Partners Master Fund I, Ltd. and Harbinger Capital Partners Special Situations Fund L.P.

New Junior Subordinated Notes means, in the event that the Backstop Rights Offering Effective Date Condition is satisfied, the New Junior Subordinated Notes due 2013 to be issued by Reorganized Bally under the New Junior Subordinated Notes Indenture in the aggregate principal amount not to exceed 21.7% of the Allowed Prepetition Senior Subordinated Notes Claims plus 21.7% of the Allowed Class 6-B-1 Claims. The New Junior Subordinated Notes shall be subordinate in priority of payment to the New Senior Second Lien Notes, the Rights Offering Senior Subordinated Notes and the New Subordinated Notes.

New Junior Subordinated Notes Indenture means, in the event that the Backstop Rights Offering Effective Date Condition is satisfied, that certain Amended and Restated Indenture, to be dated as of the Effective Date, among Reorganized Bally, as issuer, and the New

Junior Subordinated Notes Indenture Trustee, relating to the New Junior Subordinated Notes, substantially in the form of Exhibit H attached to this Plan.

New Junior Subordinated Notes Indenture Trustee means, in the event that the Backstop Rights Offering Effective Date Condition is satisfied, HSBC Bank USA, National Association, as the indenture trustee under the New Junior Subordinated Notes Indenture, and its successors and assigns.

New Lenders means each of the financial institutions party to the New Credit Agreement and identified as "Lenders" therein, and their respective successors and assigns.

New Securities and Documents has the meaning given in Section 5.6 of this Plan.

New Senior Second Lien Notes means the New Senior Second Lien Notes due 2011 to be issued by Reorganized Bally under the New Senior Second Lien Notes Indenture in the aggregate principal amount of \$247,337,500. The New Senior Second Lien Notes shall be senior in priority of payment over the Rights Offering Senior Subordinated Notes (if any), the New Subordinated Notes (if any), the New Junior Subordinated Notes (if any) and the New Harbinger Subordinated Notes (if any). The New Senior Second Lien Notes will be secured by Liens on the same assets that secure the obligations under the New Credit Agreement, which Liens shall be subordinate in priority to the Liens securing the obligations under the New Credit Agreement.

New Senior Second Lien Notes Indenture means that certain Indenture, to be dated as of the Effective Date, among Reorganized Bally, as issuer, certain Affiliate Debtors, as guarantors, and the New Senior Second Lien Notes Indenture Trustee, relating to the New Senior Second Lien Notes, substantially in the forms of Exhibits G-1 (if the Harbinger Investment Effective Date Condition is satisfied) and G-2 (if the Backstop Rights Offering Condition is satisfied) attached to this Plan.

New Senior Second Lien Notes Indenture Trustee means U.S. Bank National Association, as the indenture trustee under the New Senior Second Lien Notes Indenture, and its successors and assigns.

New Stockholders Agreement means, in the event that the Backstop Rights Offering Effective Date Condition is satisfied, that certain stockholders agreement to be entered into on the Effective Date by and among Reorganized Bally and all holders of the New Common Stock, substantially in the form of Exhibit I attached to this Plan.

New Subordinated Notes means, in the event that the Backstop Rights Offering Effective Date Condition is satisfied, the New Senior Subordinated Notes due 2013 to be issued by Reorganized Bally under the New Subordinated Notes Indenture in the aggregate principal amount not to exceed 24.8% of the Allowed Prepetition Senior Subordinated Notes Claims plus 24.8% of the Allowed Class 6-B-1 Claims. The New Subordinated Notes shall be subordinate in priority of payment to the New Senior Second Lien Notes and the Rights Offering Senior Subordinated Notes, and senior in priority of payment to the New Junior Subordinated Notes.

New Subordinated Notes Indenture means, in the event that the Backstop Rights Offering Effective Date Condition is satisfied, that certain Amended and Restated Indenture, to be dated as of the Effective Date, among Reorganized Bally, as issuer, and the New Subordinated Notes Indenture Trustee, relating to the New Subordinated Notes, substantially in the form of Exhibit H attached to this Plan.

New Subordinated Notes Indenture Trustee means, in the event that the Backstop Rights Offering Effective Date Condition is satisfied, HSBC Bank USA, National Association, as the indenture trustee under the New Subordinated Notes Indenture, and its successors and assigns.

Non-Tax Priority Claim means a Claim, other than an Administrative Claim or Priority Tax Claim, that is entitled to priority in payment pursuant to section 507(a) of the Bankruptcy Code.

Old Affiliate Interests means, collectively, the shares of stock, whether common or preferred, general and limited partnership interests, or member or other ownership interests of the Affiliate Debtors, as applicable, issued and outstanding as of the Petition Date, and all options, warrants, calls, rights, puts, awards, commitments, or any other agreements of any character to acquire such stock or interests.

Old Common Stock means the common stock of Bally that is outstanding immediately prior to the Petition Date, including, without limitation, treasury stock and any other such interests that are authorized to be issued but have not been issued and the 510(b) Equity Claims.

Old Common Stock Cash Amount means, in the event that the Harbinger Investment Effective Date Condition is satisfied, \$16.5 million.

Old Common Stock Cash Portion means, in the event that the Harbinger Investment Effective Date Condition is satisfied, that portion of the Old Common Stock Cash Amount reserved solely for the Holders of Allowed Interests represented by Old Common Stock, which portion equals the Old Common Stock Cash Amount multiplied by the fraction the numerator of which equals the Old Common Stock Cash Amount and the denominator of which equals the sum of (x) the Old Common Stock Cash Amount plus (y) the aggregate face amount of all Allowed 510(b) Equity Claims; provided, however, that for purposes of making interim distributions hereunder to holders of Allowed Class 7 Claims, clause (y) of the denominator shall be the aggregate maximum amount of all 510(b) Equity Claims asserted by the holders thereof.

Old Common Stock Remaining Cash Portion means, in the event that the Harbinger Investment Effective Date Condition is satisfied, that portion of the Old Common Stock Cash Amount reserved solely for the Holders of Allowed 510(b) Equity Claims, which portion equals the Old Common Stock Cash Amount less the Old Common Stock Cash Portion.

Old Unexercised Equity Interests means all unexercised options, warrants, calls, rights, puts, awards, commitments, or any other agreements of any character to acquire Old Common Stock.

Other Released Parties means, collectively, the Released Parties other than the Debtor-Related Released Parties.

Other Secured Claim means a Secured Claim other than an Administrative Claim, Secured Tax Claim or Prepetition Lenders Claim.

Person means any natural person, corporation, general or limited partnership, limited liability company, firm, trust, association, government, governmental agency or other Entity, whether acting in an individual, fiduciary or other capacity.

Petition Date means the date on which the Debtors file their petitions for relief commencing the Chapter 11 Cases.

Plan means this first amended joint prepackaged chapter 11 plan of reorganization, including the Exhibits and Plan Schedules and all supplements, appendices, and schedules thereto, either in its present form or as the same may be altered, amended, modified or otherwise supplemented from time to time.

Plan-Related Claim means any and all claims against any Debtor or any other Released Party as to which such person or entity is exculpated pursuant to Section 1125(e) of the Bankruptcy Code, including, without limitation, to the maximum extent provided by Section 1125(e) of the Bankruptcy Code, any and all claims against any Debtor or other Released Party for violating any provisions of the Securities Act of 1933 or any State or local law requiring registration for offer or sale of a security or registration or licensing of an issuer of, underwriter of, or broker or dealer in, a security, in connection with any or all of the prepetition or postpetition solicitation of acceptances of the Plan, the prepetition or postpetition solicitation of subscriptions with respect to the Rights Offering, the Subscription and Backstop Purchase Agreement, the Investment Agreement (if the Harbinger Investment Effective Date Condition is satisfied), the First Restructuring Support Agreement, the Second Restructuring Support Agreement, and the acts taken under any of the foregoing.

Plan Schedule means a schedule annexed to either this Plan or as an appendix to the Disclosure Statement (as amended, modified or otherwise supplemented from time to time).

Prepetition Agent means JPMorgan Chase Bank, N.A., as the administrative agent under the Prepetition Credit Agreement, and its successors and assigns.

Prepetition Credit Agreement means that certain amended and restated credit agreement, dated as of October 16, 2006 (as amended, modified, or supplemented from time to time), by and among Bally, as borrower, certain Affiliate Debtors as Guarantors, Prepetition Agent and Prepetition Lenders.

Prepetition Credit Agreement Notes means, collectively, the notes evidencing the Prepetition Credit Agreement Obligations that were issued by Bally prior to the Petition Date to the Prepetition Agent and Prepetition Lenders under the Prepetition Credit Agreement.

Prepetition Credit Agreement Obligations means the "Obligations" as described and defined in the Prepetition Credit Agreement.

Prepetition Credit Documents means the "Credit Documents" as described and defined in the Prepetition Credit Agreement.

Prepetition Lenders means each of the financial institutions party to the Prepetition Credit Agreement and identified as "Lenders" therein, and their respective successors and assigns.

Prepetition Lenders Claims means any and all Claims of, and any other obligations and liabilities owed to, the Prepetition Agent and Prepetition Lenders arising from or related to the Prepetition Credit Agreement and any other Prepetition Credit Document, including, without limitation, the Prepetition Credit Agreement Obligations, which Claims shall be deemed, pursuant to sections 502 and 506 of the Bankruptcy Code, Allowed secured claims pursuant to this Plan in the aggregate principal amount of \$262,400,000, plus interest and reasonable fees, costs and expenses that have accrued but remain unpaid as of the Effective Date pursuant to the Prepetition Credit Documents (which final aggregate amount shall be set forth in the Confirmation Order).

Prepetition Lenders Liens means the security interests and Liens granted by any Debtor to the Prepetition Agent and/or any Prepetition Lender in order to secure the repayment of any Prepetition Lenders Claims.

Prepetition Management Incentive Plan means that certain management incentive plan approved by the Board of Directors of Bally on or about May 28, 2007, as may be amended from time to time, and described in <u>Exhibit J</u> attached hereto.

Prepetition Noteholders Committee means the ad hoc committee of Holders of Prepetition Senior Notes and Prepetition Senior Subordinated Notes in existence as of the Petition Date.

Prepetition Noteholders Committee Professionals means Akin Gump Strauss Hauer & Feld and Houlihan Lokey Howard & Zukin Capital, Inc.

Prepetition Senior Noteholders means the Holders of the Prepetition Senior Notes Claims.

Prepetition Senior Notes means the 10.5% Senior Notes due 2011 issued by Bally prior to the Petition Date under the Prepetition Senior Notes Indenture.

Prepetition Senior Notes Claims means any and all Claims, obligations and liabilities arising from or related to the Prepetition Senior Notes and/or Prepetition Senior Notes Indenture, including, without limitation, the "Indenture Obligations" (as such term is defined in the Prepetition Senior Notes Indenture), which Claims shall be Allowed in the aggregate principal amount of \$235,000,000, plus interest and reasonable fees, costs and expenses that have accrued but remain unpaid as of the Petition Date pursuant to the Prepetition Senior Notes Indenture (which final aggregate amount shall be set forth in the Confirmation Order).

Prepetition Senior Notes Indenture means that certain Indenture, dated as of July 2, 2003, among Bally, as issuer, certain Affiliate Debtors, as guarantors, and the Prepetition

Senior Notes Indenture Trustee, relating to the Prepetition Senior Notes (as amended, modified, or supplemented from time to time).

Prepetition Senior Notes Indenture Amendment Fee means that certain amendment fee in the aggregate amount of \$4,700,000, which fee shall be (i) in consideration of the amendments to the Prepetition Senior Notes Indenture as reflected in the New Senior Second Lien Notes Indenture; and (ii) paid to the Prepetition Senior Notes Indenture Trustee for the benefit of all Prepetition Senior Noteholders and shared on a Pro Rata basis with such holders.

Prepetition Senior Notes Indenture Trustee means U.S. Bank National Association, as the indenture trustee under the Prepetition Senior Notes Indenture, and its successors and assigns.

Prepetition Senior Notes Indenture Trustee Fees means the reasonable fees and reasonable unpaid out-of-pocket costs and expenses incurred by the Prepetition Senior Notes Indenture Trustee through the Effective Date in accordance with the Prepetition Senior Note Indenture.

Prepetition Senior Subordinated Noteholders means the Holders of the Prepetition Senior Subordinated Notes Claims.

Prepetition Senior Subordinated Notes means, collectively, the 9.875% Senior Subordinated Notes due 2007, Series B, and the 9.875% Senior Subordinated Notes due 2007, Series D, issued by Bally prior to the Petition Date under the Prepetition Senior Subordinated Notes Indenture.

Prepetition Senior Subordinated Notes Claims means any and all Claims, obligations and liabilities arising from or related to the Prepetition Senior Subordinated Notes and/or Prepetition Senior Subordinated Notes Indenture, including, without limitation, the "Indenture Obligations" (as such term is defined in the Prepetition Senior Subordinated Notes Indenture), which Claims shall be Allowed in the aggregate amount (including accrued and unpaid interest) of \$323,041,667.

Prepetition Senior Subordinated Notes Indenture means, collectively, that certain Indenture, dated as of October 7, 1997, and that certain Indenture, dated as of December 16, 1998, in each case by and between Bally, as issuer, and the Prepetition Senior Subordinated Notes Indenture Trustee, relating to the Prepetition Senior Subordinated Notes (as amended, modified, or supplemented from time to time).

Prepetition Senior Subordinated Notes Indenture Trustee means HSBC Bank USA, National Association, as the indenture trustee under the Prepetition Senior Subordinated Notes Indenture, and its successors and assigns.

Prepetition Senior Subordinated Notes Indenture Trustee Fees means the reasonable fees and reasonable unpaid out-of-pocket costs and expenses incurred by the Prepetition Senior Subordinated Notes Indenture Trustee through the Effective Date in accordance with the Prepetition Senior Subordinated Note Indenture.

Priority Tax Claim means a Claim of a governmental unit of the kind specified in sections 502(i) and 507(a)(8) of the Bankruptcy Code, including a Secured Tax Claim.

Professional means (a) any professional employed in the Chapter 11 Cases pursuant to section 327 or 1103 of the Bankruptcy Code or otherwise and (b) any professional or other Entity seeking compensation or reimbursement of expenses in connection with the Chapter 11 Cases pursuant to section 503(b)(4) of the Bankruptcy Code.

Professional Fees means an Administrative Claim of a Professional for compensation for services rendered or reimbursement of costs, expenses or other charges incurred after the Petition Date and prior to and including the Effective Date (including expenses of the members of the Creditors' Committee incurred as members of the Creditors' Committee in discharge of their duties as such).

Professional Fees Bar Date means the Business Day that is sixty (60) days after the Effective Date or such other date as approved by order of the Bankruptcy Court.

Pro Rata means with respect to a distribution regarding a particular Class (or several Classes taken as a whole), the proportion that (a) the Allowed amount of a Claim or Interest in a particular Class (or several Classes taken as a whole) bears to (b) the aggregate Allowed amount of all Claims or Interests in such Class (or several Classes taken as a whole), unless this Plan provides otherwise.

Quarterly Distribution Date means the last Business Day of the month following the end of each calendar quarter after the Effective Date; <u>provided</u>, <u>however</u>, that if the Effective Date is within thirty (30) days of the end of a calendar quarter, then the first Quarterly Distribution Date will be the last Business Day of the month following the end of the first calendar quarter after the calendar quarter in which the Effective Date falls.

Registration Rights Agreement means, in the event that the Backstop Rights Offering Effective Date Condition is satisfied, that certain registration rights agreement to be entered into on the Effective Date between Reorganized Bally and certain holders of the New Common Stock, substantially in the form of Exhibit K attached to this Plan.

Reinstated means, with respect to any Claim, (a) leaving unaltered the legal, equitable, and contractual rights to which a Claim entitles the holder of such Claim in accordance with Section 1124 of the Bankruptcy Code or (b) notwithstanding any contractual provision or applicable law that entitles the holder of such Claim to demand or receive accelerated payment of such Claim after the occurrence of a default: (i) curing any such default that occurred before or after the Petition Date, other than a default of a kind specified in section 365(b)(2) of the Bankruptcy Code or of a kind that section 365(b)(2) of the Bankruptcy Code expressly does not require to be cured; (ii) reinstating the maturity of such Claim as such maturity existed before such default; (iii) compensating the holder of such Claim for any damages incurred as a result of any reasonable reliance by such Holder on such contractual provision or such applicable law; (iv) if such Claim arises from any failure to perform a non-monetary obligation, other than a default arising from failure to operate a non-residential real property lease subject to section 365(b)(1)(A) of the Bankruptcy Code, compensating the Holder of such Claim (other than any

Debtor or an insider of any Debtor) for any actual pecuniary loss incurred by such Holder as a result of such failure; and (v) not otherwise altering the legal, equitable, or contractual rights to which such Claim entitles the Holder of such Claim.

Rejection Claim means an unsecured claim listed on the Rejection Claims List.

Rejection Claims List means (i) with respect to Bally, the list of Rejection Claims set forth on Exhibit E-1 to this Plan, and (ii) with respect to the Affiliate Debtors, the list of Rejection Claims set forth on Exhibit E-2 to this Plan.

Related Persons means, with respect to any Person, such Person's predecessors, successors, assigns and present and former affiliates (whether by operation of law or otherwise) and each of their respective members, partners, equity-holders, officers, directors, employees, representatives, advisors, attorneys, agents and professionals, in each case acting in such capacity on or any time after the Petition Date, and any Person claiming by or through any of them; provided, however, that no insurers of the Debtors and no Person or Entity listed on Exhibit E-1 or E-2 to this Plan shall constitute a Related Person.

Released Parties means, collectively, (i) the Debtors, their Estates, and the Reorganized Debtors, (ii) the Holders of Prepetition Lenders Claims and the agents under the Prepetition Credit Agreement, (iii) the Holders of DIP Lenders Claims and the agents under the DIP Credit Agreement, (iv) the Prepetition Senior Notes Indenture Trustee and each Prepetition Senior Noteholder (solely in its capacity as a Prepetition Senior Noteholder), (v) the Prepetition Senior Subordinated Notes Indenture Trustee and each Prepetition Senior Subordinated Noteholder (solely in its capacity as a Prepetition Senior Subordinated Noteholder), (vi) each Backstop Party (solely in its capacity as a Backstop Party), (vii) the Creditors' Committee, if any, (viii) the Prepetition Noteholders Committee, (ix) the New Investors in the event that the Harbinger Investment Effective Date Condition is satisfied, and (x) Liberation in the event that the Harbinger Investment Effective Date Condition is satisfied, and the respective Related Persons of each of the foregoing; provided, however, that no Person or Entity listed on Exhibit E-1 or E-2 to this Plan shall be considered or deemed a Released Party.

Reorganized Bally means Bally Total Fitness Holding Corporation, a Delaware corporation, as reorganized pursuant to this Plan on or after the Effective Date.

Reorganized Debtors means the Debtors as reorganized pursuant to this Plan on or after the Effective Date.

Restructuring Transactions has the meaning ascribed thereto in <u>Section 5.2</u> of this Plan.

Rights means, in the event that the Backstop Rights Offering Effective Date Condition is satisfied, the non-detachable, non-certificated rights to purchase the Rights Offering Senior Subordinated Notes in an aggregate principal amount not to exceed the Rights Offering Amount as provided for in the Rights Offering.

Rights Offering means, in the event that the Backstop Rights Offering Effective Date Condition is satisfied, that certain rights offering of Rights Offering Senior Subordinated

Notes whereby Rights Offering Recipients shall be offered the opportunity to subscribe for the Rights Offering Senior Subordinated Notes at the Subscription Price (and if not exercised by such Persons, including by exercise of oversubscription rights, to be purchased by the Backstop Parties in accordance with the terms and conditions of the Subscription and Backstop Purchase Agreement).

Rights Offering Amount means, in the event that the Backstop Rights Offering Effective Date Condition is satisfied, an amount equal to 27.9% of Allowed Prepetition Senior Subordinated Notes Claims in Class 6-A and Allowed Rejection Claims against only Bally in Class 6-B-1.

Rights Offering Consideration means Cash.

Rights Offering Period means, in the event that the Backstop Rights Offering Effective Date Condition is satisfied, the time period during which the Rights Offering Recipients may subscribe to purchase the Rights Offering Senior Subordinated Notes, which period shall commence on the Subscription Commencement Date and expire on the Subscription Expiration Date.

Rights Offering Recipients means, in the event that the Backstop Rights Offering Effective Date Condition is satisfied, Holders of Allowed Claims in Classes 6-A and 6-B-1 existing as of the applicable Rights Offering Recipients Record Date.

Rights Offering Recipients Record Date means, in the event that the Backstop Rights Offering Effective Date Condition is satisfied, (i) with respect to any Prepetition Senior Subordinated Noteholder, the earliest date reasonably practicable occurring after the Petition Date and following approval of such date by order of the Bankruptcy Court and (ii) with respect to any holder of an Allowed Class 6-B-1 Claim, the later of the Effective Date and the date upon which such Claim becomes an Allowed Claim.

Rights Offering Senior Subordinated Notes means, in the event that the Backstop Rights Offering Effective Date Condition is satisfied, the senior subordinated notes in an aggregate principal amount not to exceed the Rights Offering Amount, which Rights Offering Senior Subordinated Notes shall be subordinate in priority of payment to the New Senior Second Lien Notes and senior in priority of payment to the New Subordinated Notes and the New Junior Subordinated Notes.

Rights Offering Senior Subordinated Notes Indenture means, in the event that the Backstop Rights Offering Effective Date Condition is satisfied, that certain Indenture, to be dated as of the Effective Date, among Reorganized Bally, as issuer, and the Rights Offering Senior Subordinated Notes Indenture Trustee, relating to the Rights Offering Senior Subordinated Notes, substantially in the form of Exhibit H attached to this Plan.

Rights Offering Senior Subordinated Notes Indenture Trustee means, in the event that the Backstop Rights Offering Effective Date Condition is satisfied, the indenture trustee under the Rights Offering Senior Subordinated Notes Indenture, and its successors and assigns.

Scheduled means with respect to any Claim or Interest, the status and amount, if any, of such Claim or Interest as set forth in the Schedules, if any such Schedules are required to be filed by order of the Bankruptcy Court.

Schedules means the schedules of assets and liabilities, the list of Holders of Interests and the statements of financial affairs, if any, Filed by the Debtors pursuant to section 521 of the Bankruptcy Code and Bankruptcy Rules, as such schedules have been or may be further modified, amended or supplemented in accordance with Bankruptcy Rule 1009 or orders of the Bankruptcy Court.

Second Restructuring Support Agreements means (i) that certain Restructuring Support Agreement, dated as of August 15, 2007 between Bally, the Prepetition Senior Subordinated Noteholders party thereto, and the New Investors, in substantially the form attached hereto as Exhibit M-1, and (ii) that certain Restructuring Support Agreement, dated as of August 15, 2007 between the Debtors and Liberation, in substantially the form attached hereto as Exhibit M-2.

Section 1127(a) Order means an order of the Bankruptcy Court, in form and substance reasonably acceptable to Harbinger and the Debtors, granting the Debtors' motion for an order authorizing the Debtors, pursuant to Section 1127(a) of the Bankruptcy Code and Bankruptcy Rule 3019, to modify their original plan of reorganization, dated June 27, 2007 (the "Original Plan"), without the need to resolicit the votes of any creditor with respect to the Plan, and finding that this Plan does not adversely affect any class of creditors whose votes were solicited for the Original Plan and that this Plan is deemed accepted by all creditors who have previously accepted the Original Plan.

Secured Claim means a Claim that is secured by a Lien on property in which a Debtor's Estate has an interest or that is subject to setoff under section 553 of the Bankruptcy Code, to the extent of the value of the Claim holder's interest in the applicable Estate's interest in such property or to the extent of the amount subject to setoff, as applicable, as determined pursuant to section 506(a) of the Bankruptcy Code or, in the case of setoff, pursuant to section 553 of the Bankruptcy Code.

Secured Tax Claim means any Secured Claim which, absent its secured status, would be entitled to priority in right of payment under section 507(a)(8) of the Bankruptcy Code.

Securities Act means the Securities Act of 1933, 15 U.S.C. §§ 77c-77aa, as now in effect or hereafter amended.

Subscribing Rights Offering Recipient means, in the event that the Backstop Rights Offering Effective Date Condition is satisfied, a Rights Offering Recipient who executes and delivers the Subscription Form to Bally or other applicable Disbursing Agent prior to the expiration of the applicable Rights Offering Period.

Subscription and Backstop Purchase Agreement means that certain Subscription and Backstop Purchase Agreement dated June 27, 2007 by and among Bally and the Backstop Parties, a copy of which is attached to this Plan as Exhibit C.

Subscription Commencement Date means, in the event that the Backstop Rights Offering Effective Date Condition is satisfied, with respect to any Rights Offering Recipient, the earliest date (a) reasonably practicable occurring after the Rights Offering Recipients Record Date applicable for such Rights Offering Recipient and (b) approved by order of the Bankruptcy Court.

Subscription Expiration Date means, in the event that the Backstop Rights Offering Effective Date Condition is satisfied, the date on which the Rights Offering Period shall expire as set forth in the Subscription Form, which date shall be the date that is twenty (20) Business Days after the Subscription Commencement Date (or such later date as approved by order of the Bankruptcy Court).

Subscription Form means, in the event that the Backstop Rights Offering Effective Date Condition is satisfied, each "Subscription Form for Rights Offering in Connection with the Debtors' Joint Prepackaged Plan of Reorganization Under Chapter 11 of the Bankruptcy Code" to be completed and executed by each Subscribing Rights Offering Recipient in connection with its purchase of Rights Offering Senior Subordinated Notes and to be delivered to Bally or other applicable Disbursing Agent prior to the expiration of the applicable Rights Offering Period.

Subscription Notification Date means, in the event that the Backstop Rights Offering Effective Date Condition is satisfied, a date that is not later than five (5) Business Days following the applicable Subscription Expiration Date.

Subscription Payment Date means, in the event that the Backstop Rights Offering Effective Date Condition is satisfied, a date that is not later than five (5) Business Days following the applicable Subscription Notification Date (or such later date as approved in writing by the Debtors or Reorganized Debtors); provided, however, that such date must occur on or prior to the Effective Date with respect to the Prepetition Senior Subordinated Noteholders.

Subscription Price means, in the event that the Backstop Rights Offering Effective Date Condition is satisfied, \$1.00 per dollar of principal amount of Rights Offering Senior Subordinated Notes.

Subsequent Distribution means any distribution of property under this Plan to Holders of Allowed Claims and, in the event that the Harbinger Investment Effective Date Condition is satisfied, Allowed Interests other the initial distribution given on the Initial Distribution Date.

Substantive Consolidation Order means the order, or provision of the Confirmation Order, substantively consolidating the Chapter 11 Cases on the limited basis as provided in <u>Section 5.1</u> of this Plan.

Transfer or Transferable means, with respect to any security or the right to receive a security or to participate in any offering of any security, including the Rights Offering, (i) the sale, transfer, pledge, hypothecation, encumbrance, assignment, constructive sale, participation in, or other disposition of such security or right or the beneficial ownership thereof, (ii) the offer to make such a sale, transfer, constructive sale, or other disposition, and (iii) each

option, agreement, arrangement, or understanding, whether or not in writing and whether or not directly or indirectly, to effect any of the foregoing. The term "constructive sale" for purposes of this definition means (i) a short sale with respect to such security or right, (ii) entering into or acquiring an offsetting derivative contract with respect to such security or right, (iii) entering into or acquiring a futures or forward contract to deliver such security or right, or (iv) entering into any transaction that has substantially the same effect as any of the foregoing. The term "beneficially owned" or "beneficial ownership" as used in this definition shall include, with respect to any security or right, the beneficial ownership of such security or right by a Person and by any direct or indirect subsidiary of such Person.

Unexercised Rights has the meaning given in Section 5.13(a) of this Plan.

Unimpaired Claim means a Claim that is not impaired within the meaning of section 1124 of the Bankruptcy Code.

Unimpaired Unsecured Claim means a Claim that is not an Administrative Claim, Priority Tax Claim, Non-Tax Priority Claim, Other Secured Claim, Prepetition Lenders Claim, Impaired Unsecured Claim, 510(b) Equity Claim or Intercompany Claim.

Voting Deadline means July 27, 2007.

Voting Record Date means June 22, 2007.

- 1.2 **Exhibits and Plan Schedules**. All Exhibits and Plan Schedules are incorporated into and are a part of this Plan as if set forth in full herein. Holders of Claims and Interests may obtain a copy of the Exhibits and Plan Schedules upon written request to the Debtors. The Exhibits and Plan Schedules may be inspected in the office of the clerk of the Bankruptcy Court or its designee during normal business hours or obtained by written request to counsel to the Debtors.
- 1.3 **Rules of Interpretation and Computation of Time**. For purposes of this Plan, unless otherwise provided herein:
- (a) whenever from the context it is appropriate, each term, whether stated in the singular or the plural, will include both the singular and the plural;
- (b) unless otherwise provided in this Plan, any reference in this Plan to a contract, instrument, release, or other agreement or document being in a particular form or on particular terms and conditions means that such document will be substantially in such form or substantially on such terms and conditions;
- (c) any reference in this Plan to an existing document or schedule Filed or to be Filed means such document or schedule, as it may have been or may be amended, modified, or supplemented pursuant to this Plan;
- (d) any reference to an Entity as a Holder of a Claim or Interest includes that Entity's successors and assigns;

- (e) all references in this Plan to Sections, Articles, and Schedules are references to Sections, Articles, and Schedules of or to this Plan;
- (f) the words "herein," "hereunder," and "hereto" refer to this Plan in its entirety rather than to a particular portion of this Plan;
- (g) captions and headings to Articles and Sections are inserted for convenience of reference only and are not intended to be a part of or to affect the interpretation of this Plan;
- (h) subject to the provisions of any contract, certificates of incorporation, by-laws, instrument, release, or other agreement or document entered into in connection with this Plan, the rights and obligations arising under this Plan shall be governed by, and construed and enforced in accordance with, federal law, including the Bankruptcy Code and Bankruptcy Rules;
- (i) the rules of construction set forth in section 102 of the Bankruptcy Code will apply to this Plan; and
- (j) in computing any period of time prescribed or allowed by this Plan, the provisions of Bankruptcy Rule 9006(a) will apply.

ARTICLE TWO

CLASSIFICATION OF CLAIMS AND INTERESTSAll Claims and Interests, except Administrative Claims and Priority Tax Claims, are placed in the Classes set forth below. In accordance with section 1123(a)(1) of the Bankruptcy Code, Administrative Claims and Priority Tax Claims have not been classified as described below.

This Plan constitutes a single plan of reorganization for all Debtors for all purposes, including, without limitation, for voting, confirmation, and distribution purposes. A Claim or Interest is placed in a particular Class only to the extent that the Claim or Interest falls within the description of that Class and is classified in other Classes to the extent that any portion of the Claim or Interest falls within the description of such other Classes. A Claim or Interest is also placed in a particular Class for the purpose of receiving distributions pursuant to this Plan only to the extent that such Claim or Interest is an Allowed Claim or Interest in that Class and such Claim or Interest has not been paid, released, or otherwise settled prior to the Effective Date. As described more fully in Section 5.1 below, this Plan contemplates and is predicated upon entry of an order substantively consolidating the Debtors for the limited purposes of voting, confirmation and distribution with respect to Allowed Class 5 Claims and Allowed Class 6-B-2 Claims only.

Summary of Classification of Claims and Interests				
<u>Class</u> <u>Claim</u> <u>Status</u> <u>Voting Rights</u>				
1.	Non-Tax Priority Claims	Unimpaired	Deemed to Accept	
2.	Other Secured Claims	Unimpaired	Deemed to Accept	

Summary of Classification of Claims and Interests					
Class	<u>Claim</u>	<u>Status</u>	Voting Rights		
3.	Unimpaired Unsecured Claims	Unimpaired	Deemed to Accept		
4.	Prepetition Lenders Claims	Unimpaired	Deemed to Accept		
5.	Prepetition Senior Notes Claims	Impaired	Entitled to Vote		
6-A.	Prepetition Senior Subordinated Notes Claims	Impaired	Entitled to Vote		
6-B-1.	Rejection Claims Against Only Bally	Impaired	Deemed to Reject		
6-B-2.	Rejection Claims Against Any Affiliate Debtor	Impaired	Deemed to Reject		
7.	Old Common Stock	Impaired	Deemed to Reject		
8.	Old Unexercised Equity Interests	Impaired	Deemed to Reject		
9.	Old Equity Interests in Affiliate Debtors	Unimpaired	Deemed to Accept		

2.1 **Unclassified Claims** (not entitled to vote on this Plan)

- (a) Administrative Claims.
- (b) Priority Tax Claims.
- 2.2 **Unimpaired Classes of Claims and Interests** (deemed to have accepted this Plan and, therefore, not entitled to vote on this Plan under section 1126(f) of the Bankruptcy Code).
 - (i) <u>Class 1</u>: Class 1 consists of all Non-Tax Priority Claims.
 - (ii) <u>Class 2</u>: Class 2 consists of all Other Secured Claims. Class 2 consists of separate subclasses for each Other Secured Claim that may exist against the Debtors.
 - (iii) <u>Class 3</u>: Class 3 consists of all Unimpaired Unsecured Claims.
 - (iv) <u>Class 4</u>: Class 4 consists of all Prepetition Lenders Claims.

- (v) Class 9: Class 9 consists of all Old Affiliate Interests.
- 2.3 **Impaired Classes of Claims** (entitled to vote on this Plan).
 - (i) <u>Class 5</u>: Class 5 consists of all Prepetition Senior Notes Claims.
 - (ii) <u>Class 6-A</u>: Class 6-A consists of all Prepetition Senior Subordinated Notes Claims.
- 2.4 **Impaired Classes of Claims and Interests** (deemed to have rejected this Plan and, therefore, not entitled to vote on this Plan).
 - (i) <u>Class 6-B-1</u>: Class 6-B-1 consists of all Rejection Claims against only Bally, but not Rejection Claims against Bally for which one or more Affiliate Debtors are also liable pursuant to a guaranty or otherwise.
 - (ii) <u>Class 6-B-2</u>: Class 6-B-2 consists of all Rejection Claims against any Affiliate Debtor (including, without limitation, Rejection Claims against Bally for which one or more Affiliate Debtors are also liable pursuant to a guaranty or otherwise).
 - (iii) Class 7: Class 7 consists of all Old Common Stock.
 - (iv) <u>Class 8</u>: Class 8 consists of all Old Unexercised Equity Interests of Bally.

ARTICLE THREE TREATMENT OF CLAIMS AND INTERESTS

3.1 Unclassified Claims

- (a) Administrative Claims Generally. Subject to the provisions of sections 330(a), 331, and 503(b) of the Bankruptcy Code, each Administrative Claim shall be paid by the Debtors or Reorganized Debtors, at their election, (i) in full, in Cash, in such amounts as are incurred in the ordinary course of business by the Debtors, or in such amounts as such Administrative Claim is Allowed by the Bankruptcy Court upon the later of the Effective Date or the date upon which there is a Final Order allowing such Administrative Claim, (ii) upon such other terms as may exist in the ordinary course of such Debtor's business or (iii) upon such other terms as may be agreed upon in writing between the Holder of such Administrative Claim and the Debtors, in each case in full satisfaction, settlement, discharge and release of, and in exchange for, such Administrative Claim.
- (i) Professional Fees. All final fee applications for Professional Fees incurred prior to the Effective Date and for services rendered during or in connection with the Chapter 11 Cases shall be filed with the Bankruptcy Court no later than the Professional Fees

Bar Date; *provided, however*, that the reasonable fees and expenses incurred on or after the Petition Date by the Prepetition Noteholders Committee Professionals pursuant to agreements with the Debtors entered into prior to, on, or subsequent to the Petition Date, shall be paid by the Debtors or Reorganized Debtors as Administrative Claims in the ordinary course of the Debtors' business, without application by or on behalf of any such parties to the Bankruptcy Court, and without notice and a hearing, unless specifically required by the Bankruptcy Court. If the Debtors or Reorganized Debtors and any such professional cannot agree on the amount of fees and expenses to be paid to such party, the reasonableness of any such fees and expenses shall be determined by the Bankruptcy Court.

- Claims Arising Under the DIP Credit Agreement. On the Effective (ii) Date, any and all DIP Lenders Claims shall be (A) paid in full in Cash, (B) assumed by the applicable Reorganized Debtors on terms and conditions acceptable to the Holders of such Claims, which terms and conditions may be evidenced by the New Credit Agreement or in some other manner acceptable to such Holders with terms no less favorable in the aggregate for the Debtors and/or Reorganized Debtors than those in the New Credit Agreement, or (C) satisfied in such other manner with terms no less favorable in the aggregate for the Debtors and/or Reorganized Debtors than those in the New Credit Agreement as the applicable Debtors or Reorganized Debtors and such Holders shall have agreed in writing. On the full payment or other satisfaction of such Claims, unless such DIP Lenders Liens have been continued as part of the treatment of such Claims pursuant to clause (B) or (C) above and the Confirmation Order, the DIP Lenders Liens shall be deemed released, terminated and extinguished, in each case without further notice to or order of the Bankruptcy Court, act or action under applicable law, regulation, order, or rule or the vote, consent, authorization or approval of any Person. To the extent the DIP Credit Documents bind the Holders of DIP Lenders Claims as to the manner in which their Claims may be treated pursuant to this Plan, no further agreement or consent from such Holders shall be necessary in order to provide such treatment in accordance with the provisions hereof.
- Priority Tax Claims. The legal, equitable and contractual rights of the Holders of Priority Tax Claims are unaltered by this Plan. On, or as soon as reasonably practicable after, the later of (i) the Initial Distribution Date if such Priority Tax Claim is an Allowed Priority Tax Claim as of the Effective Date or (ii) the date on which such Priority Tax Claim becomes an Allowed Priority Tax Claim, each Holder of an Allowed Priority Tax Claim shall receive in full satisfaction, settlement, discharge and release of, and in exchange for, such Allowed Priority Tax Claim, at the election of the applicable Debtor (A) Cash equal to the amount of such Allowed Priority Tax Claim; (B) such other less favorable treatment as to which the applicable Debtor or Reorganized Debtor and the Holder of such Allowed Priority Tax Claim shall have agreed upon in writing; or (C) such other treatment such that it will not be impaired pursuant to section 1124 of the Bankruptcy Code, including payment in accordance with the provisions of section 1129(a)(9)(C) of the Bankruptcy Code; provided, however, that Priority Tax Claims incurred by the applicable Debtor in the ordinary course of business may be paid in the ordinary course of business in accordance with such applicable terms and conditions relating thereto in the discretion of the applicable Debtor or Reorganized Debtor without further notice to or order of the Bankruptcy Court. Each Holder of an Allowed Secured Tax Claim shall retain the Liens securing its Allowed Secured Tax Claim as of the Effective Date until full and final payment of such Allowed Secured Tax Claim is made as provided herein. On the full payment or other satisfaction of such obligations, the Liens securing such Allowed Secured Tax Claim shall

be deemed released, terminated and extinguished, in each case without further notice to or order of the Bankruptcy Court, act or action under applicable law, regulation, order, or rule or the vote, consent, authorization or approval of any Person.

3.2 Unimpaired Classes of Claims

- (a) <u>Class 1</u>: Non-Tax Priority Claims. The legal, equitable and contractual rights of the Holders of Class 1 Claims are unaltered by this Plan. On, or as soon as reasonably practicable after, the later of (i) the Initial Distribution Date if such Class 1 Claim is an Allowed Class 1 Claim on the Effective Date or (ii) the date on which such Class 1 Claim becomes an Allowed Class 1 Claim, each Holder of an Allowed Class 1 Claim shall receive in full satisfaction, settlement, discharge and release of, and in exchange for, such Allowed Class 1 Claim, at the election of the applicable Debtor: (A) Cash equal to the amount of such Allowed Class 1 Claim; (B) such other less favorable treatment as to which the applicable Debtor or Reorganized Debtor and the Holder of such Allowed Class 1 Claim shall have agreed upon in writing; or (C) such other treatment such that it will not be impaired pursuant to section 1124 of the Bankruptcy Code; provided, however, that Class 1 Claims incurred by the applicable Debtor in the ordinary course of business may be paid in the ordinary course of business in accordance with the terms and conditions of any agreements relating thereto in the discretion of the applicable Debtor or Reorganized Debtor without further notice to or order of the Bankruptcy Court.
- <u>Class 2</u>: Other Secured Claims. The legal, equitable and contractual rights of the Holders of Class 2 Claims are unaltered by this Plan. On, or as soon as reasonably practicable after, the later of (i) the Initial Distribution Date if such Class 2 Claim is an Allowed Class 2 Claim on the Effective Date or (ii) the date on which such Class 2 Claim becomes an Allowed Class 2 Claim, each Holder of an Allowed Class 2 Claim shall receive in full satisfaction, settlement, discharge and release of, and in exchange for, such Allowed Class 2 Claims, at the election of the applicable Debtor: (A) Cash equal to the amount of such Allowed Class 2 Claim; (B) such other less favorable treatment as to which the applicable Debtor or Reorganized Debtor and the Holder of such Allowed Class 2 Claim shall have agreed upon in writing; or (C) such other treatment such that it will not be impaired pursuant to section 1124 of the Bankruptcy Code; provided, however, that Class 2 Claims incurred by the applicable Debtor in the ordinary course of business may be paid in the ordinary course of business in accordance with the terms and conditions of any agreements relating thereto in the discretion of the applicable Debtor or Reorganized Debtor without further notice to or order of the Bankruptcy Court. Each Holder of an Allowed Other Secured Claim shall retain the Liens securing its Allowed Other Secured Claim as of the Effective Date until full and final payment of such Allowed Other Secured Claim is made as provided herein. On the full payment or other satisfaction of such obligations, the Liens securing such Allowed Other Secured Claim shall be deemed released, terminated and extinguished, in each case without further notice to or order of the Bankruptcy Court, act or action under applicable law, regulation, order, or rule or the vote, consent, authorization or approval of any Person.
- (c) <u>Class 3</u>: Unimpaired Unsecured Claims. The legal, equitable and contractual rights of the Holders of Class 3 Claims are unaltered by this Plan. On, or as soon as reasonably practicable after, the later of (i) the Initial Distribution Date if such Class 3 Claim is

Allowed on the Effective Date or (ii) the date on which such Class 3 Claim becomes Allowed, each Holder of an Allowed Class 3 Claim shall receive in full satisfaction, settlement, discharge and release of, and in exchange for, such Allowed Class 3 Claim, at the election of the applicable Debtor: (A) Cash equal to the amount of such Allowed Class 3 Claim; (B) such other less favorable treatment as to which the applicable Debtor or Reorganized Debtor and the Holder of such Allowed Class 3 Claim shall have agreed upon in writing; or (C) such other treatment such that it will not be impaired pursuant to section 1124 of the Bankruptcy Code; provided, however, that Class 3 Claims incurred by the applicable Debtor in the ordinary course of business may be paid in the ordinary course of business in accordance with the terms and conditions of any agreements relating thereto in the discretion of the applicable Debtor or Reorganized Debtor without further notice to or order of the Bankruptcy Court.

Class 4: Prepetition Lenders Claims. Unless earlier refinanced by a DIP Loan Facility, on the Effective Date, any and all Allowed Class 4 Claims shall be (A) paid in full in Cash, (B) assumed by the applicable Reorganized Debtors on terms and conditions acceptable to the Holders of such Claims, which terms and conditions may be evidenced by the New Credit Agreement or in some other manner acceptable to such Holders with terms no less favorable in the aggregate for the Debtors and/or Reorganized Debtors than those in the New Credit Agreement, or (C) satisfied in such other manner with terms no less favorable in the aggregate for the Debtors and/or Reorganized Debtors than those in the New Credit Agreement as the applicable Debtors or Reorganized Debtors and such Holders shall have agreed in writing. On the full payment or other satisfaction of such Claims, unless such Prepetition Lenders Liens have been continued as part of the treatment of such Claims pursuant to clause (B) or (C) above and the Confirmation Order, the Prepetition Lenders Liens shall be deemed released, terminated and extinguished, in each case without further notice to or order of the Bankruptcy Court, act or action under applicable law, regulation, order, or rule or the vote, consent, authorization or approval of any Person. To the extent the Prepetition Credit Documents bind the Holders of Class 4 Claims as to the manner in which their Claims may be treated pursuant to this Plan, no further agreement or consent from such Holders shall be necessary in order to provide such treatment in accordance with the provisions hereof.

3.3 Impaired Classes of Claims and Interests

- I. In the event that the Harbinger Investment Effective Date Condition is satisfied:
- (a) <u>Class 5</u>: Prepetition Senior Notes Claims. On the Effective Date, the Prepetition Senior Notes Indenture shall be replaced in its entirety by the New Senior Second Lien Notes Indenture. On, or as soon as reasonably practicable after, the Effective Date, the Prepetition Senior Notes Indenture Trustee shall receive in full satisfaction, settlement, discharge and release of, and in exchange for, the Allowed Class 5 Claims, the following for its benefit and the benefit of, and to be shared on a Pro Rata basis among, each Prepetition Senior Noteholder:
 - (i) the New Senior Second Lien Notes; and
 - (ii) the Prepetition Senior Notes Indenture Amendment Fee.

- (b) <u>Class 6-A: Prepetition Senior Subordinated Notes Claims</u>. On, or as soon as reasonably practicable after the Effective Date, each holder of a Prepetition Senior Subordinated Notes Claim shall, in the sole discretion of the Debtors and in full satisfaction, settlement, discharge and release of, and in exchange for, such Allowed Class 6-A Claim, their pro rata share of (i) Cash in the amount of \$123.5 million and (ii) New Harbinger Subordinated Notes. On the Effective Date, subject to the provisions of the foregoing sentence, the Prepetition Senior Subordinated Notes Indenture shall be replaced in its entirety by the New Subordinated Notes Indenture.
- <u>Class 6-B-1 and 6-B-2</u>: Rejection Claims. On, or as soon as reasonably (c) practicable after, the later of (i) the Initial Distribution Date if such Class 6-B-1 or 6-B-2 Claim is Allowed on the Effective Date or (ii) the date on which such Class 6-B-1 or 6-B-2 Claim becomes Allowed, each Holder of an Allowed Class 6-B-1 and 6-B-2 Claim shall receive, at the election of the Debtors or the Reorganized Debtors, as applicable, and in full satisfaction, settlement, release and discharge of, and in exchange for, such Allowed Class 6-B-1 or 6-B-2 Claim, (A) Cash equal to the amount of such Allowed Class 6-B-1 or 6-B-2 Claim; (B) such other less favorable treatment as to which the Debtors or Reorganized Debtors and the Holder of such Allowed Class 6-B-1 or 6-B-2 Claim shall have agreed upon in writing; or (C) regular installment payments in Cash: (x) of a total value, as of the Effective Date, equal to the Allowed amount of such Claim; (y) which total value shall include simple interest to accrue on any outstanding balance of such Allowed Claim starting on the Effective Date at a fixed interest rate equal to 13% per annum; and (z) over a period ending not later than five years after the Effective Date. Any such installment payments shall be made in equal quarterly Cash payments beginning on the first Quarterly Distribution Date following the Effective Date, and continuing on each subsequent Quarterly Distribution Date thereafter until payment in full of the Allowed Class 6-B-1 or 6-B-2 Claim.
- (d) <u>Class 7</u>: Old Common Stock. On, or as soon as reasonably practicable after, the later of (i) the Initial Distribution Date if such Class 7 Interest is Allowed on the Effective Date or (ii) the date on which such Class 7 Interest becomes Allowed, each Holder of an Allowed Class 7 Interest shall receive, in full satisfaction, settlement, discharge and release of, and in exchange for, such Allowed Class 7 Interest, its Pro Rata Share of the Old Common Stock Cash Amount.
- (e) <u>Class 8</u>: Old Unexercised Equity Interests. On the Effective Date, the Old Unexercised Equity Interests of Bally will be cancelled, and the Holders of such Old Unexercised Equity Interests shall not receive any distribution or retain any property on account of such Old Unexercised Equity Interests.
- II. In the event that the Backstop Rights Offering Effective Date Condition is satisfied:

- (f) <u>Class 5</u>: Prepetition Senior Notes Claims. On the Effective Date, the Prepetition Senior Notes Indenture shall be replaced in its entirety by the New Senior Second Lien Notes Indenture. On, or as soon as reasonably practicable after, the Effective Date, the Prepetition Senior Notes Indenture Trustee shall receive in full satisfaction, settlement, discharge and release of, and in exchange for, the Allowed Class 5 Claims, the following for its benefit and the benefit of, and to be shared on a Pro Rata basis among, each Prepetition Senior Noteholder:
 - (i) the New Senior Second Lien Notes; and
 - (ii) the Prepetition Senior Notes Indenture Amendment Fee.
- Rejection Claims Against Only Bally. Holders of Allowed Claims in Classes 6-A and 6-B-1 shall each receive the same treatment under this Plan. On the Effective Date, the Prepetition Senior Subordinated Notes Indenture shall be replaced in its entirety by the New Subordinated Notes Indenture and the New Junior Subordinated Notes Indenture. On, or as soon as reasonably practicable after, the later of (i) the Initial Distribution Date if such applicable Class 6-A and 6-B-1 Claim is Allowed on the Effective Date or (ii) the date on which such Class 6-A and 6-B-1 Claim becomes Allowed, each Prepetition Senior Subordinated Noteholder and Holder of an Allowed Rejection Claim against only Bally shall receive the following property, in full satisfaction, settlement, discharge and release of, and in exchange for, its Allowed Prepetition Senior Subordinated Notes Claim and Allowed Rejection Claims against only Bally:
 - (i) New Subordinated Notes with a principal amount equal 24.8% of the amount of such Allowed Claim;
 - (ii) New Junior Subordinated Notes with a principal amount equal to 21.7% of the amount of such Allowed Claim;
 - (iii) 0.00093 shares of New Common Stock for each \$1.00 of such Allowed Claim; and
 - (iv) Rights to purchase Rights Offering Senior Subordinated Notes with a principal amount equal to 27.9% of the amount of such Allowed Claim.

Notwithstanding the foregoing, each Holder of a Class 6-A Claim and Class 6-B-1 Claim shall execute and deliver the New Stockholders Agreement prior to receiving any New Common Stock. If any such holder has not executed and delivered the New Stockholders Agreement by the 60th day after the Subscription Expiration Date applicable to such holder, such holder shall no longer be eligible to receive any distribution of the New Common Stock and such holder's share of the New Common Stock will be distributed Pro Rata to the remaining Holders of Class 6-A and 6-B-1 Claims that are parties to the New Stockholders Agreement.

(h) <u>Class 6-B-2</u>: Rejection Claims Against Any Affiliate Debtor. On, or as soon as reasonably practicable after, the later of (i) the Initial Distribution Date if such Class 6-B-2 Claim is Allowed on the Effective Date or (ii) the date on which such Class 6-B-2 Claim becomes Allowed, each Holder of an Allowed Class 6-B-2 Claim shall receive, at the election of

the Debtors and in full satisfaction, settlement, release and discharge of, and in exchange for, such Allowed Class 6-B-2 Claim, (A) Cash equal to the amount of such Allowed Class 6-B-2 Claim; (B) such other less favorable treatment as to which the Debtors or Reorganized Debtors and the Holder of such Allowed Class 6-B-2 Claim shall have agreed upon in writing; or (C) regular installment payments in Cash: (x) of a total value, as of the Effective Date, equal to the Allowed amount of such Claim; (y) which total value shall include simple interest to accrue on any outstanding balance of such Allowed Claim starting on the Effective Date at a fixed interest rate equal to 12 3/8% per annum; and (z) over a period ending not later than five years after the Effective Date. Any such installment payments shall be made in equal quarterly Cash payments beginning on the first Quarterly Distribution Date following the Effective Date, and continuing on each subsequent Quarterly Distribution Date thereafter until payment in full of the Allowed Class 6-B-2 Claim.

- (i) <u>Class 7</u>: Old Common Stock. On the Effective Date, the Old Common Stock will be cancelled, and the Holders of such Old Common Stock shall not receive any distribution or retain any property on account of such Old Common Stock.
- (j) <u>Class 8</u>: Old Unexercised Equity Interests. On the Effective Date, the Old Unexercised Equity Interests of Bally will be cancelled, and the Holders of such Old Unexercised Equity Interests shall not receive any distribution or retain any property on account of such Old Unexercised Equity Interests.

3.4 Unimpaired Class of Interests

<u>Class 9</u>: Old Affiliate Interests. Notwithstanding anything in this Plan to the contrary, on the Effective Date, the Old Affiliate Interests shall remain effective and outstanding and be owned and held by the same applicable Person(s) that held and/or owned such Interests immediately prior to the Effective Date.

3.5 **Special Provision Regarding Unimpaired Claims**

Except as otherwise provided in this Plan, nothing shall affect the Debtors' or Reorganized Debtors' rights, remedies and defenses, both legal and equitable, with respect to any Unimpaired Claims or claims and actions arising from or under any executory contract or unexpired lease assumed or rejected by the Debtors, including, but not limited to, all rights with respect to legal and equitable defenses, including setoff or recoupment.

ARTICLE FOUR ACCEPTANCE OR REJECTION OF THE PLAN

4.1 Impaired Classes of Claims Entitled to Vote

Holders of Claims in Classes 5 and 6-A are entitled to vote to accept or reject this Plan.

4.2 Acceptance by an Impaired Class

In accordance with section 1126(c) of the Bankruptcy Code and except as provided in section 1126(e) of the Bankruptcy Code, an Impaired Class of Claims shall have accepted this Plan if this Plan is accepted by the Holders of at least two-thirds (2/3) in dollar amount and more than one-half (1/2) in number of the Allowed Claims of such Class that have timely and properly voted to accept or reject this Plan.

4.3 Presumed Acceptances by Unimpaired Classes

Classes 1, 2, 3, 4 and 9 are Unimpaired by this Plan. Accordingly, under section 1126(f) of the Bankruptcy Code, Holders of such Claims and Interests are conclusively presumed to accept this Plan, and the votes of the Holders of such Claims will not be solicited.

4.4 Presumed Rejection by Certain Impaired Classes

Holders of Allowed Claims in Class 6-B-1 and Class 6-B-2 are entitled to receive and retain certain property under this Plan as described more fully in Section 3.3.I.(c), II.(c) and II.(d) of this Plan. Notwithstanding such recovery, however, this Plan deems the Holders of Class 6-B-1 and Class 6-B-2 Claims to have rejected this Plan and the votes of Holders of Claims in Class 6-B-1 and Class 6-B-2 will not be solicited.

In the event that the Harbinger Investment Effective Date Condition is satisfied, Holders of Claims and Interests in Class 7 will be receiving distributions under this Plan as described more fully in Section 3.3 L(d). In the event that the Backstop Rights Offering Effective Date Condition is satisfied, Holders of Allowed Claims and Interests in Class 7 will not receive or retain any property under this Plan as described more fully in Section 3.3 II.(e). Notwithstanding their potential recovery, this Plan deems the Holders of Claims and Interests in Class 7 to have rejected this Plan and the votes of Holders of Claims and Interests in Class 7 will not be solicited.

Holders of Allowed Interests in Class 8 are not entitled to receive or retain any property under this Plan. Accordingly, under section 1126(g) of the Bankruptcy Code, the votes of Holders of Interests in Class 8 will not be solicited and such Holders are deemed to reject this Plan.

4.5 Confirmation Pursuant to Section 1129(b) of the Bankruptcy Code

Because certain Classes are deemed to have rejected this Plan, the Debtors will request confirmation of this Plan, as it may be modified from time to time, under section 1129(b) of the Bankruptcy Code; provided, however, that if Class 6-A rejects this Plan, the Debtors will not request confirmation of this Plan (as modified from time to time) without the prior written consent of the Majority Backstop Parties. The Debtors reserve the right to alter, amend, modify, revoke or withdraw this Plan or any Exhibit or Plan Schedule in order to satisfy the requirements of section 1129(b) of the Bankruptcy Code, if necessary.

4.6 Elimination of Vacant Classes

Any Class of Claims that is not occupied as of the commencement of the Confirmation Hearing by an Allowed Claim or a claim temporarily allowed under Bankruptcy

Rule 3018, or as to which no vote is cast, shall be deemed eliminated from this Plan for purposes of voting to accept or reject this Plan and for purposes of determining acceptance or rejection of this Plan by such Class pursuant to section 1129(a)(8) of the Bankruptcy Code.

ARTICLE FIVE MEANS FOR IMPLEMENTATION OF THE PLAN

5.1 Limited Substantive Consolidation for Purposes of Treating Impaired Claims Other Than Impaired Claims Against Only Bally

- Limited Substantive Consolidation. This Plan contemplates and is predicated upon entry of an order substantively consolidating the Debtors solely for the limited purposes of treating Class 5 Claims and Class 6-B-2 Claims, including, without limitation, for voting, confirmation and distribution purposes. This Plan does not contemplate the substantive consolidation of the Debtors with respect to the other Classes of Claims or Interests set forth in this Plan, or for any other purpose. Accordingly, for voting, confirmation and distribution purposes, (i) any obligation of any Debtor and all guarantees with respect to Class 5 Claims and Class 6-B-2 Claims thereof executed by one or more of the other Debtors shall be treated as a single obligation and any obligation of two or more Debtors, and all multiple Impaired Claims against such entities on account of such joint obligations, shall be treated and Allowed only as a single Impaired Claim against the consolidated Debtors, and (ii) each Class 5 Claim and Class 6-B-2 Claim filed or to be filed against any Debtor shall be deemed filed against the consolidated Debtors and shall be deemed a single Class 5 Claim and Class 6-B-2 Claim, as applicable, against and a single obligation of the consolidated Debtors. Except as set forth in this Section 5.1, such limited substantive consolidation shall not and shall not be deemed to (other than for purposes related to this Plan): (i) affect the legal and corporate structures of the Reorganized Debtors, subject to the right of the Debtors or Reorganized Debtors to effect Restructuring Transactions as provided in Section 5.2 of this Plan, (ii) cause any Debtor to be liable for any Impaired Claim or Unimpaired Claim under this Plan for which it otherwise is not liable, and the liability for any such Claim shall not be affected by such substantive consolidation, (iii) affect Intercompany Claims of Debtors against Debtors, (iv) modify, affect or otherwise alter the Old Affiliate Interests, (v) affect any obligations under any leases or contracts assumed in this Plan or otherwise arising subsequent to the filing of the Chapter 11 Cases, or (vi) affect any obligations to pay quarterly fees to the United States Trustee.
- such limited substantive consolidation Order. Unless the Bankruptcy Court has approved such limited substantive consolidation of the Chapter 11 Cases by a prior order, this Plan shall serve as, and shall be deemed to be, a motion for entry of an order substantively consolidating the Debtors on the limited basis as provided in Section 5.1 hereof. If no objection to substantive consolidation is timely filed and served by any Holder of an Impaired Claim affected by this Plan as provided herein on or before the deadline for objection to confirmation of this Plan, the Substantive Consolidation Order (which may be the Confirmation Order) may be entered by the Bankruptcy Court without further notice and hearing as to such substantive consolidation. If any such objections are timely filed and served, a hearing with respect to the substantive consolidation of the Chapter 11 Cases and the objections thereto shall be scheduled by the Bankruptcy Court, which hearing may, but is not required to, coincide with the Confirmation Hearing.

5.2 **Restructuring Transactions**

On or after the Effective Date and without limiting any rights and remedies of the Debtors or Reorganized Debtors under this Plan or applicable law, the applicable Reorganized Debtors may enter into such transactions and may take such actions as may be necessary or appropriate to effect a corporate restructuring of their respective businesses, to otherwise simplify the overall corporate structure of the Reorganized Debtors, or to reincorporate certain of the Affiliate Debtors under the laws of jurisdictions other than the laws of which the applicable Affiliate Debtors are presently incorporated. Such restructuring may include one or more mergers, consolidations, restructures, dispositions, liquidations, or dissolutions, as may be determined by the Debtors or Reorganized Debtors to be necessary or appropriate (collectively, the "Restructuring Transactions"). The actions to effect the Restructuring Transactions may include (i) the execution and delivery of appropriate agreements or other documents of merger, consolidation, restructuring, disposition, liquidation, or dissolution containing terms that are consistent with the terms of this Plan and that satisfy the applicable requirements of applicable state law and such other terms to which the applicable entities may agree; (ii) the execution and delivery of appropriate instruments of transfer, assignment, assumption, or delegation of any asset, property, right, liability, duty, or obligation on terms consistent with the terms of this Plan and having such other terms to which the applicable entities may agree; (iii) the filing of appropriate certificates or articles of merger, consolidation, or dissolution pursuant to applicable state law; and (iv) all other actions that the applicable entities determine to be necessary or appropriate, including making filings or recordings that may be required by applicable state law in connection with such transactions. The Restructuring Transactions may include one or more mergers, consolidations, restructures, dispositions, liquidations, or dissolutions, as may be determined by the Reorganized Debtors to be necessary or appropriate to result in substantially all of the respective assets, properties, rights, liabilities, duties, and obligations of certain of the Reorganized Debtors vesting in one or more surviving, resulting, or acquiring corporations. In each case in which the surviving, resulting, or acquiring corporation in any such transaction is a successor to a Reorganized Debtor, such surviving, resulting, or acquiring corporation will perform the obligations of the applicable Reorganized Debtor pursuant to this Plan to pay or otherwise satisfy the Allowed Claims against such Reorganized Debtor, except as provided in any contract, instrument, or other agreement or document effecting a disposition to such surviving, resulting, or acquiring corporation, which may provide that another Reorganized Debtor will perform such obligations.

5.3 Continued Legal Existence and Vesting of Assets in the Reorganized Debtors

Subject to the Restructuring Transactions permitted by Section 5.2 of this Plan, after the Effective Date, the Reorganized Debtors shall continue to exist as separate legal entities in accordance with the applicable law in the respective jurisdiction in which they are incorporated or formed and pursuant to their respective certificates or articles of incorporation and by-laws, or other applicable organizational documents, in effect immediately prior to the Effective Date, except to the extent such certificates or articles of incorporation and by-laws, or other applicable organizational documents, are amended, amended and restated or otherwise modified under this Plan. Notwithstanding anything to the contrary in this Plan, including Section 5.1 hereof as to the limited substantive consolidation provided therein, the Claims of a particular Debtor or Reorganized Debtor shall remain the obligations solely of such Debtor or

Reorganized Debtor and shall not become obligations of any other Debtor or Reorganized Debtor solely by virtue of this Plan or the Chapter 11 Cases. Except as otherwise provided in this Plan or the Confirmation Order, on and after the Effective Date, all property and assets of the Estates of the Debtors, including all claims, rights, and Litigation Claims of the Debtors, and any other property acquired by the Debtors or the Reorganized Debtors under or in connection with this Plan, shall vest in the Reorganized Debtors free and clear of all Claims, Liens, charges, other encumbrances, and Interests, subject to the Restructuring Transactions and Liens which survive the occurrence of the Effective Date as described in Article III of this Plan. On and after the Effective Date, the Reorganized Debtors may operate their businesses and may use, acquire, and dispose of property and compromise or settle any Claims without supervision of or approval by the Bankruptcy Court and free and clear of any restrictions of the Bankruptcy Code or the Bankruptcy Rules, other than restrictions expressly imposed by this Plan or the Confirmation Order. Without limiting the foregoing, the Reorganized Debtors may pay the charges that they incur on or after the Effective Date for Professionals' fees, disbursements, expenses, or related support services without application or notice to, or order of, the Bankruptcy Court.

5.4 Corporate Governance, Directors, Officers, and Corporate Action

- (a) Certificates of Incorporation and By-Laws. The certificates or articles of incorporation and by-laws of each of the Debtors shall be amended as necessary to satisfy the provisions of this Plan and the Bankruptcy Code, and shall (i) include, among other things, pursuant to section 1123(a)(6) of the Bankruptcy Code, a provision prohibiting the issuance of non-voting equity securities, but only to the extent required by section 1123(a)(6) of the Bankruptcy Code; (ii) authorize the issuance of New Common Stock in an amount not less than the amount necessary to permit the distributions thereof required or contemplated by this Plan; (iii) to the extent necessary or appropriate, include restrictions on the Transfer of New Common Stock; and (iv) to the extent necessary or appropriate, include such provisions as may be needed to effectuate and consummate this Plan and the transactions contemplated herein. After the Effective Date, the Reorganized Debtors may amend and restate their respective certificates or articles of incorporation and by-laws, and other applicable organizational documents, as permitted by applicable law.
 - (b) Directors and Officers of the Reorganized Debtors.
- satisfied, subject to any requirement of Bankruptcy Court approval pursuant to section 1129(a)(5) of the Bankruptcy Code, as of the Effective Date, the initial officers of Reorganized Bally shall be the officers of Bally existing immediately prior to the Effective Date. On the Effective Date, the board of directors of Reorganized Bally shall be selected by the New Investors and identified via a Plan supplement filed with the Bankruptcy Court at least 10 days prior to the Confirmation Hearing. The boards of directors and initial officers of the Reorganized Affiliate Debtors on the Effective Date shall be comprised of the same individuals who currently serve in such capacities immediately prior to the Effective Date (other than those directors of the Affiliate Debtors listed on the resignation plan supplement to be filed with the Bankruptcy Court prior to the Confirmation Hearing, which listed directors will be deemed to have resigned from the applicable Affiliate Debtors on and as of the Effective Date). Pursuant to section 1129(a)(5) of the Bankruptcy Code, the Debtors will disclose, at or prior to the

Confirmation Hearing, the identity and affiliations of any Person proposed to serve on the initial board of directors of Reorganized Bally, and, to the extent such Person is an insider other than by virtue of being a director, the nature of any compensation for such Person. The length of the initial term of each director shall be one year. Each such director and officer shall serve from and after the Effective Date pursuant to applicable law and the terms of the Amended Certificate of Incorporation, the other constituent and organizational documents of the Reorganized Debtors. The existing board of directors of Bally will be deemed to have resigned on and as of the Effective Date.

- In the event that the Backstop Rights Offering Effective Date Condition is (ii) satisfied, subject to any requirement of Bankruptcy Court approval pursuant to section 1129(a)(5) of the Bankruptcy Code, as of the Effective Date, the initial officers of Reorganized Bally shall be the officers of Bally existing immediately prior to the Effective Date. On the Effective Date, the board of directors of Reorganized Bally shall have not less than three and not more than nine members selected by the holders of a majority of the Allowed Prepetition Senior Subordinated Notes Claims and identified via a Plan supplement filed with the Bankruptcy Court at least 10 days prior to the Confirmation Hearing. The boards of directors and initial officers of the Reorganized Affiliate Debtors on the Effective Date shall be comprised of the same individuals who currently serve in such capacities immediately prior to the Effective Date (other than those directors of the Affiliate Debtors listed on the resignation plan supplement to be filed with the Bankruptcy Court prior to the Confirmation Hearing, which listed directors will be deemed to have resigned from the applicable Affiliate Debtors on and as of the Effective Date). Pursuant to section 1129(a)(5) of the Bankruptcy Code, the Debtors will disclose, at or prior to the Confirmation Hearing, the identity and affiliations of any Person proposed to serve on the initial board of directors of Reorganized Bally, and, to the extent such Person is an insider other than by virtue of being a director, the nature of any compensation for such Person. The length of the initial term of each director shall be one year. Each such director and officer shall serve from and after the Effective Date pursuant to applicable law and the terms of the Amended Certificate of Incorporation, the other constituent and organizational documents of the Reorganized Debtors. The existing board of directors of Bally will be deemed to have resigned on and as of the Effective Date.
- Corporate Action. On the Effective Date, the adoption of the Amended (c) Certificate of Incorporation and By-Laws and similar constituent and organizational documents, and the selection of directors and officers for, each of the Reorganized Debtors, and all other actions contemplated by or described in this Plan with respect thereto, shall be authorized and approved and be binding and in full force and effect in all respects (subject to the provisions of this Plan and the Confirmation Order), in each case without further notice to or order of the Bankruptcy Court, act or action under applicable law, regulation, order, or rule (other than filing such organizational documents with the applicable governmental unit as required by applicable law) or the vote, consent, authorization or approval of any Person. All matters provided for in this Plan involving the legal or corporate structure of the Debtors or the Reorganized Debtors, and any legal or corporate action required by the Debtors or the Reorganized Debtors in connection with this Plan, shall be deemed to have occurred and shall be in full force and effect in all respects, in each case without further notice to or order of the Bankruptcy Court, act or action under applicable law, regulation, order, or rule or any requirement of further action, vote or other approval or authorization by the security holders, officers or directors of the Debtors or

the Reorganized Debtors or by any other Person. On the Effective Date, the appropriate officers of the Debtors and Reorganized Debtors and members of their respective boards of directors are authorized to issue, execute, and deliver, and consummate the transactions contemplated by, the contracts, agreements, documents, guarantees, pledges, consents, securities, certificates, resolutions and instruments contemplated by or described in this Plan in the name of and on behalf of the Debtors and Reorganized Debtors, in each case without further notice to or order of the Bankruptcy Court, act or action under applicable law, regulation, order, or rule or any requirement of further action, vote or other approval or authorization by any Person.

5.5 Cancellation of Notes, Instruments, Debentures, Preferred Stock and Common Stock

On the Effective Date, except as otherwise provided in this Plan or the Confirmation Order, (i) the Prepetition Senior Notes, the Prepetition Senior Subordinated Notes, the Old Common Stock, the Old Unexercised Equity Interests, and any other notes, bonds (with the exception of any surety bonds outstanding), indentures, or other instruments or documents evidencing or creating any indebtedness or obligations of a Debtor that are Impaired under this Plan shall be cancelled and extinguished, and (ii) the obligations of the Debtors under any agreements, documents, indentures, or certificates of designation governing the Prepetition Senior Notes, Prepetition Senior Subordinated Notes, Old Common Stock, Old Unexercised Equity Interests, and any other notes, bonds, indentures, or other instruments or documents evidencing or creating any indebtedness or obligations of a Debtor that are Impaired under this Plan shall be, and are hereby, discharged, in each case without further notice to or order of the Bankruptcy Court, act or action under applicable law, regulation, order, or rule or any requirement of further action, vote or other approval or authorization by the security holders, officers or directors of the Debtors or the Reorganized Debtors or by any other Person. Notwithstanding the foregoing, the Prepetition Senior Notes Indenture and the Prepetition Senior Subordinated Notes Indenture shall continue in effect solely for the purposes of: (i) allowing Prepetition Senior Noteholders and Prepetition Senior Subordinated Noteholders to receive distributions under this Plan; and (ii) allowing and preserving the rights of the Prepetition Senior Notes Indenture Trustee and the Prepetition Senior Subordinated Notes Indenture Trustee to make distributions in satisfaction of Allowed Prepetition Senior Notes Claims and Allowed Prepetition Senior Subordinated Notes Claims, but in all cases subject to the terms and conditions of the Prepetition Senior Notes Indenture and Prepetition Senior Subordinated Notes Indenture, including, but not limited to, any charging lien or priority payment rights of the Prepetition Senior Notes Indenture Trustee and the Prepetition Senior Subordinated Notes Indenture Trustee. The Prepetition Senior Notes Indenture Trustee and the Prepetition Senior Subordinated Notes Indenture Trustee shall be entitled to reasonable compensation to the extent that they perform services for the Prepetition Senior Noteholders and the Prepetition Senior Subordinated Noteholders, respectively, (i) before the Effective Date, in Cash on the Effective Date and (ii) after the Effective Date, in Cash and in accordance with the terms of the Prepetition Senior Notes Indenture and the Prepetition Senior Subordinated Notes Indenture, in both cases without further notice to or order of the Bankruptcy Court. As of the Effective Date, the Prepetition Senior Notes and the Prepetition Senior Subordinated Notes shall be surrendered to the Prepetition Senior Notes Indenture Trustee and the Prepetition Senior Subordinated Notes Indenture Trustee, respectively, in accordance with the terms of the Prepetition Senior Notes Indenture and the Prepetition Senior Subordinated Notes Indenture. All surrendered and

canceled Prepetition Senior Notes and Prepetition Senior Subordinated Notes held by the Prepetition Senior Notes Indenture Trustee and the Prepetition Senior Subordinated Notes Indenture Trustee shall be disposed of in accordance with the applicable terms and conditions of the Prepetition Senior Notes Indenture and the Prepetition Senior Subordinated Notes Indenture.

5.6 Issuance of New Securities and Related Documentation

On, or as soon as reasonably practicable after, the Effective Date, Reorganized Bally is authorized to and shall issue the New Common Stock, the New Senior Second Lien Notes, the New Harbinger Subordinated Notes (if the Harbinger Investment Effective Date Condition is satisfied), the New Subordinated Notes (if the Backstop Rights Offering Condition is satisfied), the New Junior Subordinated Notes (if the Backstop Rights Offering Condition is satisfied), the Rights Offering Senior Subordinated Notes (if the Backstop Rights Offering Condition is satisfied), and any and all other securities, notes, stock, instruments, certificates, and other documents or agreements required to be issued, executed or delivered pursuant to this Plan (collectively with the Rights, the "New Securities and Documents"), in each case without further notice to or order of the Bankruptcy Court, act or action under applicable law, regulation, order, or rule or the vote, consent, authorization or approval of any Person. The issuance of the New Securities and Documents and the distribution thereof under this Plan (and exercise of the Rights) shall be exempt from registration under applicable securities laws pursuant to section 1145(a) of the Bankruptcy Code. Without limiting the effect of section 1145 of the Bankruptcy Code, all documents, agreements, and instruments entered into and delivered on or as of the Effective Date contemplated by or in furtherance of this Plan, including, without limitation, the New Credit Agreement, the New Senior Second Lien Notes Indenture, the New Subordinated Notes Indenture, the New Harbinger Subordinated Notes Indenture, the New Junior Subordinated Notes Indenture, the Rights Offering Senior Subordinated Notes Indenture, the Registration Rights Agreement, the New Stockholders Agreement, and any other agreement or document related to or entered into in connection with any of the foregoing, shall become, and the Subscription and Backstop Purchase Agreement or Investment Agreement, as applicable shall remain, effective and binding in accordance with their respective terms and conditions upon the parties thereto, in each case without further notice to or order of the Bankruptcy Court, act or action under applicable law, regulation, order, or rule or the vote, consent, authorization or approval of any Person (other than as expressly required by such applicable agreement).

In the event that the Harbinger Investment Effective Date Condition is satisfied, upon the Effective Date, after giving effect to the transactions contemplated hereby, the authorized capital stock of Reorganized Bally shall be that number of shares of New Common Stock designated in the by-laws or certificate of incorporation of Bally, and the shares of New Common Stock outstanding shall consist solely of the shares of New Common Stock to be issued to the New Investors. Without limiting the effect of section 1145 of the Bankruptcy Code, in the event that the Backstop Rights Offering Effective Date Condition is satisfied, on the Effective Date, Reorganized Bally will enter into the Registration Rights Agreement with each Person (a) who by virtue of holding the New Common Stock and/or its relationship with Reorganized Bally could reasonably be deemed to be an "underwriter" or "affiliate" (as such terms are used within the meaning of applicable securities laws) of Reorganized Bally, and (b) who requests in writing that Reorganized Bally execute such agreement.

5.7 Exit Financing

On the Effective Date, the Reorganized Debtors shall be authorized to enter into the New Credit Agreement, as well as execute, deliver, file, record and issue any notes, documents, or agreements in connection therewith, in each case without further notice to or order of the Bankruptcy Court, act or action under applicable law, regulation, order, or rule or the vote, consent, authorization or approval of any Person (other than as expressly required by the New Credit Agreement).

5.8 Sources of Cash for Plan Distributions

Except as otherwise provided in this Plan or the Confirmation Order, all Cash necessary for the Reorganized Debtors to make payments pursuant to this Plan shall be obtained from (i) existing Cash balances, (ii) the operations of the Debtors and the Reorganized Debtors, (iii) the New Credit Agreement, and (iv) as applicable, the Rights Offering or the Harbinger Investment. The Reorganized Debtors may also make such payments using Cash received from their subsidiaries through the Reorganized Debtors' consolidated cash management systems.

5.9 New Stockholders Agreement

In the event that the Backstop Rights Offering Effective Date Condition is satisfied, Reorganized Bally shall be authorized and directed to enter into and consummate the transactions contemplated by the New Stockholders Agreement and such documents, and any agreement or document entered into in connection therewith, shall become effective and binding in accordance with their respective terms and conditions upon the parties thereto, in each case without further notice to or order of the Bankruptcy Court, act or action under applicable law, regulation, order, or rule or the vote, consent, authorization or approval of any Person (other than as expressly required by the New Stockholders Agreement). In the event that the Harbinger Investment Effective Date Condition is satisfied, there shall be no New Stockholders Agreement in connection with this Plan.

5.10 **Investment Agreement**

To the extent not previously authorized by Final Order of the Bankruptcy Court, on the Effective Date, the Reorganized Debtors shall be authorized to consummate and effectuate the transactions contemplated by the Investment Agreement without further notice to or order of the Bankruptcy Court, act or action under applicable law, regulation, order, or rule or the vote, consent, authorization or approval of any Person.

5.11 Old Affiliate Interests

Notwithstanding anything in this Plan to the contrary, on the Effective Date, the Old Affiliate Interests shall remain effective and outstanding and be owned and held by the same applicable Person(s) that held and/or owned such Interests immediately prior to the Effective Date. Each Affiliate Debtor shall continue to be governed by the terms and conditions of its applicable organizational documents as in effect immediately prior to the Effective Date, as amended or modified by this Plan.

5.12 Intercompany Claims

Notwithstanding anything in this Plan to the contrary, on the Effective Date, the Intercompany Claims of Debtors against Debtors shall be Reinstated or discharged and satisfied at the option of the Reorganized Debtors by contributions, distributions, or otherwise.

5.13 The Rights Offering and Subscription and Backstop Purchase Agreement

In the event that the Backstop Rights Offering Effective Date Condition is satisfied, the following sections of the Plan shall be applicable to the Debtors and Reorganized Debtors:

- Issuance of Rights. In accordance with Section 3.3.II.(b) of this Plan, each (a) of the Rights Offering Recipients existing as of the Rights Offering Recipients Record Date will receive Rights to purchase Rights Offering Senior Subordinated Notes with a principal amount equal to 27.9% of the amount of its Allowed Claim, and if any of such Rights provided to Holders of Allowed Class 6-A Claims are not timely exercised by the applicable recipient thereof (the "Unexercised Rights"), any and all other Class 6-A Rights Offering Recipients who have elected in their Subscription Form to exercise their share of the Rights may also elect in the Subscription Form to oversubscribe for such Unexercised Rights as described below. In accordance with the Subscription and Backstop Purchase Agreement, each Backstop Party shall fully exercise and subscribe for its share of the Rights prior to the Subscription Expiration Date, and shall be entitled, but not obligated, to oversubscribe for Unexercised Rights as more particularly set forth in the Subscription Form. After taking into account the exercise by any and all Class 6-A Rights Offering Recipients of their respective subscription and oversubscription rights described in the Solicitation Form, to the extent that any Rights have not been duly exercised, then the Backstop Parties shall exercise their share of such Unexercised Rights based on their respective Backstop Commitments, all in accordance with the provisions of the Subscription and Backstop Purchase Agreement.
- (b) Subscription Period. The Rights Offering shall commence on the applicable Subscription Commencement Date and shall expire on the applicable Subscription Expiration Date. Each Rights Offering Recipient that intends or desires to participate in the Rights Offering must affirmatively elect to exercise its Rights, and provide written notice thereof to the Debtors or other applicable Disbursing Agent, on or prior to the applicable Subscription Expiration Date in accordance with the terms of this Plan and the Subscription Form. On the Subscription Expiration Date applicable to the Class 6-A Rights Offering Recipients, all Unexercised Rights then remaining (after taking into account the exercise by any and all Class 6-A Rights Offering Recipients of their respective oversubscription rights) shall be allocated to, and exercised by, the Backstop Parties in accordance with the terms and conditions of the Subscription and Backstop Purchase Agreement.
 - (c) Exercise of Subscription Rights and Payment of Subscription Price.
- (i) On the applicable Subscription Commencement Date, the Debtors or other applicable Disbursing Agent will mail the Subscription Form to each Rights Offering Recipient existing as of the applicable Rights Offering Recipients Record Date, together with

appropriate instructions for the proper completion, due execution, and timely delivery of the Subscription Form, as well as instructions for the payment of the eventual Subscription Price for that portion of the Rights sought to be exercised by such Person on a subscription and oversubscription basis. The Debtors may adopt, with the prior written consent of the Backstop Parties, such additional detailed procedures consistent with the provisions of this Plan to more efficiently administer the exercise of the Rights.

- In order to exercise the Rights (and the right to oversubscribe for (ii) Unexercised Rights), each Rights Offering Recipient must return a duly completed Subscription Form (making a binding and irrevocable commitment to participate in the Rights Offering and indicating the aggregate principal amount of Rights Offering Senior Subordinated Notes for which such Rights Offering Recipient desires to oversubscribe) to the Debtors or other applicable Disbursing Agent so that such form is actually received by the Debtors or other applicable Disbursing Agent on or before the applicable Subscription Expiration Date. If the Debtors or other applicable Disbursing Agent for any reason do not receive from a given holder of Rights a duly completed Subscription Form on or prior to the applicable Subscription Expiration Date, then such holder shall be deemed to have forever and irrevocably relinquished and waived its right to participate in the Rights Offering. On the applicable Subscription Notification Date, the Debtors will notify each Rights Offering Recipient of its respective allocation of Rights Offering Senior Subordinated Notes, including any allocation for Rights Offering Senior Subordinated Notes as to which it exercised oversubscription rights (which shall be determined by allocating to such oversubscribing Rights Offering Recipient the lesser of (i) the aggregate principal amount of Rights Offering Senior Subordinated Notes for which such Rights Offering Recipient desires to oversubscribe, as indicated on its Subscription Form or (ii) its pro rata share of Unexercised Rights), and in the case of the Backstop Parties, the Debtors will notify each Backstop Party on or before the third day after the Subscription Expiration Date applicable to Class 6-A Rights Offering Recipients its share of the Unexercised Rights and the amount of Rights Offering Senior Subordinated Notes relating to such Unexercised Rights that such Backstop Party is obligated to purchase pursuant to the Subscription and Backstop Purchase Agreement. Each Rights Offering Recipient (other than the Backstop Parties, whose payments will be received by the Debtors on the Effective Date in accordance with the Subscription and Backstop Purchase Agreement) who has duly exercised any Rights must tender the Subscription Price to the Debtors or other applicable Disbursing Agent so that it is actually received on or prior to the applicable Subscription Payment Date. In the event the Debtors receive any payments for the exercise of Rights prior to the Effective Date, such payments shall be held in a separate account until the Effective Date. In the event the conditions to the Effective Date are not met or waived, such payments shall be returned to the Rights Offering Recipients that made them.
- (d) Detachment Restrictions; No Revocation. The Rights are not detachable. Any such detachment or attempted detachment will be null and void and the Debtors will not treat any purported transferee of the Rights separate from the Prepetition Senior Subordinated Notes as the holder of any Rights. Once a Rights Offering Recipient has exercised any of its Rights by properly executing and delivering a Subscription Form to the Debtors or other applicable Disbursing Agent, such exercise may only be revoked, rescinded or annulled in the sole discretion of the Debtors or Reorganized Debtors.

- (e) Distribution of Rights Offering Senior Subordinated Notes. On, or as soon as reasonably practicable after, the Effective Date (or, in the case of the Rights Offering Recipients in Class 6-B-1, the applicable Subscription Payment Date), the Reorganized Debtors or other applicable Disbursing Agent shall distribute the Rights Offering Senior Subordinated Notes purchased by each Rights Offering Recipient that has properly exercised its Rights and paid the Subscription Price.
- (f) Validity of Exercise of Subscription Rights. All questions concerning the timeliness, validity, form, and eligibility of any exercise, or purported exercise, of Rights shall be determined by the Debtors or Reorganized Debtors. The Debtors or Reorganized Debtors, in their discretion reasonably exercised in good faith, may waive any defect or irregularity, or permit a defect or irregularity to be corrected within such times as they may determine, or reject the purported exercise of any Rights. Subscription Forms shall be deemed not to have been received or accepted until all irregularities have been waived or cured within such time as the Debtors or Reorganized Debtors determine in their discretion reasonably exercised in good faith. The Debtors or Reorganized Debtors will use commercially reasonable efforts to give written notice to any Rights Offering Recipient regarding any defect or irregularity in connection with any purported exercise of Rights by such Person and may permit such defect or irregularity to be cured within such time as they may determine in good faith to be appropriate; provided, however, that neither the Debtors and Reorganized Debtors nor any of their Related Persons shall incur any liability for giving, or failing to give, such notification and opportunity to cure.
- (g) Rights Offering Proceeds. The proceeds of the Rights Offering will be used in order to fund Cash payments required to be made under this Plan and for general corporate purposes of the Reorganized Debtors.
- (h) Subscription and Backstop Purchase Agreement. The Debtors intend to assume the Subscription and Backstop Purchase Agreement and consummate the transactions contemplated in such agreement, including, without limitation, payment of the Backstop Commitment Fee, in each case in accordance with the terms and conditions thereof.

ARTICLE SIX PROVISIONS GOVERNING DISTRIBUTIONS

6.1 Distributions for Claims and Interests Allowed as of the Effective Date

Except as otherwise provided herein or as ordered by the Bankruptcy Court, distributions to be made on account of Claims and Interests that are Allowed Claims and Allowed Interests (if applicable) as of the Effective Date shall be made on the Effective Date or as soon thereafter as is practicable. Any distribution to be made on the Effective Date pursuant to this Plan shall be deemed as having been made on the Effective Date if such distribution is made on the Effective Date or as soon thereafter as is practicable. Any payment or distribution required to be made under this Plan on a day other than a Business Day shall be made on the next succeeding Business Day. Distributions on account of Disputed Claims and Disputed Interests

that first become Allowed Claims and Allowed Interests after the Effective Date shall be made pursuant to <u>Section 8.3</u> of this Plan.

6.2 No Postpetition Interest on Claims

Unless otherwise specifically provided for in this Plan or the Confirmation Order, or required by applicable bankruptcy law, postpetition interest shall not accrue or be paid on any Claims, and no Holder of a Claim (other than a Holder of an Administrative Claim under the DIP Credit Agreement with respect to such Administrative Claims) shall be entitled to interest accruing on or after the Petition Date on any Claim.

6.3 **Distributions by Reorganized Debtors**

Other than as specifically set forth below, the Reorganized Debtors or the Disbursing Agent shall make all distributions required to be distributed under this Plan. Distributions on account of Prepetition Lenders Claims, Prepetition Senior Notes Claims, Prepetition Senior Subordinated Notes Claims and Old Common Stock (if applicable) shall be made to the Prepetition Agent, the Prepetition Senior Notes Indenture Trustee, Prepetition Senior Subordinated Notes Indenture Trustee and the applicable transfer agent for the Old Common Stock, respectively. The Reorganized Debtors may employ or contract with other entities to assist in or make the distributions required by this Plan.

6.4 Delivery of Distributions and Undeliverable or Unclaimed Distributions

(a) Delivery of Distributions in General. Distributions to Holders of Allowed Claims and Allowed Interests shall be made at the addresses set forth in the Debtors' records or in care of their authorized agents, as appropriate, unless such addresses are superseded by proofs of claim or interest or transfers of claim filed pursuant to Bankruptcy Rule 3001.

(b) *Undeliverable and Unclaimed Distributions.*

- (i) Holding of Undeliverable and Unclaimed Distributions. If the distribution to any Holder of an Allowed Claim or Allowed Interest is returned to the Disbursing Agent as undeliverable or is otherwise unclaimed, no further distributions shall be made to such Holder unless and until the Disbursing Agent is notified in writing of such Holder's then current address.
- (ii) After Distributions Become Deliverable. The Disbursing Agent shall make all distributions that have become deliverable or have been claimed since the Initial Distribution Date as soon as practicable after such distribution has become deliverable or has been claimed.
- (iii) Failure to Claim Undeliverable Distributions. Any Holder of an Allowed Claim or Allowed Interest (or any successor or assignee or other Person or Entity claiming by, through, or on behalf of, such Holder) that does not assert a right pursuant to this Plan for an undeliverable or unclaimed distribution within one (1) year after the later of the Effective Date or the date such distribution is due shall be deemed to have forfeited its rights for such undeliverable or unclaimed distribution and shall be forever barred and enjoined from

asserting any such rights for an undeliverable or unclaimed distribution against the Debtors or their Estates, the Reorganized Debtors or their property. In such cases, any Cash for distribution on account of such rights for undeliverable or unclaimed distributions shall become the property of the Estates free of any restrictions thereon and notwithstanding any federal or state escheat laws to the contrary. Any New Common Stock, New Senior Second Lien Notes, New Subordinated Notes, New Junior Subordinated Notes, Rights Offering Senior Subordinated Notes, New Harbinger Subordinated Notes and/or other New Securities and Documents held for distribution on account of such Claim or Interest shall be canceled and of no further force or effect. Nothing contained in this Plan shall require the Debtors, Reorganized Debtors, or any Disbursing Agent to attempt to locate any Holder of an Allowed Claim or Allowed Interest.

6.5 **Record Date for Distributions**

The Disbursing Agent and the Reorganized Debtors will have no obligation to recognize the Transfer of, or the sale of any participation in, any Allowed Claim or Allowed Interest that occurs after the close of business on the Distribution Record Date, and will be entitled for all purposes herein to recognize and distribute securities, property, notices and other documents only to those Holders of Allowed Claims and Allowed Interests who are Holders of such Claims and Interests, or participants therein, as of the close of business on the Distribution Record Date. The Disbursing Agent and the Reorganized Debtors shall be entitled to recognize and deal for all purposes under this Plan with only those record holders stated on the official claims and interest register, or their books and records, as of the close of business on the Distribution Record Date.

6.6 Allocation of Plan Distributions Between Principal and Interest

To the extent that any Allowed Claim entitled to a distribution under this Plan is comprised of indebtedness and accrued but unpaid interest thereon, such distribution shall, to the extent permitted by applicable law, be allocated for income tax purposes to the principal amount of the Claim first and then, to the extent that the consideration exceeds the principal amount of the Claim, to the portion of such Claim representing accrued but unpaid interest.

6.7 **Means of Cash Payment**

Payments of Cash made pursuant to this Plan shall be in U.S. dollars and shall be made, at the option and in the sole discretion of the Reorganized Debtors, by (a) checks drawn on, or (b) wire transfer from, a domestic bank selected by the Reorganized Debtors. Cash payments to foreign creditors may be made, at the option of the Reorganized Debtors, in such funds and by such means as are necessary or customary in a particular foreign jurisdiction.

6.8 Withholding and Reporting Requirements

In connection with this Plan and all distributions hereunder, the Reorganized Debtors shall comply with all withholding and reporting requirements imposed by any federal, state, local, or foreign taxing authority, and all distributions hereunder shall be subject to any such withholding and reporting requirements. The Reorganized Debtors shall be authorized to take any and all actions that may be necessary or appropriate to comply with such withholding and reporting requirements. All persons holding Claims and Interests shall be required to

provide any information necessary to effect information reporting and the withholding of such taxes. Notwithstanding any other provision of this Plan to the contrary, (i) each Holder of an Allowed Claim or Allowed Interest shall have the sole and exclusive responsibility for the satisfaction and payment of any tax obligations imposed by any governmental unit, including income, withholding, and other tax obligations, on account of such distribution, and (ii) no distribution shall be made to or on behalf of such Holder pursuant to this Plan unless and until such Holder has made arrangements satisfactory to the Reorganized Debtors for the payment and satisfaction of such tax obligations. Any Cash, New Common Stock, New Senior Second Lien Notes, New Subordinated Notes (if any), New Harbinger Subordinated Notes (if any), New Junior Subordinated Notes (if any), Rights Offering Senior Subordinated Notes (if any), other New Securities and Documents and/or other consideration or property to be distributed pursuant to this Plan shall, pending the implementation of such arrangements, be treated as an undeliverable distribution pursuant to Section 6.4 of this Plan.

6.9 **Setoffs**

The Reorganized Debtors may, pursuant to section 553 of the Bankruptcy Code or applicable non-bankruptcy laws, but shall not be required to, set off against any Claim, the payments or other distributions to be made pursuant to this Plan in respect of such Claim, or claims of any nature whatsoever that the Debtors or the Reorganized Debtors may have against the Holder of such Claim; <u>provided</u>, <u>however</u>, that neither the failure to do so nor the allowance of any Claim hereunder shall constitute a waiver or release by the Reorganized Debtors of any such claim that the Debtors or the Reorganized Debtors may have against such Holder.

6.10 Fractional Shares

No fractional shares of New Common Stock shall be distributed by the Debtors or Reorganized Debtors. Where a fractional share would otherwise be called for, the actual issuance shall reflect a rounding up (in the case of .50 or more than .50) of such fraction to the nearest whole share of New Common Stock or a rounding down of such fraction (in the case of less than .50).

6.11 Surrender of Canceled Notes and Canceled Instruments of Securities

- (a) Generally. As a condition precedent to receiving any distribution pursuant to this Plan on account of an Allowed Claim or Allowed Interest evidenced by the instruments, securities, notes, or other documentation canceled pursuant to Section 5.5 of this Plan, the Holder of such Claim or Interest shall tender the applicable instruments, securities, notes or other documentation evidencing such Claim or Interest to the Reorganized Debtors or other applicable Disbursing Agent unless waived in writing by the Debtors or the Reorganized Debtors, as applicable.
- (b) Prepetition Notes. Each Holder of a Prepetition Senior Subordinated Notes Claim or a Prepetition Senior Notes Claim shall tender its respective Prepetition Senior Subordinated Notes or Prepetition Senior Notes relating to such Claim to the Reorganized Debtors or Disbursing Agent in accordance with written instructions to be provided to such Holders by the Reorganized Debtors or the Prepetition Senior Subordinated Notes Indenture

Trustee or the Prepetition Senior Notes Indenture Trustee as promptly as practicable following the Effective Date. Such instructions shall specify that delivery of such Prepetition Senior Subordinated Notes or Prepetition Senior Notes will be effected, and risk of loss and title thereto will pass, only upon the proper delivery of such Prepetition Senior Subordinated Notes or Prepetition Senior Notes with a letter of transmittal in accordance with such instructions. All surrendered Prepetition Senior Subordinated Notes and Prepetition Senior Notes shall be marked as canceled.

- (c) Old Common Stock. To the extent applicable, each Holder of Old Common Stock shall tender its Old Common Stock to the Reorganized Debtors or their designated agent in accordance with written instructions to be provided to such Holders by the Reorganized Debtors as promptly as practicable following the Effective Date. Such instructions shall specify that delivery of such Old Common Stock will be effected, and risk of loss and title thereto will pass, only upon the proper delivery of such Old Common Stock with a letter of transmittal in accordance with such instructions. All surrendered Old Common Stock shall be marked as canceled.
- (d) Failure to Surrender Security Instruments. Any Holder of a Prepetition Senior Notes Claim, Prepetition Senior Subordinated Notes Claim and/or Old Common Stock that fails to surrender or is deemed to have failed to surrender the applicable note or security required to be tendered hereunder within one (1) year after the Effective Date shall have its Claim and Interest and its distribution pursuant to this Plan on account of such Claim or Interest discharged and shall be forever barred from asserting any such Claim or Interest against the Reorganized Debtors or their respective property. In such cases, any Cash, New Common Stock, New Senior Second Lien Notes, New Subordinated Notes, New Junior Subordinated Notes, Rights Offering Senior Subordinated Notes, New Harbinger Subordinated Notes, other New Securities and Documents and/or other consideration or property held for distribution on account of such Claim or Interest shall be disposed of pursuant to Section 6.4(iii) of this Plan.

6.12 Lost, Stolen, Mutilated, or Destroyed Securities

In addition to any requirements under any applicable agreement and applicable law, any Holder of a Claim or Interest evidenced by a security or note that has been lost, stolen, mutilated, or destroyed shall, in lieu of surrendering such security or note to the extent required by this Plan, deliver to the Reorganized Debtors and other applicable Disbursing Agent: (x) evidence reasonably satisfactory to the Reorganized Debtors and other applicable Disbursing Agent of such loss, theft, mutilation, or destruction; and (y) such security or indemnity as may be required by the Reorganized Debtors and other applicable Disbursing Agent to hold such party harmless from any damages, liabilities, or costs incurred in treating such individual as a Holder of an Allowed Claim or Allowed Interest. Upon compliance with this Section 6.12 as determined by the Debtors or Reorganized Debtors by a Holder of a Claim or Interest evidenced by a security or note, such Holder shall, for all purposes under this Plan, be deemed to have surrendered such security or note to the Reorganized Debtors and other applicable Disbursing Agent.

6.13 Distributions from Old Common Stock Cash Amount

In the event that the Harbinger Investment Effective Date Condition is satisfied, for purposes of determining distributions to be made from the Old Common Stock Cash Amount, each Holder of an Allowed Interest represented by Old Common Stock will receive its Pro Rata share of the Old Common Stock Cash Portion, and each Holder of an Allowed 510(b) Equity Claim will receive its Pro Rata share of the Old Common Stock Remaining Cash Portion.

ARTICLE SEVEN TREATMENT OF EXECUTORY CONTRACTS AND UNEXPIRED LEASES

7.1 Assumption of Executory Contracts and Unexpired Leases

On the Effective Date, all executory contracts and unexpired leases of the Debtors will be deemed assumed in accordance with, and subject to, the provisions and requirements of sections 365 and 1123 of the Bankruptcy Code, except those executory contracts (including, without limitation, employment agreements) and unexpired leases that (i) have been rejected by order of the Bankruptcy Court, (ii) are the subject of a motion to reject pending on the Effective Date, (iii) are identified on Exhibit E-1 or Exhibit E-2 hereto (which Exhibits E-1 and E-2 may be amended by the Debtors (with the consent of the Majority Backstop Parties, if the Subscription and Backstop Purchase Agreement is then in effect, and/or the New Investors, if the Investment Agreement is then in effect) to add or remove executory contracts and unexpired leases by filing with the Bankruptcy Court amended Exhibits E-1 and E-2 and serving them on the affected contract parties at any time on or prior to five (5) days prior to the deadline set by the Bankruptcy Court for Filing objections to confirmation of this Plan), or (iv) are rejected pursuant to the terms of this Plan. Without amending or altering any prior order of the Bankruptcy Court approving the assumption or rejection of any executory contract or unexpired lease, entry of the Confirmation Order by the Bankruptcy Court shall constitute approval of such assumptions and rejections pursuant to sections 365(a) and 1123 of the Bankruptcy Code. To the extent any provision in any executory contract or unexpired lease assumed pursuant to this Plan (including, without limitation, any "change of control" provision) restricts or prevents, or purports to restrict or prevent, or is breached or deemed breached by, the applicable Reorganized Debtor's assumption of such executory contract or unexpired lease, then such provision shall be deemed modified such that the transactions contemplated by this Plan shall not entitle the nondebtor party thereto to terminate such executory contract or unexpired lease or to exercise any other default-related rights with respect thereto. Each executory contract and unexpired lease assumed pursuant to this Article VII shall revest in and be fully enforceable by the respective Reorganized Debtor in accordance with its terms, except as modified by the provisions of this Plan, any order of the Bankruptcy Court authorizing and providing for its assumption, or applicable law.

7.2 Claims Based on Rejection of Executory Contracts or Unexpired Leases

All proofs of claim with respect to Claims arising from or in connection with the rejection of executory contracts or unexpired leases, if any, must be filed with the Bankruptcy Court within thirty (30) days after the date of entry of an order of the Bankruptcy Court approving such rejection or, if listed in Exhibits E-1 or E-2, thirty (30) days after the date of

entry of the Confirmation Order. Any Claims arising from or in connection with the rejection of an executory contract or unexpired lease not filed within such time will be forever barred from assertion against the Debtors or Reorganized Debtors, their Estates, or property unless otherwise ordered by the Bankruptcy Court or provided for in this Plan.

7.3 Cure of Defaults of Assumed Executory Contracts and Unexpired Leases

Any monetary amounts by which each executory contract and unexpired lease to be assumed pursuant to this Plan is in default shall be satisfied, pursuant to section 365(b)(1) of the Bankruptcy Code, by payment of the default amount in Cash on the Effective Date or on such other terms as the parties to each such executory contract or unexpired lease may otherwise agree in writing. In the event of a dispute pertaining to assumption or assignment, the cure payments required by section 365(b)(1) of the Bankruptcy Code shall be made following the resolution of the dispute in accordance with Section 8.1(b) of this Plan.

7.4 Compensation and Benefit Programs

Except as otherwise expressly provided in this Plan or listed on Exhibit E-1 or Exhibit E-2 hereto, all employment and severance policies, and all compensation and benefit plans, policies, and programs of the Debtors applicable to their employees, retirees, and non-employee directors and the employees and retirees of its subsidiaries, including, without limitation, all savings plans, retirement plans, healthcare plans, disability plans, severance benefit plans, incentive plans (including, without limitation, the Prepetition Management Incentive Plan), life, and accidental death and dismemberment insurance plans, are treated as executory contracts under this Plan and on the Effective Date will be assumed pursuant to the provisions of sections 365 and 1123 of the Bankruptcy Code. Any payment obligations under any assumed employment contracts and benefit plans that have been or purport to have been accelerated as a result of the commencement of any Chapter 11 Case or the consummation of any transactions contemplated by this Plan shall be reinstated and such acceleration shall be rescinded and deemed not to have occurred.

7.5 Workers' Compensation Programs

Except as otherwise expressly provided in this Plan, as of the Effective Date, the Debtors and the Reorganized Debtors shall continue to honor their obligations under: (i) all applicable workers' compensation laws in states in which the Reorganized Debtors operate; and (ii) the Debtors' written contracts, agreements, agreements of indemnity, self-insurer workers' compensation bonds, and any other policies, programs, and plans regarding or relating to workers' compensation and workers' compensation insurance. All such contracts and agreements are treated as executory contracts under this Plan and on the Effective Date will be assumed pursuant to the provisions of sections 365 and 1123 of the Bankruptcy Code.

ARTICLE EIGHT PROCEDURES FOR RESOLVING DISPUTED, CONTINGENT AND UNLIQUIDATED CLAIMS

8.1 **Resolution of Disputed Claims**

- (a) Rejection Claims and 510(b) Equity Claims. Holders of Rejection Claims and 510(b) Equity Claims must File proofs of claims prior to the Bar Date. No later than the Claims Objection Deadline (unless extended by an order of the Bankruptcy Court), the Debtors or the Reorganized Debtors, as the case may be, shall file objections to such Claims with the Bankruptcy Court and serve such objections upon the Holders of such Claims to which objections are made. Nothing contained herein, however, shall limit the Reorganized Debtors' right to object to Claims, if any, filed or amended after the Claims Objection Deadline. The Debtors and the Reorganized Debtors shall be authorized to, and shall, resolve all Rejection Claims and 510(b) Equity Claims by withdrawing or settling such objections thereto, or by litigating to Final Order in the Bankruptcy Court, the validity, nature, and/or amount thereof.
- All Other Claims and Interests. Except as otherwise provided in this Plan, holders of Claims and Interests other than Holders of Rejection Claims and 510(b) Equity Claims shall not be required to file a proof of claim or proof of interest, and no such parties should file a proof of claim or proof of interest. Unless disputed by a holder of a Claim or Interest or by the Debtors, the amount set forth in the Schedules (if the Debtors are required to file Schedules) or in the books and records of the Debtors (if the Debtors are not required to file Schedules) shall constitute the amount of the Allowed Claim or Allowed Interest of such holder. If any such holder of a Claim or Interest disagrees with the Debtors' determination with respect to the Allowed amount of such Holder's Claim or Interest, such Holder must so advise the Debtors in writing (at any time whether prior to or after the Effective Date), in which event the Claim or Interest will be a Disputed Claim or Disputed Interest. The Debtors intend to attempt to resolve any such disputes consensually or through judicial means outside the Bankruptcy Court (and no further Bankruptcy Court order shall be required in connection with such resolutions). Nevertheless, no later than ninety (90) days after the Effective Date (or such later date as approved by this Court for cause shown after notice and hearing), the Debtors may, in their discretion, File with the Bankruptcy Court (or any other court of competent jurisdiction) an objection to the allowance of any Claim or Interest or any other appropriate motion or adversary proceeding with respect thereto. All such objections will be litigated to Final Order; provided, however, that the Debtors may compromise and settle, withdraw or resolve by any other method approved by the Bankruptcy Court, any objection to Claims and Interests without further order of the Bankruptcy Court.

8.2 No Distributions Pending Allowance

Notwithstanding any other provision of this Plan to the contrary, no payments or distributions of any kind or nature shall be made with respect to all or any portion of a Disputed Claim or Disputed Interest unless and until all objections to such Disputed Claim or Disputed Interest have been settled or withdrawn or have been determined by Final Order, and the Disputed Claim has become an Allowed Claim or the Disputed Interest has become an Allowed Interest.

8.3 Distributions on Account of Disputed Claims Once They Are Allowed and Additional Distributions on Account of Previously Allowed Claims

On each Quarterly Distribution Date (or such earlier date as determined by the Reorganized Debtors in their sole discretion but subject to Section 8.2 of this Plan), the Reorganized Debtors will make distributions (a) on account of any Disputed Claim that has become an Allowed Claim during the preceding calendar quarter or any Disputed Interest that has become an Allowed Interest during the preceding calendar quarter, and (b) on account of previously Allowed Claims and Allowed Interests of property that would have been distributed to the Holders of such Claim or Interest on the dates distributions previously were made to Holders of Allowed Claims and Allowed Interests in such Class had the Disputed Claims or Disputed Interests that have become Allowed Claims or Allowed Interests been Allowed on such dates. Such distributions will be made pursuant to the applicable provisions of Article III of this Plan. Holders of such Claims and Interests that are ultimately Allowed will also be entitled to receive, on the basis of the amount ultimately Allowed, the amount of any dividends or other distributions, if any, received on account of the shares of New Common Stock, New Subordinated Notes, New Junior Subordinated Notes, Rights Offering Senior Subordinated Notes, and New Harbinger Subordinated Notes, between the date such Claim or Interest is Allowed and the date such stock or notes are actually distributed to the Holders of such Allowed Claim or Allowed Interest.

8.4 Reserve for Disputed 510(b) Equity Claims and Old Common Stock

Prior to making any distributions of Cash to Holders of Allowed 510(b) Equity Claims or Allowed Old Common Stock, the applicable Disbursing Agent shall establish appropriate reserves for Disputed 510(b) Equity Claims and Disputed Interests to withhold from any such distributions one hundred percent (100%) of distributions to which Holders of Disputed 510(b) Equity Claims and Disputed Old Common Stock would be entitled under this Plan as of such date if such Disputed 510(b) Equity Claims and Disputed Old Common Stock were Allowed in the amount asserted by the Holder thereof in its respective timely filed proof of claim or proof of interest; provided, however, that the Debtors and the Reorganized Debtors shall have the right to file a motion seeking to estimate such amounts. The Disbursing Agent shall also establish appropriate reserves for Disputed Claims in other Classes as it determines necessary and appropriate.

ARTICLE NINE CONFIRMATION AND CONSUMMATION OF THE PLAN

9.1 **Conditions to Confirmation**

It shall be a condition precedent to confirmation of this Plan that the Bankruptcy Court shall have entered a Confirmation Order reasonably acceptable in form and substance to the Debtors, the Majority Backstop Parties (if the Subscription and Backstop Purchase Agreement is then in effect) and the New Investors (if the Investment Agreement is then in effect).

9.2 **Conditions to Effective Date**

Each of the following is a condition precedent to the occurrence of the Effective Date:

- (a) The Effective Date shall have occurred on or before the Applicable Outside Date.
- (b) The Confirmation Order confirming this Plan, as such Plan may have been amended or modified, in form and substance reasonably satisfactory to (i) the Debtors, (ii) the Majority Backstop Parties if the Subscription and Backstop Purchase Agreement is then in effect, and/or (iii) the New Investors if the Investment Agreement is then in effect, shall have been entered and docketed by the Bankruptcy Court, and such order shall have become a Final Order and shall provide that:
- (i) the Debtors and Reorganized Debtors are authorized to take all actions necessary or appropriate to enter into, implement, and consummate the contracts, instruments, releases, leases, indentures, and other agreements or documents contemplated by or described in this Plan;
- (ii) the provisions of the Confirmation Order are non-severable and mutually dependent;
- (iii) the Reorganized Debtors are authorized to issue, as applicable, (A) the New Common Stock, (B) the New Senior Second Lien Notes, (C) subject to satisfaction of the Harbinger Investment Effective Date Condition (as defined below), the New Harbinger Senior Subordinated Notes, (D) subject to satisfaction of the Backstop Rights Offering Effective Date Condition (as defined below), the New Subordinated Notes, the New Junior Subordinated Notes, and the Rights Offering Senior Subordinated Notes, and (E) any other New Securities and Documents, and enter into the New Credit Agreement, in each case without further notice to or order of the Bankruptcy Court, act or action under applicable law, regulation, order, or rule or the vote, consent, authorization or approval of any Person (other than as expressly required by such applicable agreement);
- (iv) the New Common Stock, the New Senior Second Lien Notes, any New Harbinger Subordinated Notes, any New Subordinated Notes, any New Junior Subordinated Notes, any Rights Offering Senior Subordinated Notes (and offer of the Rights for the Rights Offering Period to Holders of Allowed Class 6-A and 6-B-1 Claims), the Investment Agreement and any other New Securities and Documents issued or deemed issued under this Plan in exchange for Claims against the Debtors, or principally in exchange for such Claims and partly for cash or property, are exempt from registration under the Securities Act of 1933 pursuant to section 1145 of the Bankruptcy Code, except to the extent that (i) the New Common Stock is issued to the New Investors and/or (ii) Holders of any of the foregoing are "underwriters," as that term is defined in section 1145 of the Bankruptcy Code; and
- (v) the Debtors, the Reorganized Debtors, the Prepetition Noteholders Committee, the Prepetition Senior Noteholders, the Prepetition Senior Subordinated Noteholders, the Prepetition Senior Notes Indenture Trustee, the Prepetition Senior Subordinated Notes

Indenture Trustee, the Backstop Parties, the New Investors (if the Harbinger Investment Effective Date Condition has been satisfied) and Liberation (if the Harbinger Investment Effective Date Condition has been satisfied) and their respective Related Persons shall be deemed to have (a) solicited acceptances of the Plan and subscriptions to the Rights Offering in good faith and in compliance with the applicable provisions of the Bankruptcy Code, including without limitation, sections 1125 and 1126(b) of the Bankruptcy Code, and any applicable non-bankruptcy law, rule or regulation governing the adequacy of disclosure in connection with such solicitation and (b) participated in good faith and in compliance with the applicable provisions of the Bankruptcy Code in the offer and issuance of any securities under the Plan and, therefore, are not, and on account of such offer, issuance and solicitation will not be, liable at any time for any violation of any applicable law, rule, or regulation governing the solicitation of acceptances or rejections of the Plan or the offer and issuance of any securities under the Plan, including pursuant to the Subscription and Backstop Purchase Agreement and/or the Investment Agreement.

- (c) The following agreements, in form and substance reasonably acceptable to the Reorganized Debtors and Majority Backstop Parties (in the event the Backstop Rights Offering Effective Date Condition is satisfied) or the New Investors (in the event the Harbinger Investment Effective Date Condition is satisfied) shall have been executed and delivered by the Reorganized Debtors and such other parties deemed necessary by the Reorganized Debtors and the Majority Backstop Parties or the New Investors, as applicable, and all conditions precedent thereto shall have been satisfied:
 - (i) the New Credit Agreement and all related documents provided for therein or contemplated thereby;
 - (ii) the New Senior Second Lien Notes Indenture;
 - (iii) the New Junior Subordinated Notes Indenture (if any);
 - (iv) the New Subordinated Notes Indenture (if any);
 - (v) the New Harbinger Subordinated Notes Indenture (if any);
 - (vi) the Rights Offering Senior Subordinated Notes Indenture (if any);
 - (vii) the Registration Rights Agreement (if any); and
 - (viii) the New Stockholders Agreement (if any).
- (i) If the Investment Agreement is then in effect, the Harbinger Investment shall have been made in Cash to Reorganized Bally no later than the Applicable Outside Date, and the New Investors shall have purchased and received all shares of New Common Stock to be distributed to them under the Investment Agreement in consideration of such Harbinger Investment (the condition described in this clause (i), the "Harbinger Investment Effective Date Condition"), or (ii) if the Investment Agreement has terminated or expired on or before the Applicable Outside Date, without the Harbinger Investment having been made, or if the Investors for any reason fail on or before the Applicable Outside Date, to make the Harbinger Investment in accordance with the

provisions of the Investment Agreement and the Plan, then in connection with the Rights Offering, the Debtors shall have received in Cash the aggregate subscription payments that the Backstop Parties are obligated to pay for their share of the Rights Offering Senior Subordinated Notes, which aggregate subscription payments shall be no less than \$72,000,000 (the condition described in this clause (ii), the "Backstop Rights Offering Effective Date Condition"). Without limiting the rights of the Backstop Parties under the Subscription and Backstop Purchase Agreement to the payment of the Backstop Commitment Fee in any other circumstance, if the Harbinger Investment Effective Date Condition is satisfied, the Backstop Commitment Fee shall be paid in full in Cash.

- (d) The Amended Certificate of Incorporation and By-laws and other amended organizational documents, as necessary, shall have been filed with the applicable authority of each Debtor's respective jurisdiction of incorporation or formation in accordance with such jurisdiction's applicable laws.
- (e) All actions, documents, certificates and agreements necessary to implement this Plan shall have been effected or executed and delivered to the required parties and, to the extent required under this Plan or the Confirmation Order, filed with the applicable governmental authorities in accordance with applicable laws.

9.3 **Waiver of Conditions**

Each of the conditions set forth in <u>Section 9.2</u> of this Plan may be waived in whole or in part by the Debtors, in consultation with and after obtaining the consent of (i) the New Investors, if the Investment Agreement is then in effect and/or (ii) the Majority Backstop Parties, if the Subscription and Backstop Purchase Agreement is then in effect, in each case, given in writing or on the record in the Chapter 11 Cases, without any other notice to parties in interest or notice to or order of the Bankruptcy Court and without a hearing; provided, however, if the Harbinger Investment Effective Date Condition has been satisfied, the New Investors in their sole discretion may waive the requirement in Section 9.2(b) that the Confirmation Order become a Final Order without any other notice to parties in interest or notice to or order of the Bankruptcy Court and without a hearing, and provided, further, that if the Backstop Rights Offering Effective Date Condition has been satisfied, the Majority Backstop Parties in their sole discretion may waive the requirement in Section 9.2(b) that the Confirmation Order become a Final Order without any other notice to parties in interest or notice to or order of the Bankruptcy Court and without a hearing. The failure to satisfy or waive a condition to the Effective Date may be asserted by the Debtors or the Reorganized Debtors regardless of the circumstances giving rise to the failure of such condition to be satisfied. The failure of a Debtor or Reorganized Debtor to exercise any of the foregoing rights shall not be deemed a waiver of any other rights, and each right shall be deemed an ongoing right that may be asserted at any time.

9.4 Consequences of Non-Occurrence of Effective Date

If each of the conditions to consummation and the occurrence of the Effective Date has not been satisfied or duly waived on or before the first Business Day that is thirty (30) days after the Confirmation Date, or such later date as shall be consented to by the Debtors and (i) the New Investors, if the Investment Agreement is then in effect, and/or (ii) the Majority

Backstop Parties, if the Subscription and Backstop Purchase Agreement is then in effect, then upon motion by the Debtors and upon notice to such parties in interest as the Bankruptcy Court may direct, the Confirmation Order shall be vacated by the Bankruptcy Court. If the Confirmation Order is so vacated, the Plan shall be null and void in all respects, and nothing contained in the Plan shall constitute a waiver or release of any Claims by, or against, any of the Debtors or the allowance of any Administrative Expense or Claim.

ARTICLE TEN EFFECT OF PLAN CONFIRMATION

10.1 Binding Effect; Plan Binds All Holders of Claims and Interests

On the Effective Date, and effective as of the Effective Date, this Plan shall, and shall be deemed to, be binding upon and inure to the benefit of the Debtors, all present and former Holders of Claims against and Interests in any Debtor, and their respective successors and assigns, including, but not limited to, the Reorganized Debtors, regardless of whether any such Holder failed to vote to accept or reject this Plan or affirmatively voted to reject this Plan.

10.2 Releases and Related Injunctions

- (a) Releases by the Debtors. Subject to the provisions of Section 10.9 hereof, effective as of the Effective Date, for good and valuable consideration, the adequacy of which is hereby confirmed, the Debtors and Reorganized Debtors, in their individual capacities and as debtors in possession, will be deemed to forever release, waive, and discharge all claims, obligations, suits, judgments, damages, demands, debts, rights, Causes of Action, and liabilities (other than the rights of the Debtors or Reorganized Debtors to enforce this Plan and the contracts, instruments, releases, indentures, and other agreements or documents delivered under or in connection with this Plan or assumed pursuant to this Plan), whether liquidated or unliquidated, fixed or contingent, matured, or unmatured, known or unknown, foreseen, or unforeseen, then existing or thereafter arising, in law, equity, or otherwise that are based in whole or part on any act, omission, transaction, event, or other occurrence taking place on or prior to the Effective Date in any way relating to the Debtors, the Reorganized Debtors, the Chapter 11 Cases, the Disclosure Statement, the First Restructuring Support Agreement, the Second Restructuring Support Agreements (if the Harbinger Investment Effective Date Condition is satisfied or as otherwise provided in the Investment Agreement), the Investment Agreement (if the Harbinger Investment Effective Date Condition is satisfied or as otherwise provided in the Investment Agreement) or this Plan (or the solicitation of votes on this Plan), and that could have been asserted by or on behalf of the Debtors, their Estates or the Reorganized Debtors against any of the Released Parties, in each case without further notice to or order of the Bankruptcy Court, act or action under applicable law, regulation, order, or rule or the vote, consent, authorization or approval of any Person.
- (b) Releases by Holders of Claims and Interests of Claims Against Debtor-Related Released Parties. Subject to the provisions of Section 10.9 hereof, effective as of the Effective Date, for good and valuable consideration and in consideration for the obligations of the Debtors and the Reorganized Debtors under this Plan and the property, securities, contracts, instruments, releases, and other agreements or documents to be delivered in connection with this

Plan, to the fullest extent permissible under applicable law, the Holders of Claims or Interests, and each of their respective Related Persons, will be deemed to completely and forever release, waive, void, extinguish, and discharge each of the Debtor-Related Released Parties, from any and all claims, demands, debts, rights, Causes of Action, or liabilities (other than the right to enforce the Debtors' or the Reorganized Debtors' obligations under this Plan, and the contracts, instruments, releases, agreements, and documents delivered under this Plan), whether liquidated or unliquidated, fixed or contingent, matured or unmatured, known or unknown, foreseen or unforeseen, then existing or thereafter arising, in law, equity, or otherwise, that are based in whole or in part on any act or omission, transaction, event, or other occurrence taking place on or prior to the Effective Date in any way relating to the Debtors, the Reorganized Debtors, the Chapter 11 Cases, the Disclosure Statement, the First Restructuring Support Agreement, the Second Restructuring Support Agreements (if the Harbinger Investment Effective Date Condition is satisfied or as otherwise provided in the Investment Agreement), the Investment Agreement (if the Harbinger Investment Effective Date Condition is satisfied or as otherwise provided in the Investment Agreement) or this Plan (or the solicitation of votes on this Plan) and that could have been asserted by or on behalf of (whether directly or derivatively), or against, any or all of the Debtor-Related Released Parties, in each case without further notice to or order of the Bankruptcy Court, act or action under applicable law, regulation, order, or rule or the vote, consent, authorization or approval of any Person.

Releases by Holders of Claims and Interests of Claims Against Other Released Parties. In addition to, and without limiting, the release provisions of Section 10.2(b) in favor of the Debtor-Related Released Parties, effective as of the Effective Date, for good and valuable consideration and in consideration for the obligations of the Debtors and the Reorganized Debtors under this Plan and the property, securities, contracts, instruments, releases, and other agreements or documents to be delivered in connection with this Plan, to the fullest extent permissible under applicable law, the Holders of Claims or Interests, and each of their respective Related Persons, will be deemed to completely and forever release, waive, void, extinguish, and discharge each of the Other Released Parties, from any and all claims, demands, debts, rights, Causes of Action, or liabilities (other than the right to enforce the Debtors' or the Reorganized Debtors' obligations under this Plan, and the contracts, instruments, releases, agreements, and documents delivered under this Plan), whether liquidated or unliquidated, fixed or contingent, matured or unmatured, known or unknown, foreseen or unforeseen, then existing or thereafter arising, in law, equity, or otherwise, that are based in whole or in part on any act or omission, transaction, event, or other occurrence taking place on or prior to the Effective Date in any way relating to the Chapter 11 Cases, the Disclosure Statement, the First Restructuring Support Agreement, the Second Restructuring Support Agreements (if the Harbinger Investment Effective Date Condition is satisfied or as otherwise provided in the Investment Agreement), the Investment Agreement (if the Harbinger Investment Effective Date Condition is satisfied or as otherwise provided in the Investment Agreement) or this Plan (or the solicitation of votes on this Plan) and that could have been asserted by or on behalf (whether directly or derivatively) of the Debtors, their Estates or the Reorganized Debtors or against any of the Other Released Parties, in each case without further notice to or order of the Bankruptcy Court, act or action under applicable law, regulation, order, or rule or the vote, consent, authorization or approval of any Person.

(d) *Injunction Related to Releases*. The Confirmation Order will permanently enjoin the commencement or prosecution by any Person or Entity, whether directly, derivatively or otherwise, of any claims, obligations, suits, judgments, damages, demands, debts, rights, Causes of Action, or liabilities released pursuant to this Plan, including but not limited to the claims, obligations, suits, judgments, damages, demands, debts, rights, Causes of Action, or liabilities released in this Section 10.2.

10.3 **Discharge of Claims**

To the fullest extent provided under section 1141(d)(1)(A) and other applicable provisions of the Bankruptcy Code, except as otherwise expressly provided by this Plan or the Confirmation Order, all consideration distributed under this Plan shall be in exchange for, and in complete satisfaction, settlement, discharge, and release of, all Claims of any kind or nature whatsoever against the Debtors or any of their assets or properties, and regardless of whether any property shall have been distributed or retained pursuant to this Plan on account of such Claims. Except as otherwise expressly provided by this Plan or the Confirmation Order, upon the Effective Date, the Debtors, and each of them, shall be deemed discharged and released under and to the fullest extent provided under section 1141(d)(1)(A) of the Bankruptcy Code from any and all Claims of any kind or nature whatsoever, including, but not limited to, demands and liabilities that arose before the Confirmation Date, and all debts of the kind specified in section 502(g), 502(h), or 502(i) of the Bankruptcy Code.

10.4 Preservation of Rights of Action; Settlement of Litigation Claims

- Preservation of Rights of Action. Except as otherwise provided in this (a) Plan, the Confirmation Order, or in any document, instrument, release, or other agreement entered into in connection with this Plan or approved by order of the Bankruptcy Court, in accordance with section 1123(b) of the Bankruptcy Code, the Debtors and their Estates shall retain the Litigation Claims. The Reorganized Debtors, as the successors in interest to the Debtors and the Estates, may, and shall have the exclusive right to, enforce, sue on, settle, compromise, transfer or assign (or decline to do any of the foregoing) any or all of the Litigation Claims, including, without limitation, any and all derivative actions pending or otherwise existing against the Debtors as of the Effective Date. Notwithstanding the foregoing, the Debtors and the Reorganized Debtors shall not file, commence, or pursue any claim, right, or cause of action under section 547 of the Bankruptcy Code; provided, however, that, notwithstanding any statute of limitations, the Debtors and Reorganized Debtors shall have the right to assert or raise such Causes of Action (a) as defenses or counterclaims (up to the amount asserted in the Claims against the Debtors) with respect to any Disputed Claim, and (b) in connection with the Claims objection process with respect to a Claim that is not an Allowed Claim, in which case such Causes of Action can be raised as an objection to such Claim and not as defenses or counterclaims.
- (b) Settlement of Litigation Claims. At any time after the Confirmation Date and before the Effective Date, notwithstanding anything in this Plan to the contrary, the Debtors may settle any or all of the Litigation Claims with the approval of the Bankruptcy Court pursuant to Bankruptcy Rule 9019. After the Effective Date, the Reorganized Debtors may, and shall have the exclusive right to, compromise and settle any Claims against them and claims they may

have against other Person or Entity, including, without limitation, the Litigation Claims, without notice to or approval from the Bankruptcy Court, including, without limitation, any and all derivative actions pending or otherwise existing against the Debtors as of the Effective Date.

10.5 Exculpation and Limitation of Liability

Subject to the provisions of Section 10.9 hereof, none of the Released Parties shall have or incur any liability to, or be subject to any right of action by, any Holder of a Claim or an Interest, or any other party in interest, or any of their respective agents, employees, representatives, financial advisors, attorneys, or agents acting in such capacity, or affiliates, or any of their successors or assigns, or any other Released Party, for any act or omission in connection with, relating to, or arising out of, the Chapter 11 Cases, formulating, negotiating, or implementing this Plan, the First Restructuring Support Agreement, the Second Restructuring Support Agreements (if the Harbinger Investment Effective Date Condition is satisfied or as otherwise provided in the Investment Agreement), the Investment Agreement (if the Harbinger Investment Effective Date Condition is satisfied or as otherwise provided in the Investment Agreement), the prepetition or postpetition solicitation of acceptances of this Plan, the prepetition or postpetition solicitation of subscriptions with respect to the Rights Offering, the Subscription and Backstop Purchase Agreement and the acts taken thereunder, the Investment Agreement (if the Harbinger Investment Effective Date Condition is satisfied or as otherwise provided in the Investment Agreement) and the acts taken thereunder, the pursuit of confirmation of this Plan, the confirmation of this Plan, the consummation of this Plan, or the administration of this Plan or the property to be distributed under this Plan, except for their respective actions that constitute gross negligence or willful misconduct as determined by a Final Order entered by a court of competent jurisdiction. For avoidance of doubt, none of the Released Parties shall have or incur any liability to, or be subject to any right of action by, any Holder of a Claim or an Interest, or any other party in interest, or any of their respective agents, employees, representatives, financial advisors, attorneys, or agents acting in such capacity, or affiliates, or any of their successors or assigns, or any other Released Party, for any Plan-Related Claim. Without limiting the foregoing, the Released Parties shall in all respects be entitled to reasonably rely upon the advice of counsel with respect to their duties and responsibilities under this Plan.

10.6 **Injunctions**

(a) Except as otherwise provided in this Plan or in any document, instrument, release, or other agreement entered into in connection with this Plan or approved by order of the Bankruptcy Court, the Confirmation Order shall provide, among other things, that from and after the Effective Date all Persons or Entities who have held, hold, or may hold Claims against or Interests in the Debtors are (i) permanently enjoined from taking any of the following actions against the Estate(s), or any of their property, on account of any such Claims or Interests and (ii) permanently enjoined from taking any of the following actions against any of the Debtors, the Reorganized Debtors or their property on account of such Claims or Interests: (A) commencing or continuing, in any manner or in any place, any action, or other proceeding; (B) enforcing, attaching, collecting, or recovering in any manner any judgment, award, decree or order; (C) creating, perfecting, or enforcing any Lien or encumbrance; (D) asserting a setoff or right of subrogation of any kind against any debt, liability or obligation due to the Debtors; and (E) commencing or continuing, in any manner or in any place, any action that does not comply with

or is inconsistent with the provisions of this Plan; <u>provided</u>, <u>however</u>, that nothing contained herein shall preclude such Persons or Entities from exercising their rights pursuant to and consistent with the terms of this Plan and the contracts, instruments, releases, indentures, and other agreements or documents delivered under or in connection with this Plan.

(b) By accepting distributions pursuant to this Plan, each Holder of an Allowed Claim or Allowed Interest will be deemed to have specifically consented to the injunctions set forth in this <u>Section 10.6</u>.

10.7 Term of Bankruptcy Injunction or Stays

All injunctions or stays provided for in the Chapter 11 Cases under section 105 or 362 of the Bankruptcy Code, or otherwise, and in existence on the Confirmation Date, shall remain in full force and effect until the Effective Date.

10.8 Termination of Subordination Rights and Settlement of Related Claims

The classification and manner of satisfying all Claims and Interests under this Plan take into consideration all subordination rights, whether arising by contract or under general principles of equitable subordination, section 510(b) or 510(c) of the Bankruptcy Code, or otherwise. All subordination rights that a Holder of a Claim or Interest may have with respect to any distribution to be made pursuant to this Plan will be discharged and terminated, and all actions related to the enforcement of such subordination rights will be permanently enjoined. Accordingly, distributions pursuant to this Plan to Holders of Allowed Claims and Allowed Interests will not be subject to payment to a beneficiary of such terminated subordination rights, or to levy, garnishment, attachment, or other legal process by a beneficiary of such terminated subordination rights; provided, however, that nothing contained herein shall preclude any Person or Entity from exercising their rights pursuant to and consistent with the terms of this Plan and the contracts, instruments, releases, indentures, and other agreements or documents delivered under or in connection with this Plan.

10.9 Limitations of Applicability of Releases and Exculpatory Provisions to Governmental Entities

Except with respect to Plan-Related Claims, nothing in the Confirmation Order or the Plan (i) shall effect a release of any claim by the United States Government or any of its agencies or any state and local authority whatsoever, including, without limitation, any claim arising under the Internal Revenue Code, the environmental laws or any criminal laws of the United States or any state and local authority against the Released Parties, but excluding any Plan-Related Claims, (ii) enjoin the United States or any state or local authority from bringing any claim, suit, action or other proceedings against the Released Parties for any liability whatever, including, without limitation, any claim, suit or action arising under the Internal Revenue Code, the environmental laws or any criminal laws of the United States or any state or local authority, but excluding any Plan-Related Claims, or (iii) exculpate any Released Party from any liability to the United States Government or any of its agencies or any state and local authority whatsoever, including, without limitation, any liabilities arising under the Internal Revenue Code, the environmental laws or any criminal laws of the United States or any state and

local authority against the Released Parties, but excluding any liabilities in respect of any Plan-Related Claims. Notwithstanding anything to the contrary in the Confirmation Order or the Plan, the Securities and Exchange Commission ("SEC") expressly reserves its right to continue to investigate, and, in its sole discretion, prosecute and enforce any and all claims against any or all of the Debtors or the Reorganized Debtors arising from any prepetition violations by any Debtor of any of the U.S. securities laws other than Plan-Related Claims (collectively, the "Reserved SEC Claims"), including, without limitation, any claims for disgorgement of any benefits received by any Debtor as a result of any such violations and any claims for penalties imposed by the SEC in respect of any such violations. For avoidance of doubt, pursuant to Section 10.2(b) of the Plan, all Plan-Related Claims of the United States Government or any of its agencies or any state and local authority whatsoever shall be, and hereby are, released, waived and discharged. Nothing in the Confirmation Order or the Plan shall result in the discharge of any Reserved SEC Claims, and the SEC expressly reserves its rights to assert that any and all Reserved SEC Claims are non-dischargeable as against the Reorganized Debtors pursuant to Sections 1141(d)(6)(a) and 523(a)(2)(A) of the Bankruptcy Code. The SEC has advised the Court and the Debtors that as of the entry of this Order, it has not yet determined whether to assert any Reserved SEC Claims against any or all of the Debtors or Reorganized Debtors.

ARTICLE ELEVEN RETENTION OF JURISDICTION

Pursuant to sections 105(c) and 1142 of the Bankruptcy Code and notwithstanding entry of the Confirmation Order and the occurrence of the Effective Date, the Bankruptcy Court will retain jurisdiction over all matters arising in, arising under, and/or related to, the Chapter 11 Cases and this Plan to the fullest extent permitted by law, including, among other things, jurisdiction to:

- (a) Allow, disallow, determine, liquidate, classify, estimate, or establish the priority or secured or unsecured status of any Claim or Interest, including the resolution of any request for payment of any Administrative Claim and the resolution of any objections to the allowance or priority of Claims or Interests;
- (b) Resolve any matters related to the assumption, assumption and assignment, or rejection of any executory contract or unexpired lease to which any Debtor is a party or with respect to which any Debtor or the Reorganized Debtor may be liable and to hear, determine, and, if necessary, liquidate any Claims arising therefrom;
- (c) Ensure that distributions to Holders of Allowed Claims or Allowed Interests (if any) are accomplished pursuant to the provisions of this Plan;
- (d) Decide or resolve any motions, adversary proceedings, contested, or litigated matters and any other matters and grant or deny any applications involving the Debtors that may be pending on the Effective Date;
- (e) Enter such orders as may be necessary or appropriate to implement or consummate the provisions of this Plan and all contracts, instruments, releases, and other

agreements or documents created in connection with this Plan, the Disclosure Statement, or the Confirmation Order;

- (f) Resolve any cases, controversies, suits, or disputes that may arise in connection with the consummation, interpretation, or enforcement of this Plan, including, without limitation, the Rights Offering or any other contract, instrument, release, or other agreement or document that is executed or created pursuant to this Plan, or any Entity's rights arising from or obligations incurred in connection with this Plan or such documents;
- (g) Modify this Plan before or after the Effective Date pursuant to section 1127 of the Bankruptcy Code or modify the Disclosure Statement, the Confirmation Order, or any contract, instrument, release, or other agreement or document created in connection with this Plan, the Disclosure Statement or the Confirmation Order, or remedy any defect or omission or reconcile any inconsistency in any Bankruptcy Court order, this Plan, the Disclosure Statement, the Confirmation Order, or any contract, instrument, release, or other agreement or document created in connection with this Plan, the Disclosure Statement, or the Confirmation Order, in such manner as may be necessary or appropriate to consummate this Plan;
- (h) Hear and determine all applications for compensation and reimbursement of expenses of Professionals under this Plan or under sections 327, 330, 331, 363, 503(b), 1103, and 1129(c)(9) of the Bankruptcy Code, <u>provided</u>, <u>however</u>, that from and after the Effective Date the payment of fees and expenses of the Reorganized Debtors, including counsel fees, shall be made in the ordinary course of business and shall not be subject to the approval of the Bankruptcy Court;
- (i) Issue injunctions, enter and implement other orders, or take such other actions as may be necessary or appropriate to restrain interference by any Person or Entity with consummation, implementation, or enforcement of this Plan or the Confirmation Order;
- (j) Hear and determine causes of action by or on behalf of the Debtors or the Reorganized Debtors;
- (k) Hear and determine matters concerning state, local and federal taxes in accordance with sections 346, 505, and 1146 of the Bankruptcy Code;
- (l) Enter and implement such orders as are necessary or appropriate if the Confirmation Order is for any reason or in any respect modified, stayed, reversed, revoked, or vacated, or distributions pursuant to this Plan are enjoined or stayed;
- (m) Determine any other matters that may arise in connection with or relate to this Plan, the Disclosure Statement, the Confirmation Order, or any contract, instrument, release, or other agreement, or document created in connection with this Plan, the Disclosure Statement or the Confirmation Order:
- (n) Enforce all orders, judgments, injunctions, releases, exculpations, indemnifications, and rulings entered in connection with the Chapter 11 Cases;

- (o) Hear and determine all matters related to (i) the property of the Estates from and after the Confirmation Date and (ii) the activities of the Reorganized Debtors;
- (p) Hear and determine the Litigation Claims by or on behalf of the Debtors or Reorganized Debtors;
- (q) Hear and determine such other matters as may be provided in the Confirmation Order or as may be authorized under the Bankruptcy Code; and
 - (r) Enter an order closing the Chapter 11 Cases.

ARTICLE TWELVE MISCELLANEOUS PROVISIONS

12.1 Effectuating Documents and Further Transactions

Each of the Debtors or the Reorganized Debtors is authorized to execute, deliver, file, or record such contracts, instruments, releases, consents, certificates, resolutions, programs and other agreements and/or documents and take such acts and actions as may be reasonable, necessary or appropriate to effectuate, implement, consummate and/or further evidence the terms and conditions of this Plan, any notes or securities issued pursuant to this Plan, and any transactions described in or contemplated by this Plan.

12.2 **Authority to Act**

Prior to, on, or after the Effective Date (as appropriate), all matters expressly provided for under this Plan that would otherwise require approval of the stockholders, security holders, officers, directors, partners, managers, members or other owners of one or more of the Debtors or the Reorganized Debtors shall be deemed to have occurred and shall be in effect prior to, on, or after the Effective Date (as appropriate) pursuant to the applicable law of the states in which the Debtors or Reorganized Debtors are formed, without any requirement of further vote, consent, approval, authorization or other action by such stockholders, security holders, officers, directors, partners, managers, members or other owners of such entities or notice to, order of, or hearing before the Bankruptcy Court.

12.3 **Exemption from Transfer Taxes**

Pursuant to section 1146(c) of the Bankruptcy Code, (a) the issuance, transfer, or exchange (or deemed issuance, transfer or exchange) of notes or equity securities under this Plan, including, without limitation, the New Senior Second Lien Notes, the New Subordinated Notes, the New Harbinger Subordinated Notes, the New Junior Subordinated Notes, the New Common Stock, the Rights, the Rights Offering Senior Subordinated Notes and the other New Securities and Documents; (b) the creation of any mortgage, deed of trust, Lien, pledge, or other security interest; (c) the making or assignment of any lease or sublease; or (d) the making or delivery of any deed or other instrument of transfer under, in furtherance of, or in connection with, this Plan (including, without limitation, any merger agreements, agreements of consolidation, restructuring, disposition, liquidation, or dissolution, deeds, bills of sale, and transfers of tangible property) will not be subject to any stamp tax, recording tax, personal property tax, real estate

transfer tax, sales tax, use tax, transaction privilege tax (including, without limitation such taxes on prime contracting and owner-builder sales), privilege taxes (including, without limitation, privilege taxes on construction contracting with regard to speculative builders and owner builders), and other similar taxes. Unless the Bankruptcy Court orders otherwise, all sales, transfers, and assignments of owned and leased property approved by the Bankruptcy Court on or prior to the Effective Date, shall be deemed to have been in furtherance of, or in connection with, this Plan.

12.4 Bar Dates for Administrative Claims

To the extent necessary, the Confirmation Order will establish an Administrative Claims Bar Date for filing Administrative Claims, except for Administrative Claims arising under the DIP Credit Agreement as provided in Section 3.1(a)(ii) of this Plan and Administrative Claims for Professional Fees (which shall be subject to the Professional Fees Bar Date). Holders of alleged Administrative Claims not paid prior to the Effective Date shall submit proofs of Claim on or before such Administrative Claims Bar Date or forever be barred from doing so (unless such alleged Administrative Claim is incurred in the ordinary course of business by the Debtors and is not yet past-due, in which case the applicable Administrative Claims Bar Date shall be thirty (30) days after such due date or as otherwise ordered by the Bankruptcy Court). The notice of Confirmation to be delivered pursuant to Bankruptcy Rule 3020(c) and 2002(f) will set forth such date and constitute notice of this Administrative Claims Bar Date. The Debtors and the Reorganized Debtors shall have thirty (30) days (or such longer period as may be allowed by order of the Bankruptcy Court) following the Administrative Claims Bar Date to review and File objections to such Administrative Claims, if necessary, and the Bankruptcy Court shall hear and determine the amount of such Administrative Claims.

12.5 **Payment of Statutory Fees**

All fees payable pursuant to section 1930 of title 28, United States Code, as determined by the Bankruptcy Court at the Confirmation Hearing, shall be paid on the Effective Date.

12.6 Amendment or Modification of the Plan

Subject to section 1127 of the Bankruptcy Code and, to the extent applicable, sections 1122, 1123, and 1125 of the Bankruptcy Code, the Debtors reserve the right to alter, amend, or modify this Plan at any time prior to or after the Confirmation Date but prior to the substantial consummation of this Plan; provided, however, that (i) any alternations, amendments or other modifications to the Plan affecting the transactions contemplated by this Plan if consummated on the basis of the satisfaction of the Harbinger Investment Effective Date Condition shall not be made (x) without the written consent of the New Investors, which consent shall not be unreasonably withheld, and (y) if such alterations, amendments or other modifications affect the treatment of the Claims of the Holders of the Prepetition Subordinated Notes, reduce the amount of the Harbinger Investment or materially affect the capital structure of the Reorganized Debtors, without the written consent of the Majority Backstop Parties, which consent shall not be unreasonably withheld, and (ii) any alternations, amendments or other modifications affecting the transactions contemplated by this Plan if consummated on the basis

of the satisfaction of the Backstop Rights Offering Effective Date Condition shall not be made without the written consent of the Majority Backstop Parties, which consent shall not be unreasonably withheld. A Holder of a Claim that has accepted this Plan shall be deemed to have accepted this Plan, as altered, amended or modified, if the proposed alteration, amendment or modification does not materially and adversely change the treatment of the Claim of such Holder.

12.7 Severability of Plan Provisions

If, prior to the Confirmation Date, any term or provision of this Plan is determined by the Bankruptcy Court to be invalid, void, or unenforceable, the Bankruptcy Court will have the power to alter and interpret such term or provision to make it valid or enforceable to the maximum extent practicable, consistent with the original purpose of the term or provision held to be invalid, void, or unenforceable, and such term or provision will then be applicable as altered or interpreted. Notwithstanding any such holding, alteration or interpretation, the remainder of the terms and provisions of this Plan will remain in full force and effect and will in no way be affected, impaired, or invalidated by such holding, alteration, or interpretation. The Confirmation Order will constitute a judicial determination and will provide that each term and provision of this Plan, as it may have been altered or interpreted in accordance with the foregoing, is valid and enforceable pursuant to its terms.

12.8 Successors and Assigns

This Plan shall be binding upon and inure to the benefit of the Debtors, and their respective successors and assigns, including, without limitation, the Reorganized Debtors. The rights, benefits, and obligations of any Person or Entity named or referred to in this Plan shall be binding on, and shall inure to the benefit of, any heir, executor, administrator, successor, or assign of such Person or Entity.

12.9 Revocation, Withdrawal, or Non-Consummation

The Debtors reserve the right to revoke or withdraw this Plan as to any or all of the Debtors prior to the Confirmation Date and to file subsequent plans of reorganization. If the Debtors revoke or withdraw this Plan as to any or all of the Debtors, or if confirmation or consummation as to any or all of the Debtors does not occur, then, with respect to such Debtors, except as otherwise provided by the Debtors, (a) this Plan shall be null and void in all respects, (b) any settlement or compromise embodied in this Plan (including the fixing or limiting to an amount certain any Claim or Interest or Class of Claims or Interests), assumption or rejection of executory contracts or leases affected by this Plan, and any document or agreement executed pursuant to this Plan shall be deemed null and void, and (c) nothing contained in this Plan shall (i) constitute a waiver or release of any Claims by or against, or any Interests in, such Debtors or any other Person or Entity, (ii) prejudice in any manner the rights of such Debtors or any other Person or Entity, or (iii) constitute an admission of any sort by the Debtors or any other Person or Entity.

12.10 **Notice**

All notices, requests, and demands to or upon the Debtors or the Reorganized Debtors to be effective shall be in writing and, unless otherwise expressly provided herein, shall be deemed to have been duly given or made when actually delivered or, in the case of notice by facsimile transmission, when received and telephonically confirmed, addressed as follows:

If to any Debtor or Reorganized Debtor:

Bally Total Fitness 8700 Bryn Mawr Avenue Chicago, Illinois 60631

Attention: Marc D. Bassewitz

Fax: (773) 399-0126

with a copy (which shall not constitute notice hereunder), to: Latham & Watkins LLP Sears Tower, Suite 5800 233 South Wacker Drive Chicago, Illinois 60606 Attention: David S. Heller

Tel: (312) 876-7700 Fax: (312) 993-9767

If to the Backstop Parties:

Akin Gump Strauss Hauer & Feld LLP 590 Madison Avenue New York, NY 10022

Attn: Daniel Golden, Esq. (dgolden@akingump.com)
Attn: David Botter, Esq. (dbotter@akingump.com)

Facsimile: (212) 872-1002

If to the New Investors:

Kasowitz, Benson, Torres & Friedman LLP 1633 Broadway New York, New York 10019 Attention: Andrew K. Glenn

Tel: (212) 507-1700 Fax: (212) 507-1800

Kramer Levin Naftalis & Frankel LLP 1177 Avenue of the Americas New York, New York 10036 Attention: Shari K. Krouner

Tel: (212) 715-9222 Fax: (212) 715-8000

12.11 Governing Law

Except to the extent that the Bankruptcy Code, the Bankruptcy Rules or other federal law is applicable, or to the extent that an exhibit or schedule to this Plan provides otherwise, the rights and obligations arising under this Plan shall be governed by, and construed and enforced in accordance with, the laws of New York, without giving effect to the principles of conflicts of law of such jurisdiction.

12.12 Tax Reporting and Compliance

The Reorganized Debtors are hereby authorized, on behalf of each of the Debtors, to request an expedited determination under section 505(b) of the Bankruptcy Code of the tax liability of the Debtors for all taxable periods ending after the Petition Date through, and including, the Effective Date.

12.13 Schedules

All exhibits and schedules to this Plan, including the Exhibits and Plan Schedules, are incorporated and are a part of this Plan as if set forth in full herein.

12.14 Filing of Additional Documents

On or before substantial consummation of this Plan, the Debtors shall File such agreements and other documents as may be necessary or appropriate to effectuate and further evidence the terms and conditions of this Plan.

12.15 No Strict Construction

This Plan is the product of extensive discussions and negotiations between and among, inter alia, the Debtors, the Backstop Parties, the Prepetition Noteholders Committee and

the New Investors. Each of the foregoing was represented by counsel of its choice who either (a) participated in the formulation and documentation of, or (b) was afforded the opportunity to review and provide comments on, this Plan, the Disclosure Statement, Exhibits and Plan Schedules, and the agreements and documents ancillary or related thereto. Accordingly, unless explicitly indicated otherwise, the general rule of contract construction known as "contra proferentem" shall not apply to the construction or interpretation of any provision of this Plan, the Disclosure Statement, Exhibits and Plan Schedules, and the documents ancillary and related thereto.

12.16 Conflicts

In the event that a provision of the Disclosure Statement conflicts with a provision of this Plan, the terms of this Plan shall govern and control to the extent of such conflict.

12.17 **Dissolution of Committee**

The Creditors' Committee appointed in the Chapter 11 Cases pursuant to section 1102 of the Bankruptcy Code, if any, shall be dissolved on the Confirmation Date.

12.18 Fees and Expenses

From and after the Effective Date, the Reorganized Debtors shall, in the ordinary course of business and without the necessity for any approval by the Bankruptcy Court, pay the reasonable fees and expenses of professional persons thereafter incurred, including those fees and expenses incurred in connection with the implementation and consummation of this Plan.

Dated: New York, New York September 17, 2007

Respectfully Submitted,

BALLY TOTAL FITNESS HOLDING CORPORATION

By: /s/ Don R. Kornstein Name: Don R. Kornstein

Title: Chief Restructuring Officer

Each Debtor Listed on Appendix 1

By: <u>/s/ Don R. Kornstein</u> Name: Don R. Kornstein

Title: Chief Restructuring Officer

David S. Heller Richard A. Levy Keith A. Simon Sears Tower, Suite 5800 233 South Wacker Drive Chicago, Illinois 60606-6401 Telephone: (312) 876-7700 Facsimile: (312) 993-9767

-and-

Henry P. Baer, Jr. (HB-3866) Joseph Furst, III (JF-6136) 885 Third Avenue, Suite 1000 New York, New York 10022 Telephone: (212) 906-1200

Counsel for Debtors and Debtors-in-Possession

Appendix 1

Bally ARA Corporation

Bally Fitness Franchising, Inc.

Bally Franchise RSC, Inc.

Bally Franchising Holdings, Inc.

Bally Real Estate I LLC

Bally REFS West Hartford, LLC

Bally Sports Clubs, Inc.

Bally Total Fitness Corporation

Bally Total Fitness Franchising, Inc.

Bally Total Fitness Holding Corporation

Bally Total Fitness International, Inc.

Bally Total Fitness of California, Inc.

Bally Total Fitness of Colorado, Inc.

Bally Total Fitness of Connecticut Coast, Inc.

Bally Total Fitness of Connecticut Valley, Inc.

Bally Total Fitness of Greater New York, Inc.

Bally Total Fitness of Minnesota, Inc.

Bally Total Fitness of Missouri, Inc.

Bally Total Fitness of Philadelphia, Inc.

Bally Total Fitness of Rhode Island, Inc.

Bally Total Fitness of the Mid-Atlantic, Inc.

Bally Total Fitness of the Midwest, Inc.

Bally Total Fitness of the Southeast, Inc.

Bally Total Fitness of Toledo, Inc.

Bally Total Fitness of Upstate New York, Inc.

BTF Cincinnati Corporation

BTF Europe Corporation

BTF Indianapolis Corporation

BTF Minneapolis Corporation

BTF/CFI. Inc.

BTFCC, Inc.

BTFF Corporation

Greater Philly No. 1 Holding Company Greater Philly No. 2 Holding Company Health & Tennis Corporation of New York Holiday Health Clubs of the East Coast, Inc.

Holiday/Southeast Holding Corp.

Jack LaLanne Holding Corp. New Fitness Holding Co., Inc.

Nycon Holding Co., Inc.

Rhode Island Holding Company

Tidelands Holiday Health Clubs, Inc.

U.S. Health, Inc.

Exhibit 2

[Redline of Plan of Reorganization]

IN THE UNITED STATES BANKRUPTCY COURT FOR THE SOUTHERN DISTRICT OF NEW YORK

In re)	Chapter 11
BALLY TOTAL FITNESS OF GREATER NEW YORK, INC., <u>et al</u> ., ¹)))	Case No. 07-12395 (BRL) (Jointly Administered)
Debtors.)	Honorable Burton R. Lifland

FIRST AMENDED JOINT PREPACKAGED CHAPTER 11 PLAN OF REORGANIZATION OF BALLY TOTAL FITNESS HOLDING CORPORATION AND ITS AFFILIATE DEBTORS

Dated: August September 17, 2007

New York, New York

The Debtors in these proceedings are: Bally Total Fitness Holding Corporation, Bally Total Fitness Corporation, Bally ARA Corporation, Bally Fitness Franchising, Inc., Bally Franchise RSC, Inc., Bally Franchising Holdings, Inc., Bally Real Estate I LLC, Bally REFS West Hartford, LLC, Bally Sports Clubs, Inc., Bally Total Fitness Franchising, Inc., Bally Total Fitness of California, Inc., Bally Total Fitness of Colorado, Inc., Bally Total Fitness of Connecticut Coast, Inc., Bally Total Fitness of Connecticut Valley, Inc., Bally Total Fitness of Greater New York, Inc., Bally Total Fitness of Minnesota, Inc., Bally Total Fitness of Missouri, Inc., Bally Total Fitness of Philadelphia, Inc., Bally Total Fitness of Rhode Island, Inc., Bally Total Fitness of the Mid-Atlantic, Inc., Bally Total Fitness of the Midwest, Inc., Bally Total Fitness of the Southeast, Inc., Bally Total Fitness of Toledo, Inc., Bally Total Fitness of Upstate New York, Inc., BTF Cincinnati Corporation, BTF Europe Corporation, BTF Indianapolis Corporation, BTF Minneapolis Corporation, BTF/CFI, Inc., BTFCC, Inc., BTFF Corporation, Greater Philly No. 1 Holding Company, Greater Philly No. 2 Holding Company, Health & Tennis Corporation of New York, Holiday Health Clubs of the East Coast, Inc., Holiday/Southeast Holding Corp., Jack LaLanne Holding Corp., New Fitness Holding Co., Inc., Nycon Holding Co., Inc., Rhode Island Holding Company, Tidelands Holiday Health Clubs, Inc., and U.S. Health, Inc.

TABLE OF CONTENTS

ARTICLE C	NE DEFINED TERMS AND RULES OF INTERPRETATION	1
1.1	Defined Terms	1
1.2	Exhibits and Plan Schedules	<u>19<mark>20</mark></u>
1.3	Rules of Interpretation and Computation of Time	19 <u>20</u>
ARTICLE T	WO CLASSIFICATION OF CLAIMS AND INTERESTS	<u>2021</u>
2.1	Unclassified Claims	21 22
2.2	Unimpaired Classes of Claims and Interests	21 22
2.3	Impaired Classes of Claims	22 23
2.4	Impaired Classes of Claims and Interests	22 23
ARTICLE T	HREE TREATMENT OF CLAIMS AND INTERESTS	22 23
3.1	Unclassified Claims	22 23
3.2	Unimpaired Classes of Claims	<u>24<mark>25</mark></u>
3.3	Impaired Classes of Claims and Interests	25 26
3.4	Unimpaired Class of Interests.	28 29
3.5	Special Provision Regarding Unimpaired Claims	28 <u>29</u>
ARTICLE F	OUR ACCEPTANCE OR REJECTION OF THE PLAN	29
4.1	Impaired Classes of Claims Entitled to Vote	29
4.2	Acceptance by an Impaired Class	
4.3	Presumed Acceptances by Unimpaired Classes	
4.4	Presumed Rejection by Certain Impaired Classes	
4.5	Confirmation Pursuant to Section 1129(b) of the Bankruptcy Code	
4.6	Elimination of Vacant Classes	
ARTICLE F	IVE MEANS FOR IMPLEMENTATION OF THE PLAN	<u>30<mark>31</mark></u>
5.1	Limited Substantive Consolidation for Purposes of Treating Impaired	2021
<i>5</i> 2	Claims Other Than Impaired Claims Against Only Bally	
5.2	Restructuring Transactions Continued Legal Existence and Vesting of Assets in the Reorganized	31 <u>34</u>
5.3	Debtors	32
5.4	Corporate Governance, Directors, Officers, and Corporate Action	
5.5	Cancellation of Notes, Instruments, Debentures, Preferred Stock and	·
	Common Stock	34 35
5.6	Issuance of New Securities and Related Documentation	35 <mark>36</mark>
5.7	Exit Financing	
5.8	Sources of Cash for Plan Distributions	

5	5.9 New Stockholders Agreement	36 <u>37</u>
5	5.10 Investment Agreement	36 <u>37</u>
5	5.11 Old Affiliate Interests	36 37
5	5.12 Intercompany Claims	
5	The Rights Offering and Subscription and Backstop Purchase Agreement	
ARTICI	LE SIX PROVISIONS GOVERNING DISTRIBUTIONS	<u>3940</u>
6	Distributions for Claims and Interests Allowed as of the Effective Date	<u>3940</u>
ϵ	No Postpetition Interest on Claims.	4 <u>041</u>
ϵ	5.3 Distributions by Reorganized Debtors	_40 <u>41</u>
6	Delivery of Distributions and Undeliverable or Unclaimed Distributions	<u>4041</u>
6	6.5 Record Date for Distributions	<u>4142</u>
6	Allocation of Plan Distributions Between Principal and Interest	41 <u>42</u>
6	5.7 Means of Cash Payment	41 <u>42</u>
6	5.8 Withholding and Reporting Requirements	<u>41<mark>42</mark></u>
6	5.9 Setoffs	42 <u>43</u>
6	5.10 Fractional Shares	<u>4243</u>
6	5.11 Surrender of Canceled Notes and Canceled Instruments of Securities	<u>4243</u>
6	5.12 Lost, Stolen, Mutilated, or Destroyed Securities	<u>4344</u>
	LE SEVEN TREATMENT OF EXECUTORY CONTRACTS AND JNEXPIRED LEASES	44 <u>45</u>
7	7.1 Assumption of Executory Contracts and Unexpired Leases	_44 <u>45</u>
7	7.2 Claims Based on Rejection of Executory Contracts or Unexpired Leases	
7	Cure of Defaults of Assumed Executory Contracts and Unexpired Leases	
7	7.4 Compensation and Benefit Programs	
7	7.5 Workers' Compensation Programs	
	LE EIGHT PROCEDURES FOR RESOLVING DISPUTED, CONTINGENT	45 <u>46</u>
8	Resolution of Disputed Claims	45 <u>46</u>
	3.2 No Distributions Pending Allowance	
	B.3 Distributions on Account of Disputed Claims Once They Are Allowed and	10 <u>17</u>
	Additional Distributions on Account of Previously Allowed Claims	4 <mark>647</mark>
8	Reserve for Disputed 510(b) Equity Claims and Old Common Stock.	
ARTICI	LE NINE CONFIRMATION AND CONSUMMATION OF THE PLAN	
C	9.1 Conditions to Confirmation	<u> 4748</u>
	0.2 Conditions to Effective Date	
	9.3 Waiver of Conditions	50 51
	0.4 Consequences of Non-Occurrence of Effective Date	50 <u>51</u>

ARTICLE TE	EN EFFECT OF PLAN CONFIRMATION	<u>5052</u>
10.1	Binding Effect; Plan Binds All Holders of Claims and Interests	<u>5052</u>
10.2	Releases and Related Injunctions	
10.3	Discharge of Claims	
10.4	Preservation of Rights of Action; Settlement of Litigation Claims	53 <u>54</u>
10.5	Exculpation and Limitation of Liability	
10.6	Injunctions	
10.7	Term of Bankruptcy Injunction or Stays	<u>55<mark>56</mark></u>
10.8	Termination of Subordination Rights and Settlement of Related Claims	
ARTICLE EL	EVEN RETENTION OF JURISDICTION	<u>55<mark>57</mark></u>
ARTICLE TV	VELVE MISCELLANEOUS PROVISIONS	<u>57<u>59</u></u>
12.1	Effectuating Documents and Further Transactions	<u>57</u> <u>59</u>
12.2	Authority to Act	<u>57<mark>59</mark></u>
12.3	Exemption from Transfer Taxes	
12.4	Bar Dates for Administrative Claims	<u>5860</u>
12.5	Payment of Statutory Fees.	
12.6	Amendment or Modification of the Plan	<u>58<mark>61</mark></u>
12.7	Severability of Plan Provisions	<u>5961</u>
12.8	Successors and Assigns	
12.9	Revocation, Withdrawal, or Non-Consummation	59 62
12.10	Notice	506 2
12.11	Governing Law	
12.12	•	
12.13		
12.14	Filing of Additional Documents	
	No Strict Construction	
	Conflicts	
	Dissolution of Committee	
	Fees and Expenses	

EXHIBITS

Exhibit A Amended Certificate of Incorporation of Reorganized Bally Exhibit B Amended By-Laws of Reorganized Bally Exhibit C Subscription and Backstop Purchase Agreement Exhibit D DIP Credit Agreement Exhibit E-1 Rejection Claims List For Bally Exhibit E-2 Rejection Claims List For Affiliate Debtors Exhibit F Form of New Credit Agreement Exhibit G-1 New Senior Second Lien Notes Indenture (if the Harbinger Investment Effective Date Condition is satisfied) Exhibit G-2 New Senior Second Lien Notes Indenture (if the Backstop Rights Offering Condition is satisfied) Exhibit H Form of Rights Offering Senior Subordinated Notes Indenture, New Subordinated Notes Indenture and New Junior Subordinated Notes Indenture Exhibit I New Stockholders Agreement Exhibit J Prepetition Management Incentive Plan Exhibit K Registration Rights Agreement Exhibit L First Restructuring Support Agreement Exhibits M-1 and M-Second Restructuring Support Agreements Exhibit N Investment Agreement Exhibit O New Harbinger Subordinated Notes Indenture

PLAN SCHEDULES

Plan Schedule 1.1(a) Non-Exclusive List of Litigation Claims, including derivative actions

Plan Schedule 1.1(b) Non-Exclusive List of 510(b) Equity Claims

INTRODUCTION

Bally Total Fitness Holding Corporation ("Bally") and the other above-captioned debtors and debtors-in-possession (collectively, the "Debtors") propose the following first amended prepackaged joint plan of reorganization for the resolution of the outstanding claims against and interests in the Debtors. Reference is made to the Disclosure Statement (as that term is defined herein and distributed contemporaneously herewith) for a discussion of the Debtors' history, business, properties and operations, projections for those operations, risk factors, a summary and analysis of this Plan (as that term is defined herein), and certain related matters including, among other things, certain tax matters, the securities to be issued under this Plan and the proposed substantive consolidation of the Debtors' cases for certain limited purposes. Subject to certain restrictions and requirements set forth in 11 U.S.C. § 1127 and Fed. R. Bankr. P. 3019, the Debtors reserve the right to alter, amend, modify, revoke or withdraw this Plan prior to its substantial consummation.

ARTICLE ONE DEFINED TERMS AND RULES OF INTERPRETATION

1.1 **Defined Terms**. Capitalized terms used in this Plan shall have the meanings set forth below. Any term that is not otherwise defined herein, but that is used in the Bankruptcy Code or the Bankruptcy Rules, will have the meaning given to that term in the Bankruptcy Code or the Bankruptcy Rules, as applicable.

510(b) Equity Claims means any Claim subordinated pursuant to section 510(b) of the Bankruptcy Code, including, without limitation, those claims arising from the rescission of a purchase or sale of Old Common Stock or rights relating to such Old Common Stock, or any Claim for damages arising from the purchase or sale of Old Common Stock or any Claim for reimbursement, contribution, or indemnification arising from or relating to any such Claims. A non-exclusive list of the 510(b) Equity Claims is attached to this Plan as Plan Schedule 1.1(b).

Administrative Claim means a Claim for costs and expenses of administration of the Chapter 11 Cases that are Allowed under section 503(b), 507(b), or 1114(e)(2) of the Bankruptcy Code, including, without limitation: (a) any actual and necessary costs and expenses incurred after the Petition Date of preserving the Debtors' Estates and operating the businesses of the Debtors (such as wages, salaries, and commissions for services and payments for inventory, leased equipment, and premises); (b) compensation for legal, financial, advisory, accounting, and other services and reimbursement of expenses Allowed by the Bankruptcy Court under section 327, 330, 331, 363, or 503(b) of the Bankruptcy Code to the extent incurred prior to the Effective Date; (c) all fees and charges assessed against the Debtors' Estates under section 1930, chapter 123, of title 28, United States Code; (d) any amounts and obligations owed and outstanding under the DIP Credit Agreement; (e) the Backstop Commitment Fee, to the extent payable, and, as set forth more fully in, and in accordance with, the Subscription and Backstop Purchase Agreement or the Investment Agreement, as applicable, the reasonable fees and expenses of the Backstop Parties, the New Investors and Liberation in connection with the Chapter 11 Cases and the negotiation, confirmation and implementation of this Plan and the transactions contemplated hereby; (f) the Prepetition Senior Notes Indenture Amendment Fee; (g) the Prepetition Senior Notes Indenture Trustee Fees, (h) the Prepetition Senior Subordinated

CH\<u>961384.15</u>961384.20

Creditors' Committee means the official committee of unsecured creditors of the Debtors, if any, appointed by the United States Trustee in the Chapter 11 Cases pursuant to section 1102 of the Bankruptcy Code.

Debtor(s) means, individually, any of the Debtors and, collectively, all of the above-captioned debtors and debtors-in-possession.

Estates, the Reorganized Debtors, the New Investors and their respective Related Persons other than any Related Person of any Debtor or Reorganized Debtor who is not entitled to be indemnified, by contract or applicable law, from any Debtor or Reorganized Debtor from and against any claims, obligations, suits, judgments, damages, demands, debts, rights, Causes of Action and liabilities that may be asserted by any Holders of Claims or Holders of any Interests against such Related Person in its or his capacity as such.

DIP Agent means the administrative agent under the DIP Credit Agreement, and its successors and assigns.

DIP Credit Agreement means the debtor-in-possession secured credit agreement substantially in the form attached hereto as <u>Exhibit D</u>.

DIP Credit Agreement Obligations means any loans and other indebtedness and obligations of any or all of the Debtors to any or all of the DIP Agent and the DIP Lenders pursuant to the DIP Credit Agreement and the other DIP Credit Documents.

DIP Credit Documents means all of the agreements, documents and instruments entered into in connection with the DIP Credit Agreement.

DIP Lenders means each of the financial institutions party to the DIP Credit Agreement and identified as "Lenders" therein, and their respective successors and assigns.

DIP Lenders Claims means any and all "Obligations" as defined in the DIP Credit Agreement, including, without limitation, the DIP Credit Agreement Obligations and any and all other Claims of, and any other obligations and liabilities owed to, the DIP Agent and DIP Lenders arising from or related to the DIP Credit Agreement and any other DIP Credit Document, including, without limitation, the DIP Credit Agreement Obligations.

DIP Lenders Liens means any security interests and Liens granted by any Debtor to the DIP Agent and/or any DIP Lender in order to secure the repayment of any DIP Lenders Claims.

DIP Loan Facility means the credit facility in the aggregate principal amount not to exceed \$292,000,000 to be provided to the Debtors during the Chapter 11 Cases pursuant to the DIP Credit Agreement if the Debtors determine a DIP Credit Agreement is necessary, the proceeds of which would be used to refinance the Prepetition Lenders Claims and for general corporate purposes.

Disbursing Agent means the Reorganized Debtors or any party designated by the Reorganized Debtors to serve as disbursing agent under this Plan. For purposes of distributions under this Plan to the Holders of Allowed Prepetition Lenders Claims, Allowed Prepetition Senior Notes Claims, and Allowed Prepetition Senior Subordinated Notes Claims and, in the event that the Harbinger Investment Effective Date Condition is satisfied, the Old Common Stock, the Prepetition Agent, the Prepetition Senior Notes Indenture Trustee, the Prepetition Senior Subordinated Notes Indenture Trustee and the transfer agent for the Old Common Stock (which transfer agent will a successor transfer agent to be retained by the Debtors prior to the commencement of the Confirmation Hearing, and will not be the existing transfer agent), respectively, will be and shall act as the Disbursing Agent.

Disclosure Statement means that certain disclosure statement (including all exhibits and schedules thereto) dated as of the date hereof (and any amendments or supplements thereto), relating to this Plan, which was distributed by Bally on or about such date to the Prepetition Senior Noteholders and the Prepetition Senior Subordinated Noteholders existing as of the Voting Record Date in connection with the prepetition solicitation of their votes pursuant to section 1126(b) of the Bankruptcy Code.

Disputed Claim means a Claim, or any portion thereof, that (a) if the Debtors are required by order of the Bankruptcy Court to file Schedules, (i) has not been Scheduled by the Debtors or has been Scheduled at zero, or has been Scheduled as contingent, unliquidated or disputed and for which no proof of claim has been timely filed with the Bankruptcy Court or (ii) is in excess of the amount Scheduled as other than disputed, contingent or unliquidated, (b) is the subject of an objection or request for estimation filed in the Bankruptcy Court and which objection or request for estimation has not been withdrawn or overruled by a Final Order of the Bankruptcy Court, (c) is a 510(b) Equity Claim and/or (d) is otherwise disputed by any of the Debtors in accordance with applicable law, which dispute has not been withdrawn, resolved or overruled by Final Order.

Disputed Interest means an Interest, or any portion thereof, that (a) is the subject of an objection or request for estimation filed in the Bankruptcy Court and which objection or request for estimation has not been withdrawn or overruled by a Final Order of the Bankruptcy Court, and/or (b) is otherwise disputed by any of the Debtors in accordance with applicable law, which dispute has not been withdrawn, resolved or overruled by Final Order.

Distribution Record Date means the Confirmation Effective Date.

Effective Date means the Business Day that this Plan becomes effective as provided in Article IX hereof.

Entity means an "entity" as defined in section 101(15) of the Bankruptcy Code.

Estate(s) means, individually, the estate of each of the Debtors and, collectively, the estates of all of the Debtors created under section 541 of the Bankruptcy Code.

Exchange Act means the Securities Exchange Act of 1934, 15 U.S.C. §§ 78a et seq., as amended.

Majority Backstop Parties means Backstop Parties holding in excess of 50% of the principal amount of Prepetition Senior Subordinated Notes held by all of the Backstop Parties.

Master Ballot means the ballot distributed to holders of record of the Prepetition Senior Notes and Prepetition Senior Subordinated Notes to record the votes of the beneficial holders of the Prepetition Senior Notes and Prepetition Senior Subordinated Notes, respectively, as of the Voting Record Date.

New Agent means the administrative agent under the New Credit Agreement, and its successors and assigns.

New Common Stock means the shares of common stock of Reorganized Bally authorized to be issued pursuant to this Plan and the Amended Certificate of Incorporation and By-Laws.

New Credit Agreement means that certain secured credit agreement between Reorganized Bally, as borrower, those entities identified as "Guarantors" in the New Credit Agreement, New Agent and New Lenders (as amended, modified, or supplemented from time to time, with the prior written consent of the Majority Backstop Parties (if the Subscription and Backstop Purchase Agreement is then in effect) and/or the New Investors (if the Investment Agreement is then in effect), in each case, which consent shall not be unreasonably withheld, conditioned or delayed), in an aggregate principal amount of at least \$292 million, and substantially in the form attached hereto as Exhibit F.

New Harbinger Subordinated Notes means, in the event that the Harbinger Investment Effective Date Condition is satisfied, the New Senior Subordinated Notes due 2013 to be issued by Reorganized Bally under the New Harbinger Subordinated Notes Indenture in the aggregate principal amount of \$200 million. The New Harbinger Subordinated Notes shall be subordinate in priority of payment to the New Senior Second Lien Notes or any refinancing thereof.

New Harbinger Subordinated Notes Indenture means, in the event that the Harbinger Investment Effective Date Condition is satisfied, that certain Amended and Restated Indenture, to be dated as of the Effective Date, among Reorganized Bally, as issuer, and the New Harbinger Subordinated Notes Indenture Trustee, relating to the New Harbinger Subordinated Notes, substantially in the form of Exhibit O attached to this Plan.

New Harbinger Subordinated Notes Indenture Trustee means, in the event that the Harbinger Investment Effective Date Condition is satisfied, HSBC Bank USA, <u>National Association</u>, as the indenture trustee under the New Harbinger Subordinated Notes Indenture, and its successors and assigns.

New Investors means Harbinger Capital Partners Master Fund I, Ltd. and Harbinger Capital Partners Special Situations Fund L.P.

New Junior Subordinated Notes means, in the event that the Backstop Rights Offering Effective Date Condition is satisfied, the New Junior Subordinated Notes due 2013 to

be issued by Reorganized Bally under the New Junior Subordinated Notes Indenture in the aggregate principal amount not to exceed 21.7% of the Allowed Prepetition Senior Subordinated Notes Claims plus 21.7% of the Allowed Class 6-B-1 Claims. The New Junior Subordinated Notes shall be subordinate in priority of payment to the New Senior Second Lien Notes, the Rights Offering Senior Subordinated Notes and the New Subordinated Notes.

New Junior Subordinated Notes Indenture means, in the event that the Backstop Rights Offering Effective Date Condition is satisfied, that certain Amended and Restated Indenture, to be dated as of the Effective Date, among Reorganized Bally, as issuer, and the New Junior Subordinated Notes Indenture Trustee, relating to the New Junior Subordinated Notes, substantially in the form of Exhibit H attached to this Plan.

New Junior Subordinated Notes Indenture Trustee means, in the event that the Backstop Rights Offering Effective Date Condition is satisfied, HSBC Bank USA, National Association, as the indenture trustee under the New Junior Subordinated Notes Indenture, and its successors and assigns.

New Lenders means each of the financial institutions party to the New Credit Agreement and identified as "Lenders" therein, and their respective successors and assigns.

New Securities and Documents has the meaning given in Section 5.6 of this Plan.

New Senior Second Lien Notes means the New Senior Second Lien Notes due 2011 to be issued by Reorganized Bally under the New Senior Second Lien Notes Indenture in the aggregate principal amount of \$247,337,500. The New Senior Second Lien Notes shall be senior in priority of payment over the Rights Offering Senior Subordinated Notes (if any), the New Subordinated Notes (if any), the New Junior Subordinated Notes (if any) and the New Harbinger Subordinated Notes (if any). The New Senior Second Lien Notes will be secured by Liens on the same assets that secure the obligations under the New Credit Agreement, which Liens shall be subordinate in priority to the Liens securing the obligations under the New Credit Agreement.

New Senior Second Lien Notes Indenture means that certain Indenture, to be dated as of the Effective Date, among Reorganized Bally, as issuer, certain Affiliate Debtors, as guarantors, and the New Senior Second Lien Notes Indenture Trustee, relating to the New Senior Second Lien Notes, substantially in the forms of Exhibits G-1 (if the Harbinger Investment Effective Date Condition is satisfied) and G-2 (if the Backstop Rights Offering Condition is satisfied) attached to this Plan.

New Senior Second Lien Notes Indenture Trustee means U.S. Bank National Association, as the indenture trustee under the New Senior Second Lien Notes Indenture, and its successors and assigns.

New Stockholders Agreement means, in the event that the Backstop Rights Offering Effective Date Condition is satisfied, that certain stockholders agreement to be entered into on the Effective Date by and among Reorganized Bally and all holders of the New Common Stock, substantially in the form of Exhibit I attached to this Plan.

New Subordinated Notes means, in the event that the Backstop Rights Offering Effective Date Condition is satisfied, the New Senior Subordinated Notes due 2013 to be issued by Reorganized Bally under the New Subordinated Notes Indenture in the aggregate principal amount not to exceed 24.8% of the Allowed Prepetition Senior Subordinated Notes Claims plus 24.8% of the Allowed Class 6-B-1 Claims. The New Subordinated Notes shall be subordinate in priority of payment to the New Senior Second Lien Notes and the Rights Offering Senior Subordinated Notes, and senior in priority of payment to the New Junior Subordinated Notes.

New Subordinated Notes Indenture means, in the event that the Backstop Rights Offering Effective Date Condition is satisfied, that certain Amended and Restated Indenture, to be dated as of the Effective Date, among Reorganized Bally, as issuer, and the New Subordinated Notes Indenture Trustee, relating to the New Subordinated Notes, substantially in the form of Exhibit H attached to this Plan.

New Subordinated Notes Indenture Trustee means, in the event that the Backstop Rights Offering Effective Date Condition is satisfied, HSBC Bank USA, National Association, as the indenture trustee under the New Subordinated Notes Indenture, and its successors and assigns.

Non-Tax Priority Claim means a Claim, other than an Administrative Claim or Priority Tax Claim, that is entitled to priority in payment pursuant to section 507(a) of the Bankruptcy Code.

Old Affiliate Interests means, collectively, the shares of stock, whether common or preferred, general and limited partnership interests, or member or other ownership interests of the Affiliate Debtors, as applicable, issued and outstanding as of the Petition Date, and all options, warrants, calls, rights, puts, awards, commitments, or any other agreements of any character to acquire such stock or interests.

Old Common Stock means the common stock of Bally that is outstanding immediately prior to the Petition Date, including, without limitation, treasury stock and any other such interests that are authorized to be issued but have not been issued and the 510(b) Equity Claims.

Old Common Stock Cash Amount means, in the event that the Harbinger Investment Effective Date Condition is satisfied, \$16.5 million.

Old Common Stock Cash Portion means, in the event that the Harbinger Investment Effective Date Condition is satisfied, that portion of the Old Common Stock Cash Amount reserved solely for the Holders of Allowed Interests represented by Old Common Stock, which portion equals the Old Common Stock Cash Amount multiplied by the fraction the numerator of which equals the Old Common Stock Cash Amount and the denominator of which equals the sum of (x) the Old Common Stock Cash Amount plus (y) the aggregate face amount of all Allowed 510(b) Equity Claims; provided, however, that for purposes of making interim distributions hereunder to holders of Allowed Class 7

Claims, clause (y) of the denominator shall be the aggregate maximum amount of all 510(b) Equity Claims asserted by the holders thereof.

Old Common Stock Remaining Cash Portion means, in the event that the Harbinger Investment Effective Date Condition is satisfied, that portion of the Old Common Stock Cash Amount reserved solely for the Holders of Allowed 510(b) Equity Claims, which portion equals the Old Common Stock Cash Amount less the Old Common Stock Cash Portion.

Old Unexercised Equity Interests means all unexercised options, warrants, calls, rights, puts, awards, commitments, or any other agreements of any character to acquire Old Common Stock.

Other Released Parties means, collectively, the Released Parties other than the Debtor-Related Released Parties.

Other Secured Claim means a Secured Claim other than an Administrative Claim, Secured Tax Claim or Prepetition Lenders Claim.

Person means any natural person, corporation, general or limited partnership, limited liability company, firm, trust, association, government, governmental agency or other Entity, whether acting in an individual, fiduciary or other capacity.

Petition Date means the date on which the Debtors file their petitions for relief commencing the Chapter 11 Cases.

Plan means this first amended joint prepackaged chapter 11 plan of reorganization, including the Exhibits and Plan Schedules and all supplements, appendices, and schedules thereto, either in its present form or as the same may be altered, amended, modified or otherwise supplemented from time to time.

Plan-Related Claim means any and all claims against any Debtor or any other Released Party as to which such person or entity is exculpated pursuant to Section 1125(e) of the Bankruptcy Code, including, without limitation, to the maximum extent provided by Section 1125(e) of the Bankruptcy Code, any and all claims against any Debtor or other Released Party for violating any provisions of the Securities Act of 1933 or any State or local law requiring registration for offer or sale of a security or registration or licensing of an issuer of, underwriter of, or broker or dealer in, a security, in connection with any or all of the prepetition or postpetition solicitation of acceptances of the Plan, the prepetition or postpetition solicitation of subscriptions with respect to the Rights Offering, the Subscription and Backstop Purchase Agreement, the Investment Agreement (if the Harbinger Investment Effective Date Condition is satisfied), the First Restructuring Support Agreement, the Second Restructuring Support Agreement, and the acts taken under any of the foregoing.

Plan Schedule means a schedule annexed to either this Plan or as an appendix to the Disclosure Statement (as amended, modified or otherwise supplemented from time to time).

Prepetition Senior Noteholders means the Holders of the Prepetition Senior Notes Claims.

Prepetition Senior Notes means the 10.5% Senior Notes due 2011 issued by Bally prior to the Petition Date under the Prepetition Senior Notes Indenture.

Prepetition Senior Notes Claims means any and all Claims, obligations and liabilities arising from or related to the Prepetition Senior Notes and/or Prepetition Senior Notes Indenture, including, without limitation, the "Indenture Obligations" (as such term is defined in the Prepetition Senior Notes Indenture), which Claims shall be Allowed in the aggregate principal amount of \$235,000,000, plus interest and reasonable fees, costs and expenses that have accrued but remain unpaid as of the Petition Date pursuant to the Prepetition Senior Notes Indenture (which final aggregate amount shall be set forth in the Confirmation Order).

Prepetition Senior Notes Indenture means that certain Indenture, dated as of July 2, 2003, among Bally, as issuer, certain Affiliate Debtors, as guarantors, and the Prepetition Senior Notes Indenture Trustee, relating to the Prepetition Senior Notes (as amended, modified, or supplemented from time to time).

Prepetition Senior Notes Indenture Amendment Fee means that certain amendment fee in the aggregate amount of \$4,700,000, which fee shall be (i) in consideration of the amendments to the Prepetition Senior Notes Indenture as reflected in the New Senior Second Lien Notes Indenture; and (ii) paid to the Prepetition Senior Notes Indenture Trustee for the benefit of all Prepetition Senior Noteholders and shared on a Pro Rata basis with such holders.

Prepetition Senior Notes Indenture Trustee means U.S. Bank National Association, as the indenture trustee under the Prepetition Senior Notes Indenture, and its successors and assigns.

Prepetition Senior Notes Indenture Trustee Fees means the reasonable, fees and reasonable unpaid out-of-pocket costs and expenses incurred by the Prepetition Senior Notes Indenture Trustee through the Effective Date in accordance with the Prepetition Senior Note Indenture.

Prepetition Senior Subordinated Noteholders means the Holders of the Prepetition Senior Subordinated Notes Claims.

Prepetition Senior Subordinated Notes means, collectively, the 9.875% Senior Subordinated Notes due 2007, Series B, and the 9.875% Senior Subordinated Notes due 2007, Series D, issued by Bally prior to the Petition Date under the Prepetition Senior Subordinated Notes Indenture.

Prepetition Senior Subordinated Notes Claims means any and all Claims, obligations and liabilities arising from or related to the Prepetition Senior Subordinated Notes and/or Prepetition Senior Subordinated Notes Indenture, including, without limitation, the "Indenture Obligations" (as such term is defined in the Prepetition Senior Subordinated Notes

Indenture), which Claims shall be Allowed in the aggregate amount (including accrued and unpaid interest) of \$323,041,667.

Prepetition Senior Subordinated Notes Indenture means, collectively, that certain Indenture, dated as of October 7, 1997, and that certain Indenture, dated as of December 16, 1998, in each case by and between Bally, as issuer, and the Prepetition Senior Subordinated Notes Indenture Trustee, relating to the Prepetition Senior Subordinated Notes (as amended, modified, or supplemented from time to time).

Prepetition Senior Subordinated Notes Indenture Trustee means HSBC Bank USA, National Association, as the indenture trustee under the Prepetition Senior Subordinated Notes Indenture, and its successors and assigns.

Prepetition Senior Subordinated Notes Indenture Trustee Fees means the reasonable, fees and reasonable unpaid out-of-pocket costs and expenses incurred by the Prepetition Senior Subordinated Notes Indenture Trustee through the Effective Date in accordance with the Prepetition Senior Subordinated Note Indenture.

Priority Tax Claim means a Claim of a governmental unit of the kind specified in sections 502(i) and 507(a)(8) of the Bankruptcy Code, including a Secured Tax Claim.

Professional means (a) any professional employed in the Chapter 11 Cases pursuant to section 327 or 1103 of the Bankruptcy Code or otherwise and (b) any professional or other Entity seeking compensation or reimbursement of expenses in connection with the Chapter 11 Cases pursuant to section 503(b)(4) of the Bankruptcy Code.

Professional Fees means an Administrative Claim of a Professional for compensation for services rendered or reimbursement of costs, expenses or other charges incurred after the Petition Date and prior to and including the Effective Date (including expenses of the members of the Creditors' Committee incurred as members of the Creditors' Committee in discharge of their duties as such).

Professional Fees Bar Date means the Business Day that is sixty (60) days after the Effective Date or such other date as approved by order of the Bankruptcy Court.

Pro Rata means with respect to a distribution regarding a particular Class (or several Classes taken as a whole), the proportion that (a) the Allowed amount of a Claim or Interest in a particular Class (or several Classes taken as a whole) bears to (b) the aggregate Allowed amount of all Claims or Interests in such Class (or several Classes taken as a whole), unless this Plan provides otherwise.

Quarterly Distribution Date means the last Business Day of the month following the end of each calendar quarter after the Effective Date; provided, however, that if the Effective Date is within thirty (30) days of the end of a calendar quarter, then the first Quarterly Distribution Date will be the last Business Day of the month following the end of the first calendar quarter after the calendar quarter in which the Effective Date falls.

accordance with section 1123(a)(1) of the Bankruptcy Code, Administrative Claims and Priority Tax Claims have not been classified as described below.

This Plan constitutes a single plan of reorganization for all Debtors for all purposes, including, without limitation, for voting, confirmation, and distribution purposes. A Claim or Interest is placed in a particular Class only to the extent that the Claim or Interest falls within the description of that Class and is classified in other Classes to the extent that any portion of the Claim or Interest falls within the description of such other Classes. A Claim or Interest is also placed in a particular Class for the purpose of receiving distributions pursuant to this Plan only to the extent that such Claim or Interest is an Allowed Claim or Interest in that Class and such Claim or Interest has not been paid, released, or otherwise settled prior to the Effective Date. As described more fully in Section 5.1 below, this Plan contemplates and is predicated upon entry of an order substantively consolidating the Debtors for the limited purposes of voting, confirmation and distribution with respect to Allowed Class 5 Claims and Allowed Class 6-CB-2 Claims only.

Summary of Classification of Claims and Interests			
Class	Claim	<u>Status</u>	Voting Rights
1.	Non-Tax Priority Claims	Unimpaired	Deemed to Accept
2.	Other Secured Claims	Unimpaired	Deemed to Accept
3.	Unimpaired Unsecured Claims	Unimpaired	Deemed to Accept
4.	Prepetition Lenders Claims	Unimpaired	Deemed to Accept
5.	Prepetition Senior Notes Claims	Impaired	Entitled to Vote
6-A.	Prepetition Senior Subordinated Notes Claims	Impaired	Entitled to Vote
6-B-1.	Rejection Claims Against Only Bally	Impaired	Deemed to Reject
6-B-2.	Rejection Claims Against Any Affiliate Debtor	Impaired	Deemed to Reject
7.	Old Common Stock	Impaired	Deemed to Reject
8.	Old Unexercised Equity Interests	Impaired	Deemed to Reject
9.	Old Equity Interests in	Unimpaired	Deemed to Accept

pursuant to section 1123(a)(6) of the Bankruptcy Code, a provision prohibiting the issuance of non-voting equity securities, but only to the extent required by section 1123(a)(6) of the Bankruptcy Code; (ii) authorize the issuance of New Common Stock in an amount not less than the amount necessary to permit the distributions thereof required or contemplated by this Plan; (iii) to the extent necessary or appropriate, include restrictions on the Transfer of New Common Stock; and (iv) to the extent necessary or appropriate, include such provisions as may be needed to effectuate and consummate this Plan and the transactions contemplated herein. After the Effective Date, the Reorganized Debtors may amend and restate their respective certificates or articles of incorporation and by-laws, and other applicable organizational documents, as permitted by applicable law.

- (b) Directors and Officers of the Reorganized Debtors.
- In the event that the Harbinger Investment Effective Date Condition is (i) satisfied, subject to any requirement of Bankruptcy Court approval pursuant to section 1129(a)(5) of the Bankruptcy Code, as of the Effective Date, the initial officers of Reorganized Bally shall be the officers of Bally existing immediately prior to the Effective Date. On the Effective Date, the board of directors of Reorganized Bally shall be selected by the New Investors and identified via a Plan supplement filed with the Bankruptcy Court at least 10 days prior to the Confirmation Hearing. The boards of directors and initial officers of the Reorganized Affiliate Debtors on the Effective Date shall be comprised of the same individuals who currently serve in such capacities immediately prior to the Effective Date (other than those directors of the Affiliate Debtors listed on the resignation plan supplement to be filed with the Bankruptcy Court prior to the Confirmation Hearing, which listed directors will be deemed to have resigned from the applicable Affiliate Debtors on and as of the Effective **Date**). Pursuant to section 1129(a)(5) of the Bankruptcy Code, the Debtors will disclose, at or prior to the Confirmation Hearing, the identity and affiliations of any Person proposed to serve on the initial board of directors of Reorganized Bally, and, to the extent such Person is an insider other than by virtue of being a director, the nature of any compensation for such Person. The length of the initial term of each director shall be one year. Each such director and officer shall serve from and after the Effective Date pursuant to applicable law and the terms of the Amended Certificate of Incorporation, the other constituent and organizational documents of the Reorganized Debtors. The existing board of directors of Bally will be deemed to have resigned on and as of the Effective Date.

- In the event that the Backstop Rights Offering Effective Date Condition is satisfied, subject to any requirement of Bankruptcy Court approval pursuant to section 1129(a)(5) of the Bankruptcy Code, as of the Effective Date, the initial officers of Reorganized Bally shall be the officers of Bally existing immediately prior to the Effective Date. On the Effective Date, the board of directors of Reorganized Bally shall have not less than three and not more than nine members selected by the holders of a majority of the Allowed Prepetition Senior Subordinated Notes Claims and identified via a Plan supplement filed with the Bankruptcy Court at least 10 days prior to the Confirmation Hearing. The boards of directors and initial officers of the Reorganized Affiliate Debtors on the Effective Date shall be comprised of the same individuals who currently serve in such capacities immediately prior to the Effective Date (other than those directors of the Affiliate Debtors listed on the resignation plan supplement to be filed with the Bankruptcy Court prior to the Confirmation Hearing, which listed directors will be deemed to have resigned from the applicable Affiliate Debtors on and as of the Effective Date). Pursuant to section 1129(a)(5) of the Bankruptcy Code, the Debtors will disclose, at or prior to the Confirmation Hearing, the identity and affiliations of any Person proposed to serve on the initial board of directors of Reorganized Bally, and, to the extent such Person is an insider other than by virtue of being a director, the nature of any compensation for such Person. The length of the initial term of each director shall be one year. Each such director and officer shall serve from and after the Effective Date pursuant to applicable law and the terms of the Amended Certificate of Incorporation, the other constituent and organizational documents of the Reorganized Debtors. The existing board of directors of Bally will be deemed to have resigned on and as of the Effective Date.
- Corporate Action. On the Effective Date, the adoption of the Amended Certificate of Incorporation and By-Laws and similar constituent and organizational documents, and the selection of directors and officers for, each of the Reorganized Debtors, and all other actions contemplated by or described in this Plan with respect thereto, shall be authorized and approved and be binding and in full force and effect in all respects (subject to the provisions of this Plan and the Confirmation Order), in each case without further notice to or order of the Bankruptcy Court, act or action under applicable law, regulation, order, or rule (other than filing such organizational documents with the applicable governmental unit as required by applicable law) or the vote, consent, authorization or approval of any Person. All matters provided for in this Plan involving the legal or corporate structure of the Debtors or the Reorganized Debtors, and any legal or corporate action required by the Debtors or the Reorganized Debtors in connection with this Plan, shall be deemed to have occurred and shall be in full force and effect in all respects, in each case without further notice to or order of the Bankruptcy Court, act or action under applicable law, regulation, order, or rule or any requirement of further action, vote or other approval or authorization by the security holders, officers or directors of the Debtors or the Reorganized Debtors or by any other Person. On the Effective Date, the appropriate officers of the Debtors and Reorganized Debtors and members of their respective boards of directors are authorized to issue, execute, and deliver, and consummate the transactions contemplated by, the contracts, agreements, documents, guarantees, pledges, consents, securities, certificates, resolutions and instruments contemplated by or described in this Plan in the name of and on behalf of the Debtors and Reorganized Debtors, in each case without further notice to or order of the Bankruptcy Court, act or action under applicable law, regulation, order, or rule or any requirement of further action, vote or other approval or authorization by any Person.

5.5 Cancellation of Notes, Instruments, Debentures, Preferred Stock and Common Stock

On the Effective Date, except as otherwise provided in this Plan or the Confirmation Order, (i) the Prepetition Senior Notes, the Prepetition Senior Subordinated Notes, the Old Common Stock, the Old Unexercised Equity Interests, and any other notes, bonds (with the exception of any surety bonds outstanding), indentures, or other instruments or documents evidencing or creating any indebtedness or obligations of a Debtor that are Impaired under this Plan shall be cancelled and extinguished, and (ii) the obligations of the Debtors under any agreements, documents, indentures, or certificates of designation governing the Prepetition Senior Notes, Prepetition Senior Subordinated Notes, Old Common Stock, Old Unexercised Equity Interests, and any other notes, bonds, indentures, or other instruments or documents evidencing or creating any indebtedness or obligations of a Debtor that are Impaired under this Plan shall be, and are hereby, discharged, in each case without further notice to or order of the Bankruptcy Court, act or action under applicable law, regulation, order, or rule or any requirement of further action, vote or other approval or authorization by the security holders, officers or directors of the Debtors or the Reorganized Debtors or by any other Person. Notwithstanding the foregoing, the Prepetition Senior Notes Indenture and the Prepetition Senior Subordinated Notes Indenture shall continue in effect solely for the purposes of: (i) allowing Prepetition Senior Noteholders and Prepetition Senior Subordinated Noteholders to receive distributions under this Plan; and (ii) allowing and preserving the rights of the Prepetition Senior Notes Indenture Trustee and the Prepetition Senior Subordinated Notes Indenture Trustee to make distributions in satisfaction of Allowed Prepetition Senior Notes Claims and Allowed Prepetition Senior Subordinated Notes Claims, but in all cases subject to the terms and conditions of the Prepetition Senior Notes Indenture and Prepetition Senior Subordinated Notes Indenture, including, but not limited to, any charging lien or priority payment rights of the **Prepetition Senior Notes Indenture Trustee and the Prepetition Senior Subordinated Notes Indenture Trustee**. The Prepetition Senior Notes Indenture Trustee and the Prepetition Senior Subordinated Notes Indenture Trustee shall be entitled to reasonable compensation to the extent that they perform services for the Prepetition Senior Noteholders and the Prepetition Senior Subordinated Noteholders, respectively, (i) before the Effective Date, in Cash on the Effective Date and (ii) after the Effective Date, in Cash and in accordance with the terms of the Prepetition Senior Notes Indenture and the Prepetition Senior Subordinated Notes Indenture, in both cases without further notice to or order of the Bankruptcy Court. As of the Effective Date, the Prepetition Senior Notes and the Prepetition Senior Subordinated Notes shall be surrendered to the Prepetition Senior Notes Indenture Trustee and the Prepetition Senior Subordinated Notes Indenture Trustee, respectively, in accordance with the terms of the Prepetition Senior Notes Indenture and the Prepetition Senior Subordinated Notes Indenture. All surrendered and canceled Prepetition Senior Notes and Prepetition Senior Subordinated Notes held by the Prepetition Senior Notes Indenture Trustee and the Prepetition Senior Subordinated Notes Indenture Trustee shall be disposed of in accordance with the applicable terms and conditions of the Prepetition Senior Notes Indenture and the Prepetition Senior Subordinated Notes Indenture.

5.6 Issuance of New Securities and Related Documentation

On, or as soon as reasonably practicable after, the Effective Date, Reorganized Bally is authorized to and shall issue the New Common Stock, the New Senior Second Lien

satisfaction and payment of any tax obligations imposed by any governmental unit, including income, withholding, and other tax obligations, on account of such distribution, and (ii) no distribution shall be made to or on behalf of such Holder pursuant to this Plan unless and until such Holder has made arrangements satisfactory to the Reorganized Debtors for the payment and satisfaction of such tax obligations. Any Cash, New Common Stock, New Senior Second Lien Notes, New Subordinated Notes (if any), New Harbinger Subordinated Notes (if any), New Junior Subordinated Notes (if any), Rights Offering Senior Subordinated Notes (if any), other New Securities and Documents and/or other consideration or property to be distributed pursuant to this Plan shall, pending the implementation of such arrangements, be treated as an undeliverable distribution pursuant to Section 6.4 of this Plan.

6.9 **Setoffs**

The Reorganized Debtors may, pursuant to section 553 of the Bankruptcy Code or applicable non-bankruptcy laws, but shall not be required to, set off against any Claim, the payments or other distributions to be made pursuant to this Plan in respect of such Claim, or claims of any nature whatsoever that the Debtors or the Reorganized Debtors may have against the Holder of such Claim; <u>provided</u>, <u>however</u>, that neither the failure to do so nor the allowance of any Claim hereunder shall constitute a waiver or release by the Reorganized Debtors of any such claim that the Debtors or the Reorganized Debtors may have against such Holder.

6.10 Fractional Shares

No fractional shares of New Common Stock shall be distributed by the Debtors or Reorganized Debtors. Where a fractional share would otherwise be called for, the actual issuance shall reflect a rounding up (in the case of .50 or more than .50) of such fraction to the nearest whole share of New Common Stock or a rounding down of such fraction (in the case of less than .50).

6.11 Surrender of Canceled Notes and Canceled Instruments of Securities

- (a) Generally. As a condition precedent to receiving any distribution pursuant to this Plan on account of an Allowed Claim or Allowed Interest evidenced by the instruments, securities, notes, or other documentation canceled pursuant to Section 5.5 of this Plan, the Holder of such Claim or Interest shall tender the applicable instruments, securities, notes or other documentation evidencing such Claim or Interest to the Reorganized Debtors or other applicable Disbursing Agent unless waived in writing by the Debtors or the Reorganized Debtors, as applicable.
- (b) Prepetition Notes. Each Holder of a Prepetition Senior Subordinated Notes Claim or a Prepetition Senior Notes Claim shall tender its respective Prepetition Senior Subordinated Notes or Prepetition Senior Notes relating to such Claim to the Reorganized Debtors or Disbursing Agent in accordance with written instructions to be provided to such Holders by the Reorganized Debtors or the Prepetition Senior Subordinated Notes Indenture Trustee or the Prepetition Senior Notes Indenture Trustee as promptly as practicable following the Effective Date. Such instructions shall specify that delivery of such Prepetition Senior Subordinated Notes or Prepetition Senior Notes will be effected, and risk of loss and title thereto

will pass, only upon the proper delivery of such Prepetition Senior Subordinated Notes or Prepetition Senior Notes with a letter of transmittal in accordance with such instructions. All surrendered Prepetition Senior Subordinated Notes and Prepetition Senior Notes shall be marked as canceled.

- (c) Old Common Stock. To the extent applicable, each Holder of Old Common Stock shall tender its Old Common Stock to the Reorganized Debtors or their designated agent in accordance with written instructions to be provided to such Holders by the Reorganized Debtors as promptly as practicable following the Effective Date. Such instructions shall specify that delivery of such Old Common Stock will be effected, and risk of loss and title thereto will pass, only upon the proper delivery of such Old Common Stock with a letter of transmittal in accordance with such instructions. All surrendered Old Common Stock shall be marked as canceled.
- (d) Failure to Surrender Security Instruments. Any Holder of a Prepetition Senior Notes Claim, Prepetition Senior Subordinated Notes Claim and/or Old Common Stock that fails to surrender or is deemed to have failed to surrender the applicable note or security required to be tendered hereunder within one (1) year after the Effective Date shall have its Claim and Interest and its distribution pursuant to this Plan on account of such Claim or Interest discharged and shall be forever barred from asserting any such Claim or Interest against the Reorganized Debtors or their respective property. In such cases, any Cash, New Common Stock, New Senior Second Lien Notes, New Subordinated Notes, New Junior Subordinated Notes, Rights Offering Senior Subordinated Notes, New Harbinger Subordinated Notes, other New Securities and Documents and/or other consideration or property held for distribution on account of such Claim or Interest shall be disposed of pursuant to Section 6.4(iii) of this Plan.

6.12 Lost, Stolen, Mutilated, or Destroyed Securities

In addition to any requirements under any applicable agreement and applicable law, any Holder of a Claim or Interest evidenced by a security or note that has been lost, stolen, mutilated, or destroyed shall, in lieu of surrendering such security or note to the extent required by this Plan, deliver to the Reorganized Debtors and other applicable Disbursing Agent: (x) evidence reasonably satisfactory to the Reorganized Debtors and other applicable Disbursing Agent of such loss, theft, mutilation, or destruction; and (y) such security or indemnity as may be required by the Reorganized Debtors and other applicable Disbursing Agent to hold such party harmless from any damages, liabilities, or costs incurred in treating such individual as a Holder of an Allowed Claim or Allowed Interest. Upon compliance with this Section 6.12 as determined by the Debtors or Reorganized Debtors by a Holder of a Claim or Interest evidenced by a security or note, such Holder shall, for all purposes under this Plan, be deemed to have surrendered such security or note to the Reorganized Debtors and other applicable Disbursing Agent.

6.13 Distributions from Old Common Stock Cash Amount

In the event that the Harbinger Investment Effective Date Condition is satisfied, for purposes of determining distributions to be made from the Old Common Stock Cash Amount, each Holder of an Allowed Interest represented by Old Common

Stock will receive its Pro Rata share of the Old Common Stock Cash Portion, and each Holder of an Allowed 510(b) Equity Claim will receive its Pro Rata share of the Old Common Stock Remaining Cash Portion.

ARTICLE SEVEN TREATMENT OF EXECUTORY CONTRACTS AND UNEXPIRED LEASES

7.1 Assumption of Executory Contracts and Unexpired Leases

On the Effective Date, all executory contracts and unexpired leases of the Debtors will be deemed assumed in accordance with, and subject to, the provisions and requirements of sections 365 and 1123 of the Bankruptcy Code, except those executory contracts (including, without limitation, employment agreements) and unexpired leases that (i) have been rejected by order of the Bankruptcy Court, (ii) are the subject of a motion to reject pending on the Effective Date, (iii) are identified on Exhibit E-1 or Exhibit E-2 hereto (which Exhibits E-1 and E-2 may be amended by the Debtors (with the consent of the Majority Backstop Parties, if the Subscription and Backstop Purchase Agreement is then in effect, and/or the New Investors, if the Investment Agreement is then in effect) to add or remove executory contracts and unexpired leases by filing with the Bankruptcy Court amended Exhibits E-1 and E-2 and serving them on the affected contract parties at any time on or prior to five (5) days prior to the deadline set by the Bankruptcy Court for Filing objections to confirmation of this Plan), or (iv) are rejected pursuant to the terms of this Plan. Without amending or altering any prior order of the Bankruptcy Court approving the assumption or rejection of any executory contract or unexpired lease, entry of the Confirmation Order by the Bankruptcy Court shall constitute approval of such assumptions and rejections pursuant to sections 365(a) and 1123 of the Bankruptcy Code. To the extent any provision in any executory contract or unexpired lease assumed pursuant to this Plan (including, without limitation, any "change of control" provision) restricts or prevents, or purports to restrict or prevent, or is breached or deemed breached by, the applicable Reorganized Debtor's assumption of such executory contract or unexpired lease, then such provision shall be deemed modified such that the transactions contemplated by this Plan shall not entitle the nondebtor party thereto to terminate such executory contract or unexpired lease or to exercise any other default-related rights with respect thereto. Each executory contract and unexpired lease assumed pursuant to this Article VII shall revest in and be fully enforceable by the respective Reorganized Debtor in accordance with its terms, except as modified by the provisions of this Plan, any order of the Bankruptcy Court authorizing and providing for its assumption, or applicable law.

7.2 Claims Based on Rejection of Executory Contracts or Unexpired Leases

All proofs of claim with respect to Claims arising from or in connection with the rejection of executory contracts or unexpired leases, if any, must be filed with the Bankruptcy Court within thirty (30) days after the date of entry of an order of the Bankruptcy Court approving such rejection or, if listed in Exhibits E-1 or E-2, thirty (30) days after the date of entry of the Confirmation Order. Any Claims arising from or in connection with the rejection of an executory contract or unexpired lease not filed within such time will be forever barred from

Claims Objection Deadline (unless extended by an order of the Bankruptcy Court), the Debtors or the Reorganized Debtors, as the case may be, shall file objections to such Claims with the Bankruptcy Court and serve such objections upon the Holders of such Claims to which objections are made. Nothing contained herein, however, shall limit the Reorganized Debtors' right to object to Claims, if any, filed or amended after the Claims Objection Deadline. The Debtors and the Reorganized Debtors shall be authorized to, and shall, resolve all Rejection Claims and 510(b) Equity Claims by withdrawing or settling such objections thereto, or by litigating to Final Order in the Bankruptcy Court, the validity, nature, and/or amount thereof.

All Other Claims and Interests. Except as otherwise provided in this Plan, (b) holders of Claims and Interests other than Holders of Rejection Claims and 510(b) Equity Claims shall not be required to file a proof of claim or proof of interest, and no such parties should file a proof of claim or proof of interest. Unless disputed by a holder of a Claim or Interest or by the Debtors, the amount set forth in the Schedules (if the Debtors are required to file Schedules) or in the books and records of the Debtors (if the Debtors are not required to file Schedules) shall constitute the amount of the Allowed Claim or Allowed Interest of such holder. If any such holder of a Claim or Interest disagrees with the Debtors' determination with respect to the Allowed amount of such Holder's Claim or Interest, such Holder must so advise the Debtors in writing (at any time whether prior to or after the Effective Date), in which event the Claim or Interest will be a Disputed Claim or Disputed Interest. The Debtors intend to attempt to resolve any such disputes consensually or through judicial means outside the Bankruptcy Court (and no further Bankruptcy Court order shall be required in connection with such resolutions). Nevertheless, no later than ninety (90) days after the Effective Date (or such later date as approved by this Court for cause shown after notice and hearing), the Debtors may, in their discretion, File with the Bankruptcy Court (or any other court of competent jurisdiction) an objection to the allowance of any Claim or Interest or any other appropriate motion or adversary proceeding with respect thereto. All such objections will be litigated to Final Order; provided, however, that the Debtors may compromise and settle, withdraw or resolve by any other method approved by the Bankruptcy Court, any objection to Claims and Interests without further order of the Bankruptcy Court.

8.2 No Distributions Pending Allowance

Notwithstanding any other provision of this Plan to the contrary, no payments or distributions of any kind or nature shall be made with respect to all or any portion of a Disputed Claim or Disputed Interest unless and until all objections to such Disputed Claim or Disputed Interest have been settled or withdrawn or have been determined by Final Order, and the Disputed Claim has become an Allowed Claim or the Disputed Interest has become an Allowed Interest.

8.3 Distributions on Account of Disputed Claims Once They Are Allowed and Additional Distributions on Account of Previously Allowed Claims

On each Quarterly Distribution Date (or such earlier date as determined by the Reorganized Debtors in their sole discretion but subject to <u>Section 8.2</u> of this Plan), the Reorganized Debtors will make distributions (a) on account of any Disputed Claim that has become an Allowed Claim during the preceding calendar quarter or any Disputed Interest that

ARTICLE TEN EFFECT OF PLAN CONFIRMATION

10.1 Binding Effect; Plan Binds All Holders of Claims and Interests

On the Effective Date, and effective as of the Effective Date, this Plan shall, and shall be deemed to, be binding upon and inure to the benefit of the Debtors, all present and former Holders of Claims against and Interests in any Debtor, and their respective successors and assigns, including, but not limited to, the Reorganized Debtors, regardless of whether any such Holder failed to vote to accept or reject this Plan or affirmatively voted to reject this Plan.

10.2 Releases and Related Injunctions

- (a) Releases by the Debtors. Effective Subject to the provisions of Section 10.9 hereof, effective as of the Effective Date, for good and valuable consideration, the adequacy of which is hereby confirmed, the Debtors and Reorganized Debtors, in their individual capacities and as debtors in possession, will be deemed to forever release, waive, and discharge all claims, obligations, suits, judgments, damages, demands, debts, rights, Causes of Action, and liabilities (other than the rights of the Debtors or Reorganized Debtors to enforce this Plan and the contracts, instruments, releases, indentures, and other agreements or documents delivered under or in connection with this Plan or assumed pursuant to this Plan), whether liquidated or unliquidated, fixed or contingent, matured, or unmatured, known or unknown, foreseen, or unforeseen, then existing or thereafter arising, in law, equity, or otherwise that are based in whole or part on any act, omission, transaction, event, or other occurrence taking place on or prior to the Effective Date in any way relating to the Debtors, the Reorganized Debtors, the Chapter 11 Cases, the Disclosure Statement, the First Restructuring Support Agreement, the Second Restructuring Support Agreements (if the Harbinger Investment Effective Date Condition is satisfied or as otherwise provided in the Investment Agreement), the Investment Agreement (if the Harbinger Investment Effective Date Condition is satisfied or as otherwise provided in the Investment Agreement) or this Plan (or the solicitation of votes on this Plan), and that could have been asserted by or on behalf of the Debtors, their Estates or the Reorganized Debtors against any of the Released Parties, in each case without further notice to or order of the Bankruptcy Court, act or action under applicable law, regulation, order, or rule or the vote, consent, authorization or approval of any Person.
- Debtor-Related Released Parties. Subject to the provisions of Section 10.9 hereof, effective as of the Effective Date, for good and valuable consideration and in consideration for the obligations of the Debtors and the Reorganized Debtors under this Plan and the property, securities, contracts, instruments, releases, and other agreements or documents to be delivered in connection with this Plan, to the fullest extent permissible under applicable law, the Holders of Claims or Interests, and each of their respective Related Persons, will be deemed to completely and forever release, waive, void, extinguish, and discharge the Debtors and Reorganized Debtors, and each of their respective Debtor-Related Persons Released Parties, from any and all claims, demands, debts, rights, Causes of Action, or liabilities (other than the right to enforce the Debtors' or the Reorganized Debtors' obligations under this Plan, and the contracts, instruments, releases, agreements, and documents delivered under this Plan), whether liquidated

or unliquidated, fixed or contingent, matured or unmatured, known or unknown, foreseen or unforeseen, then existing or thereafter arising, in law, equity, or otherwise, that are based in whole or in part on any act or omission, transaction, event, or other occurrence taking place on or prior to the Effective Date in any way relating to the Debtors, the Reorganized Debtors, the Chapter 11 Cases, the Disclosure Statement, the First Restructuring Support Agreement, the Second Restructuring Support Agreements (if the Harbinger Investment Effective Date Condition is satisfied or as otherwise provided in the Investment Agreement), the Investment Agreement (if the Harbinger Investment Effective Date Condition is satisfied or as otherwise provided in the Investment Agreement) or this Plan (or the solicitation of votes on this Plan) and that could have been asserted by or on behalf of (whether directly or derivatively), or against, the Debtors, their Estates or the Reorganized Debtors or against any of their respective Related Persons, any or all of the Debtor-Related Released Parties, in each case without further notice to or order of the Bankruptcy Court, act or action under applicable law, regulation, order, or rule or the vote, consent, authorization or approval of any Person.

Releases by Prepetition Senior Noteholders. Effective Holders of Claims and Interests of Claims Against Other Released Parties. In addition to, and without limiting, the release provisions of Section 10.2(b) in favor of the Debtor-Related Released Parties, effective as of the Effective Date, for good and valuable consideration and in consideration for the obligations of the Debtors and the Reorganized Debtors the Reorganized Debtors under this Plan and the property, securities, contracts, instruments, releases, and other agreements or documents to be delivered in connection with this Plan, to the fullest extent permissible under applicable law, the Prepetition Senior Notes Indenture Trustee and each Prepetition Senior Noteholder, and each of their respective Related Persons, Holders of Claims or Interests, and each of their respective Related Persons, will be deemed to completely and forever release, waive, void, extinguish, and discharge each of the Released Parties Other Released Parties. from any and all claims, demands, debts, rights, Causes of Action, or liabilities (other than the right to enforce the Debtors' or the Reorganized Debtors' obligations under this Plan, and the contracts, instruments, releases, agreements, and documents delivered under this Plan), whether liquidated or unliquidated, fixed or contingent, matured or unmatured, known or unknown, foreseen or unforeseen, then existing or thereafter arising, in law, equity, or otherwise, that are based in whole or in part on any act or omission, transaction, event, or other occurrence taking place on or prior to the Effective Date in any way relating to the Debtors, the Reorganized Debtors, the Chapter 11 Cases, the Disclosure Statement, the First Restructuring Support Agreement, the Second Restructuring Support Agreements (if the Harbinger Investment Effective Date Condition is satisfied or as otherwise provided in the Investment Agreement), the Investment Agreement (if the Harbinger Investment Effective Date Condition is satisfied or as otherwise provided in the Investment Agreement) or this Plan (or the solicitation of votes on this Plan) and that could have been asserted by or on behalf (whether directly or derivatively) of the Debtors, their Estates or the Reorganized Debtors or against any of the Other Released Parties, in each case without further notice to or order of the Bankruptcy Court, act or action under applicable law, regulation, order, or rule or the vote, consent, authorization or approval of any Person.

(d) Releases by Prepetition Senior Subordinated Noteholders. Effective as of the Effective Date, for good and valuable consideration and in consideration for the obligations of the Debtors and the Reorganized Debtors under this Plan and the property, securities,

contracts, instruments, releases, and other agreements or documents to be delivered in connection with this Plan, to the fullest extent permissible under applicable law, the Prepetition-Senior Subordinated Notes Indenture Trustee and each Prepetition Senior Subordinated Noteholder, and each of their respective Related Persons, will be deemed to completely and forever release, waive, void, extinguish, and discharge each of the Released Parties from any and all claims, demands, debts, rights, Causes of Action, or liabilities (other than the right to enforcethe Debtors' or the Reorganized Debtors' obligations under this Plan, and the contracts, instruments, releases, agreements, and documents delivered under this Plan), whether liquidated or unliquidated, fixed or contingent, matured or unmatured, known or unknown, foreseen or unforeseen, then existing or thereafter arising, in law, equity, or otherwise, that are based inwhole or in part on any act or omission, transaction, event, or other occurrence taking place on or prior to the Effective Date in any way relating to the Debtors, the Reorganized Debtors, the Chapter 11 Cases, the Disclosure Statement, the First Restructuring Support Agreement, the Second Restructuring Support Agreements (if the Harbinger Investment Effective Date-Condition is satisfied or as otherwise provided in the Investment Agreement), the Investment Agreement (if the Harbinger Investment Effective Date Condition is satisfied or as otherwise provided in the Investment Agreement) or this Plan (or the solicitation of votes on this Plan) and that could have been asserted by or on behalf (whether directly or derivatively) of the Debtors, their Estates or the Reorganized Debtors or against any of the Released Parties, in each case without further notice to or order of the Bankruptcy Court, act or action under applicable law, regulation, order, or rule or the vote, consent, authorization or approval of any Person.

(d) (e) Injunction Related to Releases. The Confirmation Order will permanently enjoin the commencement or prosecution by any Person or Entity, whether directly, derivatively or otherwise, of any claims, obligations, suits, judgments, damages, demands, debts, rights, Causes of Action, or liabilities released pursuant to this Plan, including but not limited to the claims, obligations, suits, judgments, damages, demands, debts, rights, Causes of Action, or liabilities released in this Section 10.2.

10.3 **Discharge of Claims**

To the fullest extent provided under section 1141(d)(1)(A) and other applicable provisions of the Bankruptcy Code, except as otherwise expressly provided by this Plan or the Confirmation Order, all consideration distributed under this Plan shall be in exchange for, and in complete satisfaction, settlement, discharge, and release of, all Claims of any kind or nature whatsoever against the Debtors or any of their assets or properties, and regardless of whether any property shall have been distributed or retained pursuant to this Plan on account of such Claims. Except as otherwise expressly provided by this Plan or the Confirmation Order, upon the Effective Date, the Debtors, and each of them, shall be deemed discharged and released under and to the fullest extent provided under section 1141(d)(1)(A) of the Bankruptcy Code from any and all Claims of any kind or nature whatsoever, including, but not limited to, demands and liabilities that arose before the Confirmation Date, and all debts of the kind specified in section 502(g), 502(h), or 502(i) of the Bankruptcy Code.

10.4 Preservation of Rights of Action; Settlement of Litigation Claims

- Preservation of Rights of Action. Except as otherwise provided in this Plan, the Confirmation Order, or in any document, instrument, release, or other agreement entered into in connection with this Plan or approved by order of the Bankruptcy Court, in accordance with section 1123(b) of the Bankruptcy Code, the Debtors and their Estates shall retain the Litigation Claims. The Reorganized Debtors, as the successors in interest to the Debtors and the Estates, may, and shall have the exclusive right to, enforce, sue on, settle, compromise, transfer or assign (or decline to do any of the foregoing) any or all of the Litigation Claims, including, without limitation, any and all derivative actions pending or otherwise existing against the Debtors as of the Effective Date. Notwithstanding the foregoing, the Debtors and the Reorganized Debtors shall not file, commence, or pursue any claim, right, or cause of action under section 547 of the Bankruptcy Code; provided, however, that, notwithstanding any statute of limitations, the Debtors and Reorganized Debtors shall have the right to assert or raise such Causes of Action (a) as defenses or counterclaims (up to the amount asserted in the Claims against the Debtors) with respect to any Disputed Claim, and (b) in connection with the Claims objection process with respect to a Claim that is not an Allowed Claim, in which case such Causes of Action can be raised as an objection to such Claim and not as defenses or counterclaims.
- (b) Settlement of Litigation Claims. At any time after the Confirmation Date and before the Effective Date, notwithstanding anything in this Plan to the contrary, the Debtors may settle any or all of the Litigation Claims with the approval of the Bankruptcy Court pursuant to Bankruptcy Rule 9019. After the Effective Date, the Reorganized Debtors may, and shall have the exclusive right to, compromise and settle any Claims against them and claims they may have against other Person or Entity, including, without limitation, the Litigation Claims, without notice to or approval from the Bankruptcy Court, including, without limitation, any and all derivative actions pending or otherwise existing against the Debtors as of the Effective Date.

10.5 Exculpation and Limitation of Liability

None Subject to the provisions of Section 10.9 hereof, none of the Released Parties shall have or incur any liability to, or be subject to any right of action by, any Holder of a Claim or an Interest, or any other party in interest, or any of their respective agents, employees, representatives, financial advisors, attorneys, or agents acting in such capacity, or affiliates, or any of their successors or assigns, or any other Released Party, for any act or omission in connection with, relating to, or arising out of, the Chapter 11 Cases, formulating, negotiating, or implementing this Plan, the First Restructuring Support Agreement, the Second Restructuring Support Agreements (if the Harbinger Investment Effective Date Condition is satisfied or as otherwise provided in the Investment Agreement), the Investment Agreement (if the Harbinger Investment Effective Date Condition is satisfied or as otherwise provided in the Investment Agreement), the prepetition or postpetition solicitation of acceptances of this Plan, the prepetition or postpetition solicitation of subscriptions with respect to the Rights Offering, the Subscription and Backstop Purchase Agreement and the acts taken thereunder, the Investment Agreement (if the Harbinger Investment Effective Date Condition is satisfied or as otherwise provided in the Investment Agreement) and the acts taken thereunder, the pursuit of confirmation of this Plan, the confirmation of this Plan, the consummation of this Plan, or the administration of this Plan or the property to be distributed under this Plan, except for their respective actions that constitute gross negligence or willful misconduct as determined by a Final Order entered by

a court of competent jurisdiction. For avoidance of doubt, none of the Released Parties shall have or incur any liability to, or be subject to any right of action by, any Holder of a Claim or an Interest, or any other party in interest, or any of their respective agents, employees, representatives, financial advisors, attorneys, or agents acting in such capacity, or affiliates, or any of their successors or assigns, or any other Released Party, for any Plan-Related Claim. Without limiting the foregoing, the Released Parties shall in all respects be entitled to reasonably rely upon the advice of counsel with respect to their duties and responsibilities under this Plan.

10.6 **Injunctions**

- Except as otherwise provided in this Plan or in any document, instrument, (a) release, or other agreement entered into in connection with this Plan or approved by order of the Bankruptcy Court, the Confirmation Order shall provide, among other things, that from and after the Effective Date all Persons or Entities who have held, hold, or may hold Claims against or Interests in the Debtors are (i) permanently enjoined from taking any of the following actions against the Estate(s), or any of their property, on account of any such Claims or Interests and (ii) permanently enjoined from taking any of the following actions against any of the Debtors, the Reorganized Debtors or their property on account of such Claims or Interests: (A) commencing or continuing, in any manner or in any place, any action, or other proceeding; (B) enforcing, attaching, collecting, or recovering in any manner any judgment, award, decree or order; (C) creating, perfecting, or enforcing any Lien or encumbrance; (D) asserting a setoff or right of subrogation of any kind against any debt, liability or obligation due to the Debtors; and (E) commencing or continuing, in any manner or in any place, any action that does not comply with or is inconsistent with the provisions of this Plan; provided, however, that nothing contained herein shall preclude such Persons or Entities from exercising their rights pursuant to and consistent with the terms of this Plan and the contracts, instruments, releases, indentures, and other agreements or documents delivered under or in connection with this Plan.
- (b) By accepting distributions pursuant to this Plan, each Holder of an Allowed Claim or Allowed Interest will be deemed to have specifically consented to the injunctions set forth in this <u>Section 10.6</u>.

10.7 Term of Bankruptcy Injunction or Stays

All injunctions or stays provided for in the Chapter 11 Cases under section 105 or 362 of the Bankruptcy Code, or otherwise, and in existence on the Confirmation Date, shall remain in full force and effect until the Effective Date.

10.8 Termination of Subordination Rights and Settlement of Related Claims

The classification and manner of satisfying all Claims and Interests under this Plan take into consideration all subordination rights, whether arising by contract or under general principles of equitable subordination, section 510(b) or 510(c) of the Bankruptcy Code, or otherwise. All subordination rights that a Holder of a Claim or Interest may have with respect to any distribution to be made pursuant to this Plan will be discharged and terminated, and all actions related to the enforcement of such subordination rights will be permanently enjoined.

Accordingly, distributions pursuant to this Plan to Holders of Allowed Claims and Allowed Interests will not be subject to payment to a beneficiary of such terminated subordination rights, or to levy, garnishment, attachment, or other legal process by a beneficiary of such terminated subordination rights; <u>provided</u>, <u>however</u>, that nothing contained herein shall preclude any Person or Entity from exercising their rights pursuant to and consistent with the terms of this Plan and the contracts, instruments, releases, indentures, and other agreements or documents delivered under or in connection with this Plan.

10.9 <u>Limitations of Applicability of Releases and Exculpatory Provisions to</u> Governmental Entities

Except with respect to Plan-Related Claims, nothing in the Confirmation Order or the Plan (i) shall effect a release of any claim by the United States Government or any of its agencies or any state and local authority whatsoever, including, without limitation, any claim arising under the Internal Revenue Code, the environmental laws or any criminal laws of the United States or any state and local authority against the Released Parties, but excluding any Plan-Related Claims, (ii) enjoin the United States or any state or local authority from bringing any claim, suit, action or other proceedings against the Released Parties for any liability whatever, including, without limitation, any claim, suit or action arising under the Internal Revenue Code, the environmental laws or any criminal laws of the United States or any state or local authority, but excluding any Plan-Related Claims, or (iii) exculpate any Released Party from any liability to the United States Government or any of its agencies or any state and local authority whatsoever, including, without limitation, any liabilities arising under the Internal Revenue Code, the environmental laws or any criminal laws of the United States or any state and local authority against the Released Parties, but excluding any liabilities in respect of any Plan-Related Claims. Notwithstanding anything to the contrary in the Confirmation Order or the Plan, the Securities and Exchange Commission ("SEC") expressly reserves its right to continue to investigate, and, in its sole discretion, prosecute and enforce any and all claims against any or all of the Debtors or the Reorganized Debtors arising from any prepetition violations by any Debtor of any of the U.S. securities laws other than Plan-Related Claims (collectively, the "Reserved SEC Claims"), including, without limitation, any claims for disgorgement of any benefits received by any Debtor as a result of any such violations and any claims for penalties imposed by the SEC in respect of any such violations. For avoidance of doubt, pursuant to Section 10.2(b) of the Plan, all Plan-Related Claims of the United States Government or any of its agencies or any state and local authority whatsoever shall be, and hereby are, released, waived and discharged. Nothing in the Confirmation Order or the Plan shall result in the discharge of any Reserved SEC Claims, and the SEC expressly reserves its rights to assert that any and all Reserved SEC Claims are non-dischargeable as against the Reorganized Debtors pursuant to Sections 1141(d)(6)(a) and 523(a)(2)(A) of the Bankruptcy Code. The SEC has advised the Court and the Debtors that as of the entry of this Order, it has not yet determined whether to assert any Reserved SEC Claims against any or all of the Debtors or Reorganized Debtors.

Dated: New York, New York August September 17, 2007

Respectfully Submitted,

BALLY TOTAL FITNESS HOLDING CORPORATION

By: <u>/s/ Don R. Kornstein</u>
Name: Don R. Kornstein

Title: Chief Restructuring Officer

Each Debtor Listed on Appendix 1

By: <u>/s/ Don R. Kornstein</u>

Name: Don R. Kornstein

Title: Chief Restructuring Officer

David S. Heller Richard A. Levy Keith A. Simon Sears Tower, Suite 5800 233 South Wacker Drive Chicago, Illinois 60606-6401 Telephone: (312) 876-7700 Facsimile: (312) 993-9767

-and-

Henry P. Baer, Jr. (HB-3866) Joseph Furst, III (JF-6136) 885 Third Avenue, Suite 1000 New York, New York 10022 Telephone: (212) 906-1200

Proposed Counsel for Debtors and Debtors-in-Possession

Appendix 1

Bally ARA Corporation

Bally Fitness Franchising, Inc.

Bally Franchise RSC, Inc.

Bally Franchising Holdings, Inc.

Bally Real Estate I LLC

Bally REFS West Hartford, LLC

Bally Sports Clubs, Inc.

Bally Total Fitness Corporation

Bally Total Fitness Franchising, Inc.

Bally Total Fitness Holding Corporation

Bally Total Fitness International, Inc.

Bally Total Fitness of California, Inc.

Bally Total Fitness of Colorado, Inc.

Bally Total Fitness of Connecticut Coast, Inc.

Bally Total Fitness of Connecticut Valley, Inc.

Bally Total Fitness of Greater New York, Inc.

Bally Total Fitness of Minnesota, Inc.

Bally Total Fitness of Missouri, Inc.

Bally Total Fitness of Philadelphia, Inc.

Bally Total Fitness of Rhode Island, Inc.

Bally Total Fitness of the Mid-Atlantic, Inc.

Bally Total Fitness of the Midwest, Inc.

Bally Total Fitness of the Southeast, Inc.

Bally Total Fitness of Toledo, Inc.

Bally Total Fitness of Upstate New York, Inc.

BTF Cincinnati Corporation

BTF Europe Corporation

BTF Indianapolis Corporation

BTF Minneapolis Corporation

BTF/CFI, Inc.

BTFCC, Inc.

BTFF Corporation

Greater Philly No. 1 Holding Company Greater Philly No. 2 Holding Company Health & Tennis Corporation of New York Holiday Health Clubs of the East Coast, Inc. Holiday/Southeast Holding Corp.

Jack LaLanne Holding Corp.

New Fitness Holding Co., Inc.

Nycon Holding Co., Inc.

Rhode Island Holding Company

Tidelands Holiday Health Clubs, Inc.

U.S. Health, Inc.

CH\<u>961384.15961384.20</u>

MANAGEMENT INCENTIVE PLAN

If On the Effective Date, regardless of whether the Backstop Rights Offering Effective Date Condition (as defined in the First Amended Joint Prepackaged Chapter 11 Plan of Reorganization of Bally Total Fitness Holding Corporation and Its Affiliate Debtors) is satisfied, the Management Incentive Plan shall be as set forth in Section I of this Exhibit. If or the Harbinger Investment Effective Date Condition (as each term is defined in the First Amended Joint Prepackaged Chapter 11 Plan of Reorganization of Bally Total Fitness Holding Corporation and Its Affiliate Debtors) is satisfied, the Management Incentive Plan shall be as set forth in Section II of this Exhibit.

CH/944363.11<mark>944363.12</mark>

SECTION 1

The Management Incentive Plan is comprised of two components: (1) a performance bonus ("Performance Bonus"), and (2) a restructuring transaction bonus ("RTB").

The Performance Bonus represents current annual incentive bonus awards for ongoing participation with the Debtors but with 2007 performance targets reset based on cash revenue and eash EBITDA as well as a discretionary component. The Performance Bonus is available to 27 eligible employees and has a maximum cumulative payout of approximately \$3.8 million.

The RTB represents an incentive for key employees to promptly complete a successful restructuring. The RTB is also payable if the employee is terminated without cause or constructively within six months of the restructuring transaction. The RTB is available to 9 eligible employees, divided into three groups (Class A, B and C), and has a maximum cumulative payout of \$3.2 million. The RTB is evidenced by individual bonus agreements with each eligible participant.

Class A participants include, Don R. Kornstein, the Company's Chief Restructuring Officer, Marc Bassewitz, the Company's Senior Vice President, Secretary and General Counsel, Bill-Fanelli, Senior Vice President Planning and Development, and John Wildman, the Company's Senior Vice President, Sales and Interim Chief Marketing Officer. They each will receive a maximum RTB if Bally consummates a restructuring transaction by October 31, 2007 (the "Target Date"), with incremental monthly reductions in the RTB for each month thereafter until the restructuring transaction is completed.

The maximum value of Mr. Kornstein's RTB is reduced based on the number of months after the expiration of the Target Date in which a plan of reorganization becomes effective. If the plan of reorganization does not become effective within eight months of the expiration of the Target Date, Mr. Kornstein will no longer be eligible to receive any RTB.

For the other Class A participants, the RTB will be reduced by 15% of the maximum for the first three months following the Target Date in which the reorganization transaction takes place, and then 10% of the maximum for the next two months, with no further reductions after the fifthmonth following the Target Date. The other Class A participants will receive the minimum bonus stated below if the plan of reorganization becomes effective after more than five months after the expiration of the Target Date. The maximum and minimum amount of each Class A participant's RTB is as followsbelow:

Class A Participant	Maximum RTB	Minimum RTB
Don R. Kornstein	\$2,100,000	\$0
Marc Bassewitz	\$375,000	\$131,250
Bill Fanelli	\$195,000	\$68,250
John Wildman	\$158,375	\$55,781

Class B and Class C participants will receive a set RTB, with no reduction if the restructuring is effected after the Target Date.

SECTION II

The Management Incentive Plan is comprised of two components: (1) a performance bonus ("Performance Bonus"), and (2) a restructuring transaction bonus ("RTB").

The Performance Bonus represents current annual incentive bonus awards for ongoing participation with the Debtors but with 2007 performance targets reset based on cash revenue and cash EBITDA as well as a discretionary component; provided, however, that the Performance Bonus for each of Mr. Bassewitz and Mr. Fanelli will in no event be less than 50% of the maximum Performance Bonus to which they are each eligible. The Performance Bonus is available to 26 eligible employees and has a maximum cumulative payout of approximately \$2.9 million. Mr. Kornstein is not eligible for a Performance Bonus, only a RTB as set forth below.

The RTB represents an incentive for key employees to promptly complete a successful restructuring. The RTB is also payable if the employee is terminated without cause or constructively within six months of the restructuring transaction. The RTB is available to 9 eligible employees, divided into three groups (Class A, B and C), and has a maximum cumulative payout of \$4.1 million. The RTB is evidenced by individual bonus agreements with each eligible participant.

The amount of Mr. Kornstein's, Mr. Bassewitz's, Mr. Fanelli's and Mr. Wildman's RTB is as follow:

Class A Participant	<u>Title</u>	<u>RT</u> B
Don R. Kornstein	Chief Restructuring Officer	\$3,000,000
Marc Bassewitz	Senior Vice President,	\$375,000
	Secretary and General Counsel	
Bill Fanelli	Senior Vice President Planning	\$195,000
	and Development	
John Wildman	Senior Vice President, Sales	\$158,375
	and Interim Chief Marketing	
	Officer	

CH\944363.11944363.12

Exhibit 3

[Non-Exclusive List of Assumed Lease of Non-Residential Real Property]

ClubNo	Club Name	Club Address 1	<u>City</u>	State	Landlord or other Applicable Party
833-74	TUCSON	4690 North Oracle Road	Tucson	AZ	American Furniture Company
833-70	CAVE CREEK	12235 N. Cave Creek Road	Phoenix	AZ	Blanc Blue, LLC
833-78	CHANDLER	4960 W. Ray Road; Suite A	Chandler	AZ	Chandler Sunset, LLC
833-80	GILBERT	855 N. Gilbert Road	Gilbert	AZ	Diamante-Gilbert Crossing, LLC
833-71	ESTES	15401 North 29th Avenue	Phoenix	AZ	Ecovalue, Inc.
833-77	SCOTTSDALE ("Pima/Shea")	Pima Crossing Shopping Center	Scottsdale	AZ	Regency Centers, L.P.
833-75	MESA	Three Fountains Plaza Phs2	Mesa	AZ	Richmond Properties of California, LLC c/o Three Fountains Plaza
833-72	PEORIA	5720 W. Peoria Avenue	Glendale	AZ	San Pablo Acres, LLC
833-79	FOOTHILLS	2475 North Swan Road	Tucson	AZ	Spector Revocable Trust
833-76	INDIAN SCHOOL	3919-21 E Indian School Rd	Phoenix	AZ	Yee Family Trust
608-04	SAN JOSE SOUTH	1570 Hamilton Avenue	San Jose	CA	1570 Hamilton Avenue Associates
604-24	LONG BEACH II	4438 E. Pacific Coast Hwy.	Long Beach	CA	4438 PCH, LLC
604-36	MACY'S PLAZA	Macy's Plaza at MCI Center	Los Angeles	CA	700 South Flower Plaza, LLC
604-36ST	MACY'S PLAZA (STORAGE)	700 South Flower Street	Los Angeles	CA	700 South Flower Plaza, LLC
604-15	CERRITOS	11881 East Del Amo Blvd.	Cerritos	CA	AG/BPG Del Amo, Inc.
604-21	HOLLYWOOD II	1628 N. El Centro Avenue	Los Angeles	CA	American Legion Hollywood Post 43
604-09PK	CHATSWORTH (PARKING)	9171 De Soto Ave.	Chatsworth	CA	AMF Bowling, Inc.
604-10	WEST LOS ANGELES	1914 South Bundy Drive	Los Angeles	CA	Anchor Investments
604-53	HUNTINGTON PARK	3081 Slauson Avenue	Huntington Park	CA	Bertz Development, Inc.
606-06	BLACKHAWK FITNESS	3464 Blackhawk Plaza Circle	Danville	CA	Blackhawk CenterCal, LLC
608-07	CAPITOL EXPRESSWAY	2920 Aborn Square	San Jose	CA	Borello Management Co., LLC
604-49	LAKE FOREST	23633 El Toro Road	Lake Forest	CA	Buchheim Properties III
610-04	SHAW & BLYTHE	4308 W. Shaw Avenue	Fresno	CA	Burgess, Smith & Wathen Partnership
610-04EX	SHAW & BLYTHE (EXPANSION)	4308 W. Shaw Avenue	Fresno	CA	Burgess, Smith & Wathen Partnership
610-04EX2	SHAW & BLYTHE (EXPANSION 2)	4308 W. Shaw Avenue	Fresno	CA	Burgess, Smith & Wathen Partnership
605-01	GLENDORA	750 West Route 66	Glendora	CA	Cal Pacific Hacienda Properties, LLC
608-05	HAYWARD	771 Jackson Street	Hayward	CA	Carstens Partners LP
604-50	OXNARD	451 W. Esplanade Drive	Oxnard	CA	Centro Watt Property Owner I, LLC
604-18	ANAHEIM	310 South Magnolia	Anaheim	CA	Century American Realty, Inc.
608-01	PLEASANT HILL	2316 Monument Boulevard	Pleasant Hill	CA	Contra Costa Retail Center, LLC
605-04	NORTHRIDGE	8946 Corbin Avenue	Northridge	CA	Corbin Properties
610-02	BLACKSTONE & SHAW	5137 N. Blackstone Avenue	Fresno	CA	Country Plaza Ltd. Partnership
608-10	ALEMANY	Oceanview Village Shopping Center	San Francisco	CA	Culligan Enterprises Subsidiary One, L.P. and Bay Terrace Apartments Subsidiary
605-22	CULVER CITY II	3827 Overland Avenue	Culver City	CA	Culver Center Partners - East #1, L.P.
604-54	DOWNEY	12074 Lakewood Boulevard	Downey	CA	Downey Landing SPE, LLC
610-05	PEACH & SHAW	284 W. Shaw Avenue	Clovis	CA	Drug Fair Building Partnership
604-38	SOUTH BAY	5001 El Segundo Boulevard	Hawthorne	CA	El Segundo Plaza, L.P.
604-33	ENCINO II	17401 Ventura Blvd.	Encino	CA	Encino Courtyard, LLC
604-16	TORRANCE	20040 Hawthorne Boulevard	Torrance	CA	Escondido Mission Village
604-53PK	HUNTINGTON PARK (PARKING)	5801 S. Boyle Avenue	Los Angeles	CA	Franklin Financial Management
605-03	GLENDALE	623 S. Central Avenue	Glendale	CA	Gene Pepper
608-09	THE PAVILION	150 South First Street, Unit 119	San Jose	CA	GIP Fairmont, LLC
604-44	VISTA	1928-32 Hacienda Drive	Vista	CA	Grewal Realty Holdings, LLC
604-52	RANCHO CUCAMONGA	10848 Foothill Boulevard, Suite 110	Rancho Cucamonga	CA	Hall Investment Company
605-03PK	GLENDALE (PARKING #1)	318 W. Maple Street	Glendale	CA	Harout Yepremian
605-03PK2	GLENDALE (PARKING #2)	318 W. Maple Street	Glendale	CA	Harout Yepremian

604-05	WEST COVINA	441 North Arrise Avenue	West Covina	CA	Haccon Deal Estate Doutsonahu
608-05OF/PK	HAYWARD (OFFICE/PARKING)	441 North Azusa Avenue 24089 Watkins Street	West Covina Hayward	CA	Hassen Real Estate Partnershp Hoi Liang Phua (aka Bill Phua) and Linli Lee
604-09	CHATSWORTH	9143 De Soto Avenue	Chatsworth	CA	Indian Farms Corporation
604-07	HOLLYWOOD II (PARKING)	1607 North Gower Street	Los Angeles	CA	Irene H. Tanji
605-01	GLENDORA	750 West Route 66	Glendora	CA	Kan Lai
604-31	CORONA	390 McKinley Street	Corona	CA	Kimco Realty Corporation
604-46	NORTH LONG BEACH	5810 Downey Avenue	Long Beach	CA	LB South & Downey, LLC
608-03	WALNUT CREEK	2150 N. Broadway	Walnut Creek	CA	•
605-16	MANHATTAN BEACH	1133 Artesia Blvd.	Manhattan Beach	CA	Lupoi Properties, Ltd. Manhattan Place, Inc.
608-08	SOUTH SAN FRANCISCO	180 E. El Camino Real	South San Francisco	CA	Marbella Realty, LLC
604-51	INGLEWOOD	3531 W. Century Boulevard	Inglewood	CA	Marketplace at Hollywood Park, L.P.
604-31	SOUTH COAST	3701 S. Plaza Drive	Santa Ana	CA	Metro Town Square, LLC
606-04	DANVILLE FITNESS	101A & B Town and Country Drive	Danville	CA	Montair Associates, Ltd.
604-23	MONTEBELLO	2222 West Beverly Blvd.	Montebello	CA	Montebello Beverly Properties
605-21	SIMI VALLEY	1313 E. Los Angeles Ave.	Simi Valley	CA	Mountaingate Associates, LLC
604-09EX	CHATSWORTH (EXPANSION)	9119 DeSoto Avenue	Chatsworth	CA	Mr. Ran Radzewsky and Mr. Eliahu NerGaon
610-07	FIRST & HERNDON	6735 North First Street	Fresno	CA	Ms. Della Wathen
604-41	PUENTE HILLS II	1545 Hanover Road		CA	New Age Kaleidoscope, LLC
604-41 604-41OF	PUENTE HILLS II (OFFICE)	1661 Hanover Road; Suite 101A	City of Industry City of Industry	CA	New Age Kaleidoscope, LLC
604-20	HUNTINGTON BEACH	17091 Beach Boulevard	Huntington Beach	CA	NF Huntington Plaza, LP
605-05	NORWALK	11703 E. Rosecrans Ave.	Norwalk	CA	Norwalk Town Square Management, Inc.
604-17	MONTCLAIR II	9385 Monte Vista	Montclair	CA	Pacific Montclair LLC
605-07	PASADENA	45 South Arroyo Parkway	Pasadena	CA	Paul P. Rusnak
608-06	FREMONT	35630 Fremont Blvd.	Fremont	CA	PK II Brookvale SC, LP
604-22	RIVERSIDE II	3490 Madison Avenue	Riverside	CA	PRP Investors Madison, LLC
604-28	FULLERTON	Orangefair Mall Shopping Center	Fullerton	CA	Rancho Fullerton, L.P.
604-28	SAN BERNARDINO II	Tri-City Corp. Center	San Bernardino	CA	Rancon Realty Fund V Subsidiary LLC
604-19	MISSION VALLEY	405 Camino del Rio South	San Diego	CA	Recabaren Ranch Development, Inc.
604-43	SAN MARCOS	2055 Montiel Road	San Marcos	CA	Red Rock Center, LLC
610-07	FIRST & HERNDON	6735 North First Street	Fresno	CA	Robart Grant Robarts and Betty Robarts
604-10PK	WEST LOS ANGELES (PARKING)	1935 Armacost Avenue	Los Angeles	CA	Robert Tapper
604-35	ROSEMEAD	3508 Rosemead Boulevard	Rosemead	CA	Rosemead Place, LLC
604-90WH	WAREHOUSE (SANTA FE)	10241 Matern Place	Santa Fe Springs	CA	Santa Fe Lot Five
604-45	SCRIPPS RANCH	9850 Hibert Street	San Diego	CA	Scripps Windmill - D.Q., LLC
604-11	NORTH HOLLYWOOD	13069 Victory Boulevard	North Hollywood	CA	Sir Francis Drake Holdings, LLC
004 11		13007 Victory Boulevalu	Horar Horry wood	CH	on Traicis Diake Holdings, EDC
511-19862SD	NORWALK RSC (SATELLITE DISH/ANTENNAE	12440 E. Imperial Highway	Norwalk	CA	Sonnenblick Del Rio Norwalk, LLC et al
511-19862ST	NORWALK RSC (STORAGE)	12440 E. Imperial Highway	Norwalk	CA	Sonnenblick Del Rio Norwalk, LLC et al
511-19862	NORWALK RSC	12440 East Imperial Highway; Suite 300 and 750	Norwalk	CA	Sonnenblick Del Rio Norwalk, LLC, et al.
610-07	FIRST & HERNDON	6735 North First Street	Fresno	CA	Springsted & Wathen
604-21PK	HOLLYWOOD II (PARKING)	1607 North Gower Street	Los Angeles	CA	Suzuki Children's Trust dated August 25, 2004
604-01	NATIONAL CITY	1910 Sweetwater Road	National City	CA	Sweetwater Associates Limited Partnership
604-37	PALOS VERDES	28901 S. Western Avenue; Suite 315 and 321	Rancho Palos Verdes	CA	Terraces Rancho Palos Verdes, LLC
604-55	ORANGE II	2095 East Katella Avenue	Anaheim	CA	The Stadium Crossings Company, LLC
605-09	STUDIO CITY	11301 Ventura Blvd.	Studio City	CA	Ventura Associates LP
610-01	CLOVIS	781 W. Shaw Avenue	Clovis	CA	Western Village, Inc.
604-40	TUSTIN	630 El Camino Real	Tustin	CA	William R. Zappas, Trustee of The William R. Zappas Trust dated 5/14/90

610.06	MADERA	2220 W. Clavaland Avanua	Madam	CA.	WDI Colden State I I C
610-06 608-02	SAN LEANDRO	2330 W. Cleveland Avenue 567 Floresta Boulevard	Madera San Leandro	CA CA	WRI Golden State, L.L.C. YIP Holdings Two, LLC
641-06	AURORA	13801 East Exposition Avenue	Aurora	CO	Aurora City Square, LLC
641-13	LAKEWOOD III	360 S. Teller St.	Lakewood	co	Belmar Mainstreet Holdings I, LLC
641-99OF2	DENVER AREA OFFICE	355 S. Teller St., Suite 220	Lakewood	co	Belmar Mainstreet Holdings I, LLC
831-60	NORTH ACADEMY		Colorado Springs	co	-
641-09	GALLERIA	5620 No. Academy Boulevard 720 So. Colorado Boulevard	Glendale	co	Deserpa Colorado One, LLC
	MISSION COMMONS ("Westminster")	7635 West 88th Avenue	Arvada	co	Galleria Acquisition, Inc. Mission Commons LLC
641-07	WAREHOUSE (SHERIDAN)			CO	
641-90WH3 831-64	SOUTH ACADEMY II	4131 S. Natches Ct.	Sheridan	co	Oxford Santa Fe, Ltd.
641-08	SOUTHWEST DENVER	1801 S. Academy Blvd.	Colorado Springs Littleton	co	Pace Bally Plaza LLC
641-14	THORNTON II	Bowles Avenue Marketplace	Thornton	co	RN BAM Acquisition LLC
831-90WH1	WAREHOUSE (CO. SPRGS)	Thornton Town Center		co	Thornton Towne Center 05 A, LLC
	ENGLEWOOD II	2345 N. Academy Place	Colorado Springs		Wayne N. Johnson dba Academy Place Storage
641-12	WEST HARTFORD PARKING 3	3435 S. Inca St.	Englewood	CO	Weingarten/Miller/Englewood Joint Venture, a Texas joint venture
865-04PK3	MANCHESTER	101 South Street	West Hartford	CT	101 South, LLC
865-03	MIDDLEBURY	515 West Middle Turnpike	Manchester	CT	525 West Middle Turnpike Associates Limited Partnership
866-11	HAMDEN II	930 Straits Turnpike	Middlebury	CT	C.L.P.V., LLC
866-14	STRATFORD	2380 Dixwell Avenue	Hamden	CT	Hamdor Properties
002-76	WEST HARTFORD PARKING	411 Barnum Avenue Cutoff	Stratford,	CT	Kramont Operating Partnership, L.P.
865-04PK	WEST HARTFORD PARKING 2	1031 New Britain Avenue	West Hartford	CT	United Tool and Die Company
865-04PK2	L STREET	1031 New Britain Avenue	West Hartford	CT	United Tool and Die Company
002-04		2000 L Street NW, #1B	Washington	DC	2000 L Co. LLC
002-04RF	L STREET (ROOF)	2000 L Street NW	Washington	DC	TrizecHahn 2000 L Street LLC
853-18	COURTROOMS	750 W. Sunrise Boulevard	Ft. Lauderdale	FL	750 Sunrise Associates
853-07	MIAMI	8181 NW 12th Street	Doral	FL	Adler Office Associates
853-08	DAVIE	2701 S. University Drive	Davie	FL	Bond-Bally II Delaware Business Trust
779-08	DALE MABRY	14350 North Dale Mabry Hwy	Tampa	FL	Carrollwood Partners, LLC
780-01	WINTER PARK	1865 State Road 436	Winter Park	FL	Casselberry Partnership
780-01EX	WINTER PARK (2ND FLOOR)	1865 State Road 436	Winter Park	FL	Casselberry Partnership
853-22	PEMBROKE PINES	10840 Pines Boulevard	Pembroke Pines	FL	CP Pembroke Pines, LLC
779-05	HIGHWAY 19	28272 U.S. Hwy 19 North	Clearwater	FL	D.B. Enterprises, Inc.
780-03	ALTAMONTE SPRINGS	733 West Highway 436	Altamonte Springs	FL	G & J Management Company
779-02	HILLSBOROUGH	4340 W. Hillsborough Ave.	Tampa	FL	Hillsborough Galleria
780-05	SUN PLAZA	6385 W. Colonial Drive	Orlando	FL	KCH Properties & Passive Investors Inc
780-04	BAYHILL	4850 Lawing Lane	Orlando	FL	MDR Health Club Limited Partnership
853-24	MILLER SQUARE	13838 S.W. 56th Street	Miami	FL	Miller Square, LLC
853-27	CORAL GABLES II	Miracle Marketplace	Miami	FL	Miracle Marketplace LLC
853-15	CORAL WAY	3301 Coral Way	Miami	FL	Miracle Marketplace, LLC
853-04	BOCA RATON	21069 Military Trail	Boca Raton	FL	MSW 21069 Military Trail, L.L.C.
780-02	ORLANDO	4650 South Semoran Blvd.	Orlando	FL	Orlando Partnership
780-02EX	ORLANDO (EXPANSION)	4650 South Semoran Blvd.	Orlando	FL	Orlando Partnership
853-23	HIALEAH	1901 West 39th Street	Hialeah	FL	Realty Income Corporation d/b/a Realty Income Properties, Inc.
853-18PK	COURTROOMS (PARKING)	915 NW 7th Terrace	Ft. Lauderdale	FL	Robert Elmore
779-90	WAREHOUSE (TAMPA)	4410 West Crest Avenue	Tampa	FL	Roth Investment Partnership
779-01	TAMPA	13455 U.S. 19th North	Clearwater	FL	S & T Investments-Clearwater Partnership
853-12	KENDALL I	14875 South Dixie Highway	Miami	FL	Smorgasbord Management Company
853-26	SOUTHLAND	Southland Mall Shopping Center	Miami	FL	Southland Mall Properties LLC

853-20	KENDALL MALL	8851 S.W. 107th Avenue	Miami	FL	SVF Kendall Miami, LLC
853-03PK	WEST PALM OLYMPIAD (PKG)	603 Village Blvd.	West Palm Beach	FL	The Governors, a Pennsylvania Partnership
853-11	NORTH MIAMI BEACH	16720 NE 6th Avenue	North Miami Beach	FL	Toys 'R' Us - Delaware, Inc.
853-90WH	WAREHOUSE (POMPANO BEACH)	1411 S.W. 30th Avenue	Pompano Beach	FL	Warehouses of Pompano Beach, L.C.
853-05	WATERWAYS	3455 NE 207th Street	Miami	FL	Waterways Plaza, LLC
853-21	COMMERCIAL	959 East Commercial Blvd.	Oakland Park	FL	Weingarten Nostat, Inc.
853-90	WAREHOUSE (DANIA)	1991 Tigertail Boulevard; Suite A	Dania	FL	Wilson Hollywood Showroom, LLC
853-19	CORAL SPRINGS	620 Riverside Drive	Coral Springs	FL	Woolbright Coral Springs II, LLC
853-03	WEST PALM OLYMPIAD	501 Village Boulevard	West Palm Beach	FL	WPB Village 1, LLC and WPB Village 2, LLC
002-38	JIMMY CARTER	5050 Jimmy Carter Blvd.	Norcross	GA	Carter Rockbridge Plaza, LLC
633-01	TOWN CENTER	2600 Prado Lane	Marietta	GA	DDR MDT Towne Center Prado LLC
002-51	NORTHLAKE	3993 Lavista Road	Tucker	GA	Northlake Festival, LLC
634-01	GWINNETT	2300 Pleasant Hill Road	Duluth	GA	Phillips Investments, LLC
637-01	ALPHARETTA	11060 Alpharetta Highway	Roswell	GA	Roswell Shopping Center
635-01	STONE MOUNTAIN	5370 Highway 78 S.	Stone Mountain	GA	Stone Mountain Improvements, LLC
002-57	CUMBERLAND	2211 Cobb Parkway	Smyrna	GA	Target Corporation
701-23	HYDE PARK	1301 E. 47th Street	Chicago	IL	1301 East, L.L.C.
701-14	EAST WASHINGTON	25 East Washington Street	Chicago	IL	25 East Washington Associates, L.P.
712-19	NAPERVILLE	1531 N. Aurora Road	Naperville	IL	Bridgeview Bank and Trust Company, as Trustee under Trust Agreement dtd 3/1/93
701-12	CENTURY CITY	2828 North Clark Street	Chicago	IL	Century Mall, LLC
701-15	SKOKIE	3220 West Touhy Ave.	Skokie	IL	Chetty Sevugan
701-31	ROGERS PARK	Gateway Centre Plaza	Chicago	IL	Combined Development-Howard, L.L.C.
786-01	ROCKFORD	3800 East State Street	Rockford	IL	Crest Commercial, Inc.
712-15	CUMBERLAND	5444 N. Cumberland Avenue	Chicago	IL	Cumberland Centre, L.L.C. c/o GVA Williams
701-16-SN	SIX CORNERS SIGN	3912 N. Cicero Ave.	Chicago	IL	Donna Schierer Calderon and Lorenzo Calderon c/o Schierer & Popp Jewelers, Inc.
701-20	NORTH RIVERSIDE	1770 South Harlem Avenue	N. Riverside	IL	Exoho Harlem Associates, LLC
714-15	FAIRVIEW HEIGHTS III	49 Ludwig Drive	Fairview Heights	IL	Fairview Heights Investors, LLC
712-18	NORTHSHORE	Prairie View Shopping Ctr.	Morton Grove	IL	Federal Construction, Inc.
712-16	COUNTRYSIDE	5917 So. La Grange Road	Countryside	IL	G & W Partners
712-14	DEERFIELD	260 Waukegan Road	Deerfield	IL	Gateway Fairview, Inc. c/o RREEF Management Company
712-17	GLENDALE HEIGHTS	265 E. Army Trail Road	Glendale Heights	IL	Hilco Bally Glendale Heights, LLC
701-26	TINLEY PARK	16040 South Harlem Avenue	Tinley Park	IL	Inland Commercial Property Management, Inc. as agent for Inland Real Estate Corp
701-18	CHATHAM RIDGE	112 W. 87th Street; Suite N.	Chicago	IL	Inland Real Estate - Illinois, L.L.C.
701-16	SIX CORNERS	3970 N. Milwaukee	Chicago	IL	Inland Real Estate Corporation
241-02	RIVER EAST	355 E. Grand Avenue	Chicago	IL	Intercontinental River East, LLC
701-16PK	SIX CORNERS (PARKING)	3926 N. Milwaukee Avenue	Chicago	IL	La Salle Bank N.A
712-10	MOUNT PROSPECT	225 West Rand Road	Mount Prospect	IL	La Salle National Bank as Trustee under Trust Agreement dated February 1, 1961
712-12	SCHAUMBURG	1020 Meacham Road	Schaumburg	IL	LaSalle Bank National Association, Successor Trustee under Trust No. 47248
712-05	VERNON HILLS	30 Phillips Road	Vernon Hills	IL	LaSalle National Bank, Trustee u/t/a dated 4/27/78 and known as Trust No. 54254
701-09	OAK LAWN	6700 West 95th Street	Oak Lawn	IL	Little Company of Mary Hospital and Health Care Centers
701-21	MELROSE PARK II	1000 West North Avenue	Melrose Park	IL	McCormick City-G, LLC
701-09PK	OAK LAWN (PARKING)	9400 South Oak Park Avenue	Oak Lawn	IL	Oak Lawn Park District
511-19852	CHICAGO CORPORATE OFFICE	8700 West Bryn Mawr Ave.	Chicago	IL	PR II Presidents Plaza JV, LLC
511-19852EX	CHICAGO CORP. OFFICE (EXP)	8700 W. Bryn Mawr Avenue	Chicago	IL	PR II Presidents Plaza JV, LLC
715-01	RIVER CITY	800 South Wells Street	Chicago	IL	River City Leasehold, L.L.C.
701-10	RIVER OAKS	1500 Torrence Avenue	Calumet City	IL	Sanhill Company, Inc.
712-20	SCHAUMBURG WEST	2411-A Schaumburg Road	Schaumburg	IL	Scharrington Belmont, LLC

701-29	LAKEFRONT	1603 N. Lakeshore Drive	Chicago	IL	The Chicago Park District
712-11	MATTESON	4701 Lincoln Mall Drive	Matteson	IL	The Cirignano Family Limited Partnership
701-30	VILLA PARK	102 W. Roosevelt Rd.	Villa Park	IL	Villa Oaks, LLC
701-25	WEBSTER PLACE	1455 West Webster Avenue	Chicago	IL	Webster Place SEC, L.L.C.
701-17	WESTPORT COMMONS	3348 W. 87th St.	Chicago	IL	Westport Associates Limited Partnership
819-29	GREENWOOD	517 U.S. 31 North	Greenwood	IN	BTF Greenwood Corporation
819-24	INDIANAPOLIS ("Keystone")	8831 Keystone Crossing	Indianapolis	IN	Duke Investment Co. II
745-01	INDIAN CREEK	6700 West 110th Street	Overland	KS	Columbia Equities L.P.
745-04	OLATHE	15274 W. 119th St.	Olathe	KS	Northridge Investors, L.L.C.
002-73	BRIGHTON LANDING	25 Guest Street	Brighton,	MA	BV Development - Brighton Landing
002-74	SALEM	29 Traders Way	Salem,	MA	Highlander Plaza Limited Partnership
002-69PK	WOBURN (PARKING)	27 Normac Road	Woburn	MA	Hunneman Commercial Management
002-25	WORCESTER	Lincoln Square Plaza	Worcester	MA	Inland Western Worcester Lincoln Plaza, L.L.C.
002-71	DOWNTOWN CROSSING	17 Winter Street	Boston,	MA	Janelon Associates
002-55	CAMBRIDGE	1815 Massachusetts Avenue	Cambridge	MA	Lesley University
002-77	TOWN LINE	285 Middlesex Avenue	Medford	MA	Maldex Limited Partnership
002-52	NORTHSHORE	Routes 128 & 114	Peabody	MA	Mall at Northshore LLC
002-69	WOBURN	27 Normac Road	Woburn,	MA	Normac Road Realty Trust
002-56	LOWELL	203 Plain Street	Lowell	MA	Philip C. Haughey, Trustee of
002-72	DARTMOUTH	200 State Road	North Dartmouth	MA	PR North Dartmouth LLC
002-59	REVERE	561 Squire Road	Revere	MA	Wesley LLC
002-37	PRINCE GEORGES	3500 East West Highway	Hyattsville	MD	(Add'l) Mall at Prince George's Management Offices
002-50	ROCKVILLE	11820 Rockville Pike	Rockville	MD	Federal Realty Investment Trust
002-23	GOVERNOR PLAZA ("Glen Burnie")	6629 Ritchie Highway, #201	Glen Burnie	MD	Governor Plaza Associates
002-46	GREENBELT	7415 Greenbelt Road	Greenbelt	MD	Greenway Plaza LLC
511-19586	TOWSON OFFICE	Hampton Plaza	Towson	MD	Hampton Plaza Limited Liability Limited Partnership
002-63	GAITHERSBURG	Market Square Shopping Center	Gaithersburg	MD	Kentlands II, L.L.C.
002-17	HAMPTON MALL	9171 Central Avenue	Capitol Heights	MD	Kingdom Management, LLC
002-03	NORTHPOINT	Northpoint Plaza Shop Ctr	Baltimore	MD	Northpoint Shopping Center Limited Partnership
002-08	ROUTE 40	6516 Baltimore National Pike	Baltimore	MD	Pike Park Associates LLLP
002-37	PRINCE GEORGES	3500 East West Highway	Hyattsville	MD	Preit Services, LLC
002-79	BELCREST	2970 Belcrest Cetner Drive, Suite 201	Hyattsville	MD	Prince Georges Station Retail, LLC
002-02	WHITE MARSH	8221 Town Center Drive	Nottingham	MD	Realty Income Corporation
002-90WH2	WAREHOUSE (HANOVER)	7521 Connelley Dr., Ste E	Hanover	MD	St. John Properties, Inc.
002-65	TOWSON II	1 East Joppa Road	Towson	MD	Towson Circle Joint Venture LLP and Towson Circle LLC
002-78	WHEATON II	Westfield Shoppingtown Wheaton	Wheaton	MD	Wheaton Plaza Regional Shopping Center, LLP
002-75	PORTLAND	275 Marginal Way	Portland,	ME	Cardente Properties
702-08	BLOOMFIELD HILLS	6420 Telegraph Road	Bloomfield Hills	MI	Bloomfield Hills Associates
702-22	REDFORD PLAZA	9359 Telegraph Road, #22	Redford	MI	Centro Bradley SPC 3 LLC
702-23	COLUMBIA CENTER	203 W. Big Beaver Road	Troy	MI	Columbia Center Limited Partnership
702-25	BEL AIR	9800 E. Eight Mile Road	Detroit	MI	DDRC Michigan LLC
702-26	DEARBORN II	Fairlane Town Center	Dearborn	MI	Fairlane Town Center
702-09PK	VAN DYKE (PARKING)	30100 Van Dyke	Warren	MI	Karam Company
716-03	WATERFORD	1490 No. Oakland Blvd.	Waterford	MI	Leo P. Sklar and Libby Sklar
702-21	NOVI	Novi Town Center	Novi	MI	Novi Town Center Investors LLC
702-16	STERLING HEIGHTS	Clinton Valley Mall	Sterling Heights	MI	Ramco-Gershenson Properties, L.P.
702-15	WOODHAVEN	23303 Allen Road	Woodhaven	MI	Second Woodhaven Associates

702-12	ST. CLAIR SHORES	20701 E. Eight Mile Road	St. Clair Shores	MI	St. Clair Shores Health Club Associates
702-12 702-08PK	BLOOMFIELD HILLS (PKG)	4104 W. Maple	Bloomfield Hills	MI	The Goldsmith, Ltd.
716-03	WATERFORD	1490 No. Oakland Blvd.	Waterford	MI	The Morris Rochlin Trust UAD 3/3/94
838-44PK3	FRIDLEY (PARKING 3)	Paco Building	Fridley	MN	Alexander A. Levitan
838-48	SUN RAY	Sun Ray Shopping Center	St. Paul	MN	Bradley Operating Limited Partnership
838-46	RICHFIELD	100 West 66th Street	Richfield	MN	Centro Bradley SPE 1 LLC
838-47	ST. PAUL	1166 University Avenue	St. Paul	MN	Inland Ryan, L.L.C.
838-45PK	LITTLE CANADA (PARKING)	71 Minnesota Avenue	Little Canada	MN	Schroeder Development Co.
838-44PK2	FRIDLEY (PARKING 2)	Paco Industrial Park	Fridley	MN	Slawik Properties, a division of HAR-MAR, Inc.
838-45	LITTLE CANADA	71 Minnesota Avenue	Little Canada	MN	Supervalu Inc.
838-44PK	FRIDLEY (PARKING 1)	Commerce Circle	Fridley	MN	Trustee Group Realty Partners IV
714-09	CRESTWOOD	9744 Watson Road	Crestwood	MO	Joseph J. Barbieri Trust dated 2/11/93
714-09	CRESTWOOD	9744 Watson Road	Crestwood	MO	Judy E. Schrampf
714-09	CRESTWOOD	9744 Watson Road	Crestwood	MO	Orazio J. Gianino Trust dated 9/1/86
714-07	CLAYTON	7393 Forsyth	Clayton	MO	SRC #01 Partners
850-68	CHARLOTTE	5404 Central Avenue	Charlotte	NC	K & K Real Estate, Inc.
850-68PK	CHARLOTTE (PARKING)	5416-5422 Central Avenue	Charlotte	NC	K & K Real Estate, Inc.
850-70	SOUTH BOULEVARD	5601 South Blvd.	Charlotte	NC	Kimco Development of Tyvola, Inc.
850-69	PINEVILLE	8700 Pineville Matthews Rd., Suite 400	Charlotte	NC	Tisano Realty, Inc.
008-48PK	SPRINGFIELD (PARKING)	111 Route 22	Springfield	NJ	Babble, Inc.
008-85	JERSEY CITY	912-920 Bergen Avenue	Jersey City	NJ	Empress Enterprises 920, Inc.
008-80	UNION CITY	3149 Kennedy Boulevard	North Bergen	NJ	FC Treeco/Columbia Park LLC
008-65PK	EAST BRUNSWICK (PARKING)	244 Route 18	East Brunswick	NJ	Franklin Inn East Brunswick
008-48	SPRINGFIELD	99 U.S. Highway 22	Springfield	NJ	H.E.C. Holding Company
008-32	ENGLEWOOD CLIFFS	150 Sylvan Avenue	Englewood Cliffs	NJ	JEM CO.
022-02	ECHELON	1160 White Horse Road	Voorhees	NJ	KBH White Horse LLC & 1160 White Horse LLC
008-63	BRICK	193 Chambers Bridge Road	Brick	NJ	Ocean Ice Palace
002-62	MAPLE SHADE	2834 Route 73 North	Maple Shade	NJ	PL Maple Shade, LLC
008-65	EAST BRUNSWICK	8 Edgeboro Road	East Brunswick	NJ	Realty Income Corporation
008-32PK	ENGLEWOOD CLIFFS (PARKING)	150 Sylvan Avenue	Englewood Cliffs	NJ	Rothman & Grossman
008-82	SADDLE BROOK	189 Route 46 West	Saddle Brook	NJ	Tri State Commercial Realty, LLC
008-77	CLARK	140 Central Avenue	Clark	NJ	Villa Contracting Co., Inc.
033-02	BILTMORE	Bank of America Plaza	New York	NY	335 Madison Avenue
033-03	PENTA	139 West 32nd Street	New York	NY	401 Commercial, L.P.
008-26	MADISON AVENUE	45 East 55th Street	New York	NY	551 Madison Avenue Office, LLC
008-73	SIXTH AVENUE	641 Avenue of the Americas	New York	NY	641 LLC
008-23	KEW GARDENS	80-02 Kew Gardens Road	Queens	NY	80-02 Leasehold Company, L.P.
008-47	SHEEPSHEAD BAY	1720-28 Sheepshead Bay Rd.	Brooklyn	NY	A & J Properties LLC
863-41	GREECE	3160 W. Ridge Road	Rochester	NY	ABC, LLC
008-69	LEVITTOWN	2935 Hempstead Turnpike	Levittown	NY	Aldrich Management Co.
008-70	BAY PLAZA	2116 Bartow Avenue	Bronx	NY	Bay Plaza Community Center, L.L.C.
008-30	KINGS HIGHWAY	2032 Coney Island Avenue	Brooklyn	NY	BMS Realty Company
008-76	NEW ROCHELLE	33 LeCount Place	New Rochelle	NY	Capelli Enterprises, Inc.
008-38	WESTBURY	373 Old Country Road	Carle Place	NY	Country Plaza Associates
008-38EX	WESTBURY (EXPANSION)	373 Old Country Road	Carle Place	NY	Country Plaza Associates
631-99OF2	CRUNCH CORP OFFICE - 19TH FLOOR	11 - 13 East 26th Street	New York	NY	East Twenty Sixth Joint Venture
008-79	TILDEN	2163 Tilden Avenue	Brooklyn	NY	Flatbush Delaware Holding LLC

863-40	HENRIETTA	1225 Jefferson Road	Rochester	NY	Flaum Commons, L.P.
008-83	FORDHAM ROAD	2503 Grand Concourse	Bronx	NY	Fordham Associates LLC
008-49NL	ELMHURST	75-28 Queens Boulevard	Elmhurst	NY	High Definition Realty, LLC
008-61	RIVERDALE	298 West 231st Street	Bronx	NY	Jacques Strauss and Evelyn Wells, Trustees U/W/O Fred Strauss
008-15	COPIAGUE	1147 Sunrise Highway	Copiague	NY	Kir Copiague, L.P.
008-25	LITTLE NECK	245-24 Horace Harding Blvd	Little Neck	NY	Little Neck Commons L.L.C.
008-14	BAYSHORE II	1175 Sunrise Highway	Bayshore	NY	Martin Zelman
008-39PK2	ROCKVILLE CENTRE (ADD PRK)	Sunrise Hwy. & Montauk Hwy	Rockville Centre	NY	New York State Department of Transportation
008-22	LAKE GROVE	22 Middle Country Road	Lake Grove	NY	NP/I & G Lake Grove, LLC
863-32	AMHERST	3880 East Robinson Road	Amherst	NY	Rainbow Racquet Club
008-43	YONKERS	589 Tuckahoe Road	Yonkers	NY	Realty Income Corporation
008-84	106TH STREET	1915 Third Avenue	New York	NY	Ross & Ross, LLC
008-37	BENSONHURST	1921 86th Street	Brooklyn	NY	Selv Enterprises LLC
008-46	PORT CHESTER	260 Boston Post Road	Port Chester	NY	Simone Development Company, L.L.C.
008-39	ROCKVILLE CENTRE	60 Merrick Road	Rockville Centre	NY	Sunoce Properties, Inc.
008-81	JAMAICA CENTER	159-26 Jamaica Avenue	Jamaica	NY	The Mattone Group Jamaica Co., LLC
863-32PK	AMHERST (PARKING)	3880 East Robinson Road	Amherst	NY	Tonawanda Housing, Inc.
033-03EX	PENTA (Expansion)	A&S Plaza Mall	New York	NY	VNO 100 West 33rd Street LLC
008-71	STATEN ISLAND	2040 Forest Avenue	Staten Island	NY	Vornado Forest Plaza, L.L.C.
008-56	WOODBRIDGE	541 Main Street	Woodbridge	NY	Wolff & Samson Attorney Trust Account
033-04	WORLDWIDE PLAZA	350 West 50th Street	New York	NY	WWP Amenities MPH Partner, LLC
707-04	CENTERVILLE	Yankee Station Shop Center	Centerville	ОН	Dayton Portfolio, LP
705-99ST	STORAGE (TOLEDO)	4835 South Avenue: Unit 427	Toledo	ОН	Do-It-Yourself Storage, Inc.
705-99ST2	STORAGE (TOLEDO) #2	4835 South Avenue; Unit 426	Toledo	ОН	Do-It-Yourself Storage, Inc.
705-06	AIRPORT HIGHWAY	5424 Airport Highway	Toledo	ОН	Health & Fitness Company
819-90WH2	WAREHOUSE (TALLMADGE 2)	391 Geneva Avenue	Tallmadge	ОН	J.D. Williamson Construction Co., Inc.
819-14	STRONGSVILLE	14783 Pearl Road	Strongsville	ОН	K&F Partnership
819-22	BROADVIEW	7955 Broadview Road	Broadview Heights	ОН	Khouri Group # 2 LLC
819-01	CANTON II	4733 Hills and Dales Rd NW	Canton	ОН	LXP Canton, Inc.
705-05	SYLVANIA	5215 Monroe Street	Toledo	ОН	Monroe Street Special, Inc.
819-52	BLUE ASH	4780 Cornell Road	Blue Ash	ОН	Pancake Properties Partnership P.L.L.
819-23	WESTLAKE	1255 Columbia Road	Westlake	ОН	Perfect Balance
707-03	ENGLEWOOD	8423 North Main Street	Dayton	ОН	Randolph Investment, LLC
819-20	BEACHWOOD	3600 Park East Drive	Beachwood	ОН	Realty Income Corporation
819-54	SEVERANCE	Severance Town Center	Cleveland Heights	ОН	Severance SPE LeaseCo, L.L.C.
819-09	COLUMBUS NORTH II	2439 Fuji Drive	Columbus	OH	Textron Financial Corporation
615-11	ALOHA	17800 SW Kinnaman Road	Aloha	OR	Bales Unico, Inc.
615-09	MILWAUKIE	4330 SE International Way	Milwaukie	OR	Cheyenne Properties
615-14	FAIRVIEW	1415 N.E. 223rd Avenue	Fairview	OR	Dave and Pam Ellis
615-14	FAIRVIEW	1415 N.E. 223rd Avenue	Fairview	OR	Dejac Investments, Inc.
615-14	FAIRVIEW	1415 N.E. 223rd Avenue	Fairview	OR	Farouk Al-Hadi
615-10PK	SPORTS CENTER (PARKING)	9000 S.W. Beaverton Hillsdale Highway	Portland	OR	Jesuit High School
615-01	BEAVERTON	9650 SW Nimbus Avenue	Beaverton	OR	Nimbus Center, LLC
615-06	LAKE OSWEGO	15353 SW Sequoia Parkway	Portland	OR	Pacific Realty Associates, L.P.
615-12	CEDAR MILL	10860 SW Barnes Road	Portland	OR	Peterkort Towne Square, LLC
615-02	EAST PORTLAND	10414 S.E. Washington Street	Portland	OR	Plaza 205 Shopping Center 04 A, LLC
615-10	SPORTS CENTER	8785 SW Beaverton Hillsdale Hwy	Raleigh Hills	OR	Ronald G. More and Claudia L. More

615-08	HILLSBORO	2200 N.W. Amberbrook Drive	Beaverton	OR	Toyama Maui Plantation, Inc.
615-13	WILSONVILLE	30050 SW Town Center Loop West	Wilsonville	OR	Village-Phase I, LLC
615-07	YAMHILL	110 S.W. Yamhill Street	Portland	OR	Yamhill Enterprises, Inc.
819-53	THREE RIVERS	119 Sixth Street	Pittsburgh	PA	119 Sixth Street Limited Liability Company
002-61	WALNUT-DREXEL	Drexel Building	Philadelphia	PA	ADR Drexel, L.P.
819-51	NORTH HILLS	551 Blazier Drive	Wexford	PA	BL McCandless, LP
022-10	CHELTENHAM	1000 Easton Road	Wyncote	PA	Cedarbrook Plaza, Inc.
022-08	SOUTH PHILLY	2419 South 24th Street	Philadelphia	PA	Cedar-South Philadelphia I, LLC
022-88	NORTHEAST	9185 Roosevelt Boulevard	Philadelphia	PA	Federal Realty Investment Trust
002-90WH4	WAREHOUSE (WILLOW GROVE 1)	2328 Wyandotte Road	Willlow Grove	PA	Foley & McCrudden, G.P.
002-09	FRANKLIN MILLS	851 Franklin Mills Circle	Philadelphia	PA	Franklin Mills LP
819-30	SOUTH HILLS	3000 Oxford Drive	Bethel Park	PA	Hal A. Kestler and Gerard J. Cipriani, Partners
002-67	ARAMINGO	3400 Aramingo Avenue	Philadelphia	PA	Imperial\Aramingo Limited Partnership
022-07	KING OF PRUSSIA	256 Mall Boulevard	King of Prussia	PA	KPHI Associates
022-09	MACDADE	1324 MacDade Boulevard	Woodlyn	PA	Mac Boulevard Associates, L.P.
022-07PK	KING OF PRUSSIA (PARKING)	234 Mall Boulevard	King of Prussia	PA	Offmak Associates
002-09	FRANKLIN MILLS	851 Franklin Mills Circle	Philadelphia	PA	Realty Income Pennsylvania Properties Trust
868-17	EAST PROVIDENCE	50 Narragansett Park Drive	East Providence	RI	Amalgamated Financial Group IV
868-18	NORTH PROVIDENCE	1920 Mineral Spring Avenue	North Providence	RI	Joseph Iaciofano
625-30	ALAMO	Quarry Market	San Antonio	TX	Alamo Vista Holdings, L.L.C. & Alamo Stonecrest Holdings, L.L.C.
625-11EX2	ARLINGTON (EXPANSION 2)	2306 South Collins	Arlington	TX	Arlington Stoneridge Assocaites, Ltd.
625-11	ARLINGTON	2306 South Collins	Arlington	TX	Arlington Stoneridge Associates, Ltd.
625-11EX	ARLINGTON (EXPANSION)	2306 South Collins	Arlington	TX	Arlington Stoneridge Associates, Ltd.
624-23	WOODWINDS ("Woodlands")	129 Sawdust Road	Spring	TX	Caltim, Ltd.
624-08	TOWERS/POST OAK II	1980A South Post Oak Road	Houston	TX	Crescent Real Estate Funding X, L.P.
625-33	THOUSAND OAKS	16628 San Pedro Avenue	San Antonio	TX	CW Park Oaks, LLC & FW Park Oaks, LLC
625-10	RED BIRD	7471 Marvin D. Love Fwy.	Dallas	TX	Dallas Fitness Center Development
624-19	CLEARLAKE	20761 Gulf Freeway	Webster	TX	DZM, Inc.
624-18	NORTHWEST CROSSING	13350 Northwest Freeway (Highway 290)	Houston	TX	EQYInvest Owner I, Ltd., LLP
625-24	SAN ANTONIO	5819 NW Loop 410, #150	San Antonio	TX	F.F. Exchange, LLC
625-14	MEADOW CREEK	1121 Northwest Highway	Garland	TX	Folsom Investments, Inc.
625-19	TURTLE CREEK II	One McKinney Plaza	Dallas	TX	Gaedeke Holdings, Ltd.
624-22	NORTH SHEPHERD II ("Garden Oaks")	3936 North Shepherd Drive	Houston	TX	Houston R.E. Income Properties XIV, L.P.
624-28	BARKER CYPRESS	17750 Katy Freeway, Suite 100	Houston	TX	I-10 Barker Cypress, Ltd.
625-09	SKILLMAN	6508 Skillman, Suite 10	Dallas	TX	Jack Arian
624-14	HUMBLE		Humble	TX	Jim R. Smith
624-14	SUGARLAND	Township Center Phase II The Market Place at Sugar Land		TX	Kho's Family I, L.P.
	LEWISVILLE	*	Sugarland	TX	• •
625-31	RICHARDSON II	500 East F.M. 3040	Lewisville		Kimco Lewisville, LP.
625-29		110 W. Campbell Road	Richardson	TX	KIR Richardson 572, Inc
625-35	TOWN EAST	1265 Town East Boulevard	Mesquite	TX	Logan Mesquite Associates, L.P.
625-18	BACHMAN LAKE	9655 Webb Chapel Road	Dallas	TX	Marketplace At Webb Chapel, Ltd.
625-13	MAYFAIR MEMORIAL II	650 Grapevine Highway	Hurst	TX	Mayfair Station LLC
624-27	MEMORIAL II	9801 Katy Freeway	Houston	TX	Metro National Corporation
625-15	WHITE LAKE	1201 Oakland Boulevard	Ft. Worth	TX	Miranda Partners, L.P.
624-20	WEST UNIVERSITY	2500 Dunstan	Houston	TX	NLAF Dunstan, L.P.
625-23	RIDGEMAR MALL	6833-A Green Oaks Road	Ft. Worth	TX	Oakwind, Ltd.
625-28	PRESIDENT'S SQUARE ("Marbach")	8725 Marbach Road	San Antonio	TX	President's Square Retail Associates, Ltd.

625-34	NORTH DALLAS	14902 Preston Road; Suite 110	Dallas	TX	Prestonwood Market Square, Ltd.
624-90WH2	WAREHOUSE (AFTON RD #2)	2098 Afton Road	Houston	TX	ProLogis Trust
625-90WH2	WAREHOUSE (REGENCY)	2104 Regency Drive	Irving	TX	R & M Lynn Investment, Ltd.
624-05	PASADENA	1418 Spencer Highway	Houston	TX	Spencer Retail, LLC
624-04	SHARPSTOWN	7255 Clarewood Drive	Houston	TX	T.J. Brothers Associates
625-20	PLANO ("Town Square")	910 W. Parker Road. #250	Plano	TX	Towne Square Shopping Center, L.P.
624-10	GREENSPOINT	Northborough Shopping Center	Houston	TX	Tricoastal Properties, Inc.
625-36	CARROLLTON	Carrollton Town Center	Carrollton	TX	Trinty Denton II, Ltd.
624-11	VILLAGE PLACE ("Fondren")	7737 West Bellfort Road	Houston	TX	Village Place, Ltd.
624-07	CHAMPIONS	Champions Village III	Houston	TX	Weingarten Realty Investors
624-23SN	WOODWINDS (SIGN)	Woodwinds Shopping Center	Woodwinds	TX	Woodwinds Merchants Association
678-01	SALT LAKE CITY	7020 South Union Park Ctr.	Midvale	UT	James Campbell Company LLC
678-02	SUGARHOUSE	2505 Parley's Way	Salt Lake City	UT	Parley's Investments, L.C.
848-65	PATRICK HENRY	12255 Hornsby Lane	Newport News	VA	Bond-Bally I Delaware Business Trust
848-90WH3	WAREHOUSE (PORTSMOUTH)	4011A-7 Seaboard Court	Portsmouth	VA	Greenwood Properties I, Ltd.
002-64	PRINCE WILLIAM CROSSING	13989 Noblewood Plaza	Woodbridge	VA	GSR Investments L.L.C.
002-53	LANDMARK ("Duke Street")	6200 Little River Turnpike	Alexandria	VA	Landmark HHH, LLC
002-20	FALLS CHURCH ("Loehmann's")	Loehmann's Plaza Shop Ctr.	Falls Church	VA	Loehmann's Plaza Ltd Partnership
002-66	PENTAGON SQUARE	1201 S. Joyce St., Suite C6	Arlington	VA	Street Retail, Inc.
848-55	VIRGINIA BEACH	3960 Virginia Beach Blvd.	Virginia Beach	VA	Thalia Wayside Investment Properties, Ltd. L.L.P.
609-03	AUBURN	1314 Auburn Way North	Auburn	WA	AEB, LLC
616-01	FISHER'S LANDING	16096 S.E. 15th Street	Vancouver	WA	Barclay Village, LLC or Don Weidenweber
683-03	PUYALLUP	3600 9th Street SW	Puyallup	WA	Cafaro Northwest Partnership
683-11	EVERETT	7621 Evergreen Way	Everett	WA	Columbia Cascade Plaza LLC
609-02	PUYALLUP II	716 South Hill Park Drive	Puyallup	WA	East-West Investment Company, Inc.
616-01SN	FISHER'S LANDING (SIGN)	16096 S.E. 15th Street	Vancouver	WA	Garrett Sign Company, Inc.
683-08CL2	TACOMA (EASEMENT AGMT)	James Center	Tacoma	WA	James Center North Limited Partnership
683-05	BRIDLE TRAILS	6601 132nd Avenue NE	Kirkland	WA	Joshua Green Corporation
609-04	RENTON	17110 116th Avenue SE	Renton	WA	MBA Cascade Plaza LLC
683-09PK	BELLEVUE (ADD. PARKING)	3215 148th Avenue SE	Bellevue	WA	Milton and Sue M. Walter
609-04SN	RENTON (SIGN)	116th Avenue and 170th Street West Side	Renton	WA	Mons Pico LLC
683-08SN	TACOMA (SIGN)	1680 South Mildred	Tacoma	WA	Mr. Donald G. Huber
609-01	FEDERAL WAY II	31701 20th Avenue South	Federal Way	WA	Mr. Travis Hollman
683-10	NORTH SEATTLE	13201 Aurora Avenue North	Seattle	WA	Thrifty Payless, Inc.
710-10	BROOKFIELD	16985 West Bluemound Road	Brookfield	WI	Bluemound Office Company
710-10PK	BROOKFIELD (PARKING)	16985 West Bluemound Road	Brookfield	WI	Bluemound Office Company
710-10OF3	BROOKFIELD (2ND FLOOR)	16985 West Bluemound Road	Brookfield	WI	Corporate Development Corp.
710-16	DOWNTOWN	1237 No. Van Buren Street	Milwaukee	WI	Jerome M. Cohen
710-17	WEST ALLIS	901 South 60th Street	West Allis	WI	LST Milwaukee Limited Partnerhsip
710-18	CUDAHY	2525 East Layton Avenue	Cudahy	WI	S & T Layton Avenue Partnership
710-02	SOUTHRIDGE	5474 South 76th Street	Greendale	WI	Southridge Company
710-16PK	DOWNTOWN (PARKING)	Juneau Postal Station	Milwaukee	WI	United States Postal Service

Exhibit 4

[Notice of Confirmed Plan]

UNITED STATES BANKRUPTCY COURT SOUTHERN DISTRICT OF NEW YORK

In re:		_)	Chapter 11
	BALLY TOTAL FITNESS OF)	G N 05 10005 (DDI)
	GREATER NEW YORK, INC., et al.,)	Case No. 07-12395 (BRL)
	Debtors.)	
)	Jointly Administered

NOTICE OF ENTRY OF ORDER CONFIRMING DEBTORS'
FIRST AMENDED JOINT PREPACKAGED PLAN OF REORGANIZATION
PURSUANT TO SECTION 1129 OF THE BANKRUPTCY CODE AND
RULE 3020 OF THE FEDERAL RULES OF BANKRUPTCY PROCEDURE

TO ALL CREDITORS, EQUITY INTEREST HOLDERS, AND OTHER PARTIES IN INTEREST:

PLEASE TAKE NOTICE that an order (the "<u>Confirmation Order</u>") confirming The First Amended Joint Prepackaged Chapter 11 Plan Of Reorganization Of Bally Total Fitness Holding Corporation And Its Affiliate Debtors, dated September [____], 2007 (as amended and supplemented, the "<u>Plan</u>"), of the above-captioned debtors and debtors-in-possession (collectively, the "<u>Debtors</u>")¹, was entered by the Honorable Burton R. Lifland, United States Bankruptcy Judge, United States Bankruptcy Court for the

¹ The Debtors in these proceedings are: Bally Total Fitness of Greater New York, Inc., Bally Total Fitness Holding Corporation, Bally Total Fitness Corporation, Bally ARA Corporation, Bally Fitness Franchising, Inc., Bally Franchise RSC, Inc., Bally Franchising Holdings, Inc., Bally Real Estate I LLC, Bally REFS West Hartford, LLC, Bally Sports Clubs, Inc., Bally Total Fitness Franchising, Inc., Bally Total Fitness International, Inc., Bally Total Fitness of California, Inc., Bally Total Fitness of Colorado, Inc., Bally Total Fitness of Connecticut Valley, Inc., Bally Total Fitness of Minnesota, Inc., Bally Total Fitness of Missouri, Inc., Bally Total Fitness of Philadelphia, Inc., Bally Total Fitness of Rhode Island, Inc., Bally Total Fitness of the Mid-Atlantic, Inc., Bally Total Fitness of the Midwest, Inc., Bally Total Fitness of the Southeast, Inc., Bally Total Fitness of Toledo, Inc., Bally Total Fitness of Upstate New York, Inc., BTF Cincinnati Corporation, BTF Europe Corporation, BTF Indianapolis Corporation, BTF Minneapolis Corporation, BTF Corporation, BTF Corporation, BTF Corporation, Greater Philly No. 1 Holding Company, Greater Philly No. 2 Holding Company, Health & Tennis Corporation of New York, Holiday Health Clubs of the East Coast, Inc., Holiday/Southeast Holding Corp., Jack La Lanne Holding Corp., New Fitness Holding Co., Inc., Nycon Holding Co., Inc., Rhode Island Holding Company, Tidelands Holiday Health Clubs, Inc., and U.S. Health, Inc.

Southern District of New York (the "<u>Bankruptcy Court</u>") on September [___], 2007. Unless otherwise defined in this notice, capitalized terms used herein shall have the meanings ascribed to them in the Plan and the Confirmation Order.

PLEASE TAKE FURTHER NOTICE that the Plan and Confirmation Order are available for inspection during regular business hours in the office of the Clerk of the Bankruptcy Court at the United States Bankruptcy Court, Alexander Hamilton Custom House, One Bowling Green, New York, New York 10004. The Plan and Confirmation Order are also available by accessing the Bankruptcy Court's website www.nysb.uscourts.gov. Please note that a PACER password and login are required to access documents on the Bankruptcy Court's website. The Plan and Confirmation Order are also available free of charge on the Reorganized Debtors' restructuring website located at www.kccllc.net/bally.

PLEASE TAKE FURTHER NOTICE that the Plan and its provisions are binding on the Debtors, the Reorganized Debtors, any entity acquiring or receiving property or a distribution under the Plan, and any present or former holder of a Claim against or Interest in any of the Debtors and their successors, assigns, and parties- in-interest, including all governmental entities, whether or not the Claim or Interest of such holder is impaired under the Plan and whether or not such holder or entity voted to accept or reject the Plan.

PLEASE TAKE FURTHER NOTICE that October 31, 2007, at 5:00 p.m. (prevailing Eastern time) (the "Bar Date") was established by the Bankruptcy Court as the deadline by which holders of 510(b) Equity Claims must file proofs of claim against the Debtors.

PLEASE TAKE FURTHER NOTICE that any holder of a 510(b) Equity Claim who is required, but fails, to file a proof of claim with the Bankruptcy Court on or before the Bar Date shall be forever barred, estopped and enjoined from asserting such claim against the Debtors (or filing a proof of claim with respect thereto), and the Debtors' property shall be forever discharged from any and all indebtedness or liability with respect to such claim, and such holder shall not be treated as a creditor with respect to such claim for the purpose of distribution in these Chapter 11 Cases or be entitled to receive further notices regarding such 510(b) Equity Claim.

PLEASE TAKE FURTHER NOTICE THAT ALL PLEADINGS FILED WITH, AND ORDERS GRANTED BY, THE BANKRUPTCY COURT ARE AVAILABLE FOR INSPECTION ON THE BANKRUPTCY COURT'S INTERNET SITE AT WWW.NYSB.USCOURTS.GOV AND AT NO COST FROM THE REORGANIZED DEBTORS' RESTRUCTURING WEBSITE: WWW.KCCLLC.NET/BALLY.

Dated: September [___], 2007 New York, NY Respectfully submitted,

LATHAM & WATKINS LLP

By: /s/ David S. Heller
David S. Heller
Richard A. Levy
Keith A. Simon
Sears Tower, Suite 5800
233 South Wacker Drive
Chicago, Illinois 60606-6401
Telephone: (312) 876-7700
Facsimile: (312) 993-9767

-and-

By: /s/ Henry P. Baer, Jr.
Henry P. Baer, Jr. (HB-3866)
Joseph Furst, III (JF-6136)
885 Third Avenue, Suite 1000
New York, New York 10022
Telephone: (212) 906-1200
Facsimile: (212) 751-4864

Counsel for Debtors and Debtors-in-Possession

Exhibit 5

Guarantors of New Senior Second Lien Notes

Bally Fitness Franchising, Inc.

Bally Franchise RSC, Inc.

Bally Franchising Holdings, Inc.

Bally Total Fitness Corporation

Bally Total Fitness International, Inc.

Bally Total Fitness of Missouri, Inc.

Bally Total Fitness of Toledo, Inc.

Bally Total Fitness of Connecticut Coast, Inc.

Bally Total Fitness of Connecticut Valley, Inc.

Greater Philly No. 1 Holding Company

Greater Philly No. 2 Holding Company

Health & Tennis Corporation of New York

Holiday Health Clubs of the East Coast, Inc.

Bally Total Fitness of Upstate New York, Inc.

Bally Total Fitness of Colorado, Inc.

Bally Total Fitness of the Southeast, Inc.

Holiday/Southeast Holding Corp.

Bally Total Fitness of California, Inc.

Bally Total Fitness of the Mid-Atlantic, Inc.

BTF/CFI, Inc.

Bally Total Fitness of Greater New York, Inc.

Jack La Lanne Holding Corp.

Bally Sports Clubs, Inc.

New Fitness Holding Co., Inc.

Nycon Holding Co., Inc.

Bally Total Fitness of Philadelphia, Inc.

Bally Total Fitness of Rhode Island, Inc.

Rhode Island Holding Company

Bally Total Fitness of the Midwest, Inc.

Bally Total Fitness of Minnesota, Inc.

Tidelands Holiday Health Clubs, Inc.

U.S. Health, Inc.